Owens Realty Mortgage, Inc. Form DEF 14A May 17, 2016

1) Amount previously paid:

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 14A

Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No. ___)

Filed by the	X	Filed by a Party other than	
Registrant		the Registrant	
	nary Proxy State	ement ne Commission Only (as permitted by Rule 14a-6(e)(2))	
x Definiti	ve Proxy Statem		
O		Definitive Additional Materials	
0		Soliciting Material under §240.14a-12	
		Owens Realty Mortgage, Inc.	
		(Name of Registrant as Specified In Its Charter)	
Payment of Filin x No fee no	required. Fee comput	ted on table below per Exchange Act Rules 14a-6(i)(1) and 0-11. Itle of each class of securities to which transaction applies:	
	2) Ag	ggregate number of securities to which transaction applies:	
·	•	er underlying value of transaction computed pursuant to Exchange Act Rulhich the filing fee is calculated and state how it was determined):	le
	4	4) Proposed maximum aggregate value of transaction:	
		5) Total fee paid:	
0		Fee paid previously with preliminary materials:	
which the offs	• •	fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify to paid previously. Identify the previous filing by registration statement number its filing	_

2) Form, Schedule or Regi	stration Statement No.:	
	3) Filing Party:	
	4) Date Filed:	

May 17, 2016

To our Stockholders:

It is my pleasure to invite you to attend the 2016 Annual Meeting of Stockholders (the "Annual Meeting") of Owens Realty Mortgage, Inc. (the "Company") to be held on Monday, June 27, 2016 at 10:00 a.m., Pacific Daylight Time, at the Walnut Creek Marriott, 2355 N. Main Street, Walnut Creek, CA 94596. Details regarding admission to the Annual Meeting and the business to be conducted are more fully described in the accompanying Notice of Annual Meeting and Proxy Statement.

Your vote is very important. Whether or not you plan to attend the Annual Meeting, it is important that your shares be represented and voted. I hope that you will vote as soon as possible. Please review the instructions described in the Proxy Statement and attached proxy card, which explain how to vote via Internet, phone, or mail.

On behalf of your Board of Directors, thank you for your ongoing support of the Company.

Sincerely,

William C. Owens Chairman of the Board

{00092520; 2}

Owens Realty Mortgage, Inc.

2221 Olympic Boulevard Walnut Creek, California 94595

NOTICE OF ANNUAL MEETING OF STOCKHOLDERS TO BE HELD ON JUNE 27, 2016

To our Stockholders:

On behalf of the Board of Directors, it is my pleasure to invite you to the 2016 annual meeting of stockholders (the "Annual Meeting") of Owens Realty Mortgage, Inc. a Maryland corporation ("ORM" or the "Company"), to be held at the Walnut Creek Marriott, 2355 N. Main Street, Walnut Creek, CA 94596, on June 27, 2016, at 10:00 a.m. Pacific Daylight Time, to consider and vote on the following matters:

- 1. To elect two (2) Class III directors, each to serve for a term continuing until the annual meeting of stockholders held in 2019 and until their respective successors are duly elected and qualified;
- 2. To ratify the appointment of Crowe Horwath LLP as the Company's independent registered public accounting firm for the year ending December 31, 2016; and
- 3. To transact such other business as may properly come before the Annual Meeting or any adjournments or postponements thereof.

Only stockholders of record of the Company at the close of business on May 5, 2016 are entitled to receive notice of, and vote at, the Annual Meeting and any postponements or adjournments thereof. The Company urges you to vote promptly. You may vote by completing, signing and returning the enclosed proxy card in the envelope provided, which requires no postage if mailed in the United States. Internet and telephone voting procedures are discussed in the Proxy Statement and on the proxy card. For shares held through a bank, broker or other nominee, you must follow the instructions provided by the bank, broker or other nominee regarding how to instruct your bank, broker or other nominee to vote your shares. If you attend the Annual Meeting, you may withdraw your proxy and vote in person if you so choose.

To attend the Annual Meeting in person, please register in advance by sending an email to investors@owensmortgage.com or by phone at (925) 935-3840. Attendance at the Annual Meeting will be limited to persons presenting proof of stock ownership as of the record date and government-issued photo identification (such as a valid driver's license or passport). If you hold shares directly in your name as the stockholder of record, proof of ownership would include a copy of your account statement. If you hold shares through an intermediary, such as a broker, bank or other nominee, proof of stock ownership would include a proxy from your broker, bank or other nominee or a copy of your brokerage or bank account statement. Additionally, if you intend to vote your shares at the meeting, you must request a "legal proxy" from your broker, bank or other nominee and bring this legal proxy to the meeting.

By Order of the Board of Directors,

Daniel J. Worley Senior Vice President and Secretary Walnut Creek, California May 17, 2016 Enclosures

TABLE OF CONTENTS

Page NOTE REGARDING THIS PROXY STATEMENTi.
PROXY STATEMENT FOR THE 2016 ANNUAL MEETING1
QUESTIONS AND ANSWERS ABOUT VOTING YOUR SHARES1.
PROPOSAL 1: ELECTION OF TWO (2) DIRECTORS
PROPOSAL 2: RATIFICATION OF APPOINTMENT OF INDEPENDENT ACCOUNTING FIRM
REPORT OF THE AUDIT COMMITTEE
GOVERNANCE OF OUR COMPANY9.
INFORMATION REGARDING OUR EXECUTIVE OFFICERS
EXECUTIVE COMPENSATION
COMPENSATION COMMITTEE REPORT
COMPENSATION OF INDEPENDENT DIRECTORS
SECURITY OWNERSHIP OF MANAGEMENT AND CERTAIN BENEFICIAL OWNERS 17
RELATED PARTY TRANSACTIONS
SECTION 16(a) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE
PROPOSALS OF STOCKHOLDERS
DELIVERY AND HOUSEHOLDING OF PROXY MATERIALS
FINANCIAL STATEMENTS AND OTHER SEC FILINGS AVAILABLE
OTHER MATTERS

NOTE REGARDING THIS PROXY STATEMENT

As previously announced, as part of a plan to reorganize our business operations so that, among other things, we could elect to qualify as a real estate investment trust (a "REIT") for federal income tax purposes, effective May 20, 2013, Owens Mortgage Investment Fund, a California Limited Partnership (the "Predecessor" or "OMIF") merged with and into Owens Realty Mortgage, Inc., a Maryland corporation ("ORM") with ORM as the surviving corporation (the "Merger"), and ORM commenced conducting all of the business conducted by the Predecessor. Upon consummation of the Merger, limited partners of the Predecessor received one share of common stock, par value \$0.01 per share, of ORM (the "Common Stock"), for every 25 limited partner units of the Predecessor that they owned, and certain units of the Predecessor representing the general partner interest of Owens Financial Group, Inc. were also exchanged for Common Stock as is discussed in further detail in the Company's Annual Report on Form 10-K.

In addition, following the Merger the Company established a new Board of Directors, Audit Committee, Nominating and Corporate Governance Committee, and Compensation Committee and established new and additional governance guidelines discussed further in this Proxy Statement. The rights of the stockholders of the Company are now governed by Maryland law and the charter, bylaws and other governing documents of ORM.

References to Owens Realty Mortgage, Inc. and its subsidiaries, "ORM," the "Company," "we", "us", or "our" in this Pro Statement have the following meanings, unless we specifically state or the context requires otherwise:

- For periods prior to May 20, 2013: the Predecessor and its subsidiaries;
 - For periods from and after May 20, 2013: ORM and its subsidiaries.

i

Owens Realty Mortgage, Inc. 2221 Olympic Boulevard Walnut Creek, California 94595

PROXY STATEMENT FOR ANNUAL MEETING OF STOCKHOLDERS TO BE HELD ON JUNE 27, 2016

This Proxy Statement is being furnished to stockholders in connection with the solicitation of proxies by the Board of Directors ("Board" or "Board of Directors") of Owens Realty Mortgage, Inc., a Maryland corporation ("ORM", the "Company", "we", "our" or "us"), for use at ORM's 2016 annual meeting of stockholders (the "Annual Meeting") to be hel the Walnut Creek Marriott, 2355 N. Main Street, Walnut Creek, California 94596, on June 27, 2016, at 10:00 a.m., Pacific Daylight Time, or at any postponements or adjournments thereof. This Proxy Statement, the accompanying proxy card and our annual report to stockholders, which includes our Annual Report on Form 10-K with audited financial statements for the year ended December 31, 2015, are first being sent to our stockholders on or about May 17, 2016.

Important Notice Regarding the Availability of Proxy Materials for the Annual Meeting of Stockholders to be Held on June 27, 2016: This Proxy Statement and our Annual Report on Form 10-K are available free of charge on the internet at www.owensmortgage.com. On this site, you will be able to access this Proxy Statement, our Annual Report on Form 10-K for the fiscal year ended December 31, 2015, and any amendments or supplements to the foregoing material that is required to be furnished to stockholders.

QUESTIONS AND ANSWERS ABOUT VOTING YOUR SHARES

Why am I receiving these ORM's Board of Directors is providing these proxy materials to you in materials?

connection with ORM's Annual Meeting to be held on June 27, 2016.

Stockholders are requested to vote on the proposals described in this Proxy Statement.

What is included in the seThese materials include: materials?

- · our Proxy Statement and a proxy card for the Annual Meeting; and
- · our 2015 Annual Report on Form 10-K, which includes our audited consolidated financial statements.

What proposals are being ORM intends to present the following proposals for stockholder presented consideration and voting at the Annual Meeting: at the Annual Meeting?

· Proposal 1: Election of Two (2) Directors: the election of two (2) members of Class III of ORM's Board of Directors, each to serve until the annual meeting of stockholders in 2019 and until their respective

successors are duly elected and qualified; and

· Proposal 2: Ratification of Appointment of Independent Accounting Firm: The ratification of the appointment of Crowe Horwath LLP as ORM's independent registered public accounting firm for the year ending December 31, 2016.

The Board is not aware of any matters to be presented for action at the Annual Meeting other than the matters set forth herein. If any other matters requiring a vote of stockholders arise at the Annual Meeting, it is the intention of the persons named in the proxies to vote on such matter in accordance with their discretion.

What are the ORM Board's voting recommendations?

ORM's Board of Directors recommends that you vote as follows:

- · Proposal 1: Election of Two (2) Directors: "FOR" the election of both nominees listed in this Proxy Statement to serve as Class III Board members; and
- · Proposal 2: Ratification of Appointment of Independent Accounting Firm: "FOR" the ratification of the appointment of Crowe Horwath LLP as our independent registered public accounting firm for the year ending December 31, 2016.

What is a quorum?

A quorum is necessary to hold a valid meeting. The presence, in person or by proxy, of holders of ORM common stock ("Common Stock") entitled to cast a majority of all the votes entitled to be cast shall constitute a quorum for the Annual Meeting. Since there were 10,247,477 outstanding shares of Common Stock, each entitled to one vote per share as of the record date, we will need at least 5,123,739 votes present in person or by proxy at the Annual Meeting for a quorum to exist. If there are not sufficient votes present for a quorum or to approve or ratify any of the proposals at the time of the Annual Meeting, we expect that the meeting will be adjourned in order to permit further solicitation of proxies by the Company.

What is the effect of abstentionsShares represented by proxies received but marked as abstentions and and broker "non-votes"?

shares held in a broker's account that are voted by the broker or other nominee on some but not all matters will be treated as shares present for purposes of determining the presence of a quorum.

However, if you hold your shares in street name and do not provide voting instructions to your broker, your shares will not be voted on any proposal on which your broker does not have discretionary authority to vote (a "broker non-vote"). Under current rules, your broker will not have discretionary authority to vote your shares at the Annual Meeting on Proposal 1 (election of the directors). Consequently, if you do not give your broker instructions, your broker will not be able to vote on this proposal.

What vote is required in order to approve each proposal?

The proposals to be voted on at the Annual Meeting have the following voting requirements:

· Proposal 1: Election of Two (2) Directors: the affirmative vote of a "plurality" of the votes cast at the Annual Meeting is required to elect each nominee. In the context of the election of two directors at the Annual Meeting, it means that the two director nominees receiving the highest number of FOR votes cast will be elected. Votes "withheld" from one or more of the director nominees and broker non-votes therefore will have no effect on the outcome of the vote with respect to the election of directors.

· Proposal 2: Ratification of Appointment of Independent Accounting Firm: the affirmative vote of a majority of the votes cast at the Annual Meeting is required to ratify the appointment of Crowe Horwath LLP to serve as our independent registered public accounting firm for the year ending December 31, 2016. For purposes of the vote on Proposal 2, abstentions will not be counted as votes cast and will have no effect on the result of the vote.

Who is entitled to vote?

The Board of Directors has fixed the close of business on May 5, 2016, as the record date, and only stockholders of record at the close of business on the record date are entitled to receive notice of and to vote at the Annual Meeting and any adjournments or postponements thereof. On the record date, there were 10,247,477 shares of our Common Stock outstanding.

How many votes do I have?

Holders of record of our Common Stock on the record date will be entitled to one vote per share. Stockholders may not cumulate votes in the election of directors.

How do I attend the Annual Meeting?

To attend the Annual Meeting in person, please register in advance by sending an email to investors@owensmortgage.com or by phone at (925) 935-3840. For directions to the Annual Meeting please call (925) 935-3840.

Attendance at the Annual Meeting will be limited to persons presenting proof of stock ownership on the record date and government-issued photo identification (such as a valid driver's license or passport). If you hold shares directly in your name as the stockholder of record, proof of ownership would include a copy of your account statement. If you hold shares through an intermediary, such as a broker, bank or other nominee, proof of stock ownership would include a proxy from your broker, bank or other nominee or a copy of your brokerage or bank account statement.

How do I vote?

If your ORM shares are registered in your name, you may vote your shares in person at the Annual Meeting or by proxy. You can vote by proxy via Internet, phone or mail. Please follow the instructions provided on the proxy card or voting instruction card.

If you hold your Common Stock in an account with a bank, broker or other nominee (i.e. in "street name"), you may vote by following the instructions on the voting instruction card provided to you by your bank, broker or other nominee.

my vote?

May I revoke my proxy or change Yes. You may revoke your proxy or change your voting instructions prior to the vote at the Annual Meeting. You may enter a new vote by using the Internet or the telephone or by mailing a new proxy card or new voting instruction card bearing a later date (which will automatically revoke your earlier voting instructions), or by attending the Annual Meeting and voting in person. Your attendance at the Annual Meeting in person will not cause your previously granted proxy to be revoked unless you specifically so request.

> Notices of revocation of proxies delivered by mail must be sent and delivered to the Company's principal offices at 2221 Olympic Boulevard, Walnut Creek, CA 94595, Attention: Dan Worley, Secretary.

Who will count the vote?

A representative of Computershare, Inc. will tabulate the votes and act as the inspector of election.

Who will bear the cost of solicitingORM will pay the cost of soliciting proxies on the form accompanying votes for the Annual Meeting? these proxy materials. In addition to the use of mail, the solicitation of

these proxy materials. In addition to the use of mail, the solicitation of proxies or votes may be made in person, by telephone or by electronic communication by ORM's directors and officers and by employees of Owens Financial Group, Inc., the Company's external manager (the "Manager"), who will not receive any additional compensation for such solicitation activities. In addition, ORM may reimburse brokerage firms and other persons representing beneficial owners of shares for their reasonable expenses in forwarding solicitation material to such beneficial owners.

Where can I find the voting resultsORM will announce preliminary voting results at the Annual Meeting of the Annual Meeting? and publish preliminary, or final results if available, in a Current Report on Form 8-K within four business days of the Annual Meeting.

PROPOSAL 1: ELECTION OF TWO (2) DIRECTORS

On November 12, 2013, the Company acted by resolution of the Board of Directors to elect to be subject to all of the provisions of Sections 3-803, 3-804 and 3-805 of Title 3, Subtitle 8 of the Maryland General Corporation Law ("MGCL"). As a result of the Company's election to be subject to Section 3-803 of the MGCL, the Board is classified into three separate classes of directors. Directors in each class are elected for a staggered term of three-years each, with a term of office of only one of these three classes of directors expiring each year. Previously, the Board consisted of a single class of directors, with directors required to stand for election every year.

The Class III directors, William C. Owens and Dennis G. Schmal, have a term expiring at the Annual Meeting upon the election and qualification of their respective successors. There is also one Class I director with a term continuing until the annual meeting of stockholders in 2017 and until his successor is elected and qualified, and two Class II directors with terms continuing until the annual meeting of stockholders in 2018 and until their respective successors are elected and qualified. At each annual meeting of stockholders, the successors to the class of directors whose term expires at that annual meeting of stockholders will be elected to hold office for a term continuing until the annual meeting of stockholders held in the third year following the year of their election and until their successors are elected and qualified, or until their earlier death, resignation or removal.

The Nominating and Corporate Governance Committee of the Board has recommended, and the Board has nominated, Messrs. Owens and Schmal (the "Nominees") to stand for re-election at the Annual Meeting and to hold office until the annual meeting of stockholders to be held in 2019 and until their respective successors are duly elected and qualified. Each Nominee has agreed to serve as a director if elected and has consented to being named as a nominee. Accordingly, the Board has no reason to believe that either Nominee will be unable or unwilling to serve. If either Nominee should decline or be unable to serve as a director, it is intended that the proxy will be voted for the election of such person as is nominated as a replacement by the Nominating and Corporate Governance Committee and by the Board.

Information about the Director Nominees and the Continuing Directors

The Board is comprised of five members divided into three classes serving staggered terms. M. Lyman Bates, Jr. is the Class I director, Bryan H. Draper and James M. Kessler are the Class II directors, and William C. Owens and Dennis G. Schmal are the Class III directors nominated for re-election at this Annual Meeting. Our charter (the

"Charter") and bylaws, and the provisions of Section 3-804 of Title 3, Subtitle 8 of the MGCL applicable to the Company, provide that only the Board may increase or decrease the number of directors, but the number of directors may never be less than the minimum required by the MGCL (which is one) nor (unless our bylaws are amended) more than 15.

As a result of our adoption of the provisions of Section 3-804 of Title 3, Subtitle 8 of the MGCL, any vacancies occurring on our Board of Directors, including vacancies occurring as a result of the death, resignation, or removal of a director, or due to an increase in the size of the Board of Directors, may be filled only by the affirmative vote of a majority of the remaining directors in office, even if the remaining directors do not constitute a quorum, and any director elected to fill a vacancy will serve for the remainder of the full term of the directorship in which the vacancy occurred and until a successor is duly elected and qualifies.

Messrs. Owens and Schmal, the Nominees for election as directors at the Annual Meeting, have each been evaluated by the Nomination and Corporate Governance Committee pursuant to the guidelines described below under "Governance of Our Company—Director Qualifications" and the determination has been made that each of the Nominees fulfills and exceeds the qualities that we look for in members of our Board. There is no familial relationship among any of the members of our Board or our executive officers. Messrs. Owens and Draper are affiliates of our Manager and are not independent under the listing standards and related rules of the NYSE MKT (the "NYSE MKT rules"). Each of Messrs. Schmal, Kessler and Bates is independent as defined in the NYSE MKT rules. See "Governance of Our Company— Director Independence."

The information set forth below is as of May 6, 2016 for the Company's director Nominees and each of the Company's continuing directors. The business address of the Nominees and each of the continuing directors is c/o Owens Realty Mortgage, Inc., 2221 Olympic Boulevard, Walnut Creek, California 94595.

Nominees for Class III Directors (Current terms Expire at the 2016 Annual Meeting):

William C. Owens – Mr. Owens, age 65, has been our executive Chairman of the Board since ORM's inception in 2012. Mr. Owens also served as our President from our inception in 2012 until January 2016, and as our Chief Executive Officer from January 2013 until January 2016. He has been President of our Manager since April 1996, and is a member of the Board of Directors and the Loan Committee of the Manager and its Chief Executive Officer. Mr. Owens is expected to continue as a director and President and Chief Executive Officer of the Manager until he resigns or is replaced by a vote of the Manager's stockholders and board of directors, respectively. From 1979 until April 1996, he served as a Senior Vice President of the Manager. Mr. Owens is also currently the President of Investors Yield, Inc. (a California Trust Company) which is owned 100% by the Manager. Mr. Owens has been active in real estate construction, development, and mortgage financing since 1973. Prior to joining Owens Mortgage Company in 1979, Mr. Owens was involved in mortgage banking, property management and real estate development. As President of the Manager, Mr. Owens is responsible for the overall activities and operations of the Manager, including corporate investment, operating policy and planning. In addition, he is responsible for loan production, including the underwriting and review of potential loan investments. Mr. Owens graduated from Westmont College in 1973 and is a licensed real estate broker.

Mr. Owens' extensive experience in the mortgage financing and real estate industries, his deep knowledge of our business as our executive Chairman of the Board, as our former Chief Executive Officer and as the Chief Executive Officer of our Manager, and his prior leadership experience make him well qualified to serve as Chairman of our Board and strengthens our Board's collective qualifications, skills, experience and viewpoints.

Dennis G. Schmal – Mr. Schmal, age 69, has been a member of our Board since the consummation of the merger (the "Merger") of our predecessor, Owens Mortgage Investment Fund, a California limited partnership ("OMIF") with and into the Company on May 20, 2013. Since 2003, Mr. Schmal has served as a director and a member of the audit (chairman), nominating, and compensation committees of Merriman Holdings, Inc., a securities and investment banking firm. In May of 2015, Mr. Schmal began service as a director and as chair of the audit committee of a public technology company, Blue Calypso, Inc. Mr. Schmal has also served as a director, and is the chairman of the audit committees, of the public investment funds overseen by three asset management complexes, AssetMark GuideMark/GuidePath Funds (since 2006), the Wells Fargo ASGI Hedge Funds (since 2008), and the Cambria ETF Series (since 2013). In addition, since 2005, Mr. Schmal has served as chairman of the board of directors of Pacific Metrics Corporation, a private company in the educational assessment/software field. Mr. Schmal served as chairman of the board of directors of a technology industry startup, Sitoa Global Inc., from January 2012 until April 2013. From August 2004 to November 2011, he served as a member of the board of directors, audit committee (chairman) and compensation committee of Varian Semiconductor, a semiconductor equipment manufacturer. From October 2008 to May 2011, he served as a director and chairman of the audit committee of Grail Advisors EFT Trust, an exchange traded fund complex. From September 2006 to May 2008, Mr. Schmal served as a director and chairman of the audit

committee of North Bay Bancorp, a bank holding company. From February 1972 to April 1999, Mr. Schmal was employed by Arthur Andersen LLP, primarily as a partner overseeing the delivery of accounting and audit services. During his career with Arthur Andersen LLP, Mr. Schmal specialized in working with companies in the financial services sector, including the commercial banking, securities/investment banking and asset management industries. Mr. Schmal received a Bachelor of Science in Business Administration – Finance and Accounting Option from California State University, Fresno in 1972 and holds a CPA certificate (retired).

Mr. Schmal's extensive experience serving on boards and the audit and other key committees of multiple private and public companies, as well as his extensive expertise in accounting and financial matters, make him well qualified to serve as a director of the Company and strengthens our Board's collective qualifications, skills, experience and viewpoints.

Class I Director (Term expires at the 2017 Annual Meeting):

M. Lyman Bates, Jr. – Mr. Bates, age 74, has been a member of our Board since the consummation of the Merger on May 20, 2013. From 2005 to 2013, Mr. Bates served as a consultant to Deloitte & Touche LLP as a Regional Compliance Officer. From 1976 to 2005, Mr. Bates served as an Audit Partner for Deloitte & Touche LLP. He participated in and managed audit engagements in the real estate, construction, high tech, leasing and retail industries ranging from private start-ups to \$30 billion public corporations. He served as the National Audit Partner for the leasing industry for over ten years. Mr. Bates received a B.A. degree from Williams College and a M.B.A. with a concentration in accounting from Stanford University in 1966. He is a Certified Public Accountant licensed in California and a member of the American Institute of Certified Accountants. Mr. Bates is also a Member of the Society of Corporate Compliance & Ethics and a Certified Compliance Professional.

Mr. Bates' extensive experience in audit, accounting and other financial matters, as well as his familiarity with the leasing and real estate industries, make him well qualified to serve as a director of the Company and strengthens our Board's collective qualifications, skills, experience and viewpoints.

Class II Directors (Terms expire at the 2018 Annual Meeting):

Bryan H. Draper – Mr. Draper, age 58, has served as our President and CEO since January 2016 and as a member of our Board since the consummation of the Merger on May 20, 2013. Mr. Draper served as the Secretary and Treasurer of the Company from its inception in 2012 until January 2016, and as our Chief Financial Officer from January 2013 to January 2016. He has also been Chief Financial Officer and Corporate Secretary of our Manager since December 1987, and a member of the Board of Directors of the Manager since January 1997. Mr. Draper is also currently the Chief Financial Officer of Investors Yield, Inc. (a California Trust Company) which is owned 100% by the Manager. Mr. Draper is expected to continue as a director and Corporate Secretary and Chief Financial Officer of the Manager until he resigns or is replaced by a vote of the Manager's stockholders and board of directors, respectively. Mr. Draper is a certified public accountant and is responsible for all accounting, finance, and tax matters for the Manager. Mr. Draper received a Master's degree in business administration from the University of Southern California in 1981.

Mr. Draper's extensive experience in the mortgage financing and real estate industries, his deep knowledge of our business as our Chief Executive Officer, as our former Chief Financial Officer and as the Chief Financial Officer of our Manager, as well as his expertise in accounting and financial matters make him well qualified to serve a director of the Company and strengthens our Board's collective qualifications, skills, experience and viewpoints.

James M. Kessler – Mr. Kessler, age 63, has been a member of our Board since the consummation of the Merger on May 20, 2013. He has been the President of Stonehenge Property Group, a private real estate development and advisory services company, since its inception in August 2005. Mr. Kessler founded Stonehenge Property Group and is responsible for all of its operations including budgets, business plans, property acquisition and development, financing and leasing. From January 2004 to July 2005, he was the founder and principal of Highland Development Company, a retail acquisition and development company that was part of the Marcus & Millchap group of companies. From April 2002 to October 2003, Mr. Kessler served as the Chief Operating Officer of ScanlanKemperBard Companies, a private real estate investment firm. From July 1999 to February 2002, he served as the Chief Development Officer of Federal Realty Investment Trust, a publicly traded REIT. While at Federal Realty Investment Trust, Mr. Kessler was responsible for establishing and managing regional and satellite offices, development, redevelopment, construction, operations, asset management and leasing. From December 1989 to July

1999, Mr. Kessler was the Chief Development Officer of Burnham Pacific Properties/The Martin Group, a publically traded REIT. From 1985 to December 1989, he served as the Director of Marketing of Transpacific Development, a private real estate development and management company. Mr. Kessler received his bachelor's degree in business administration and management from Golden Gate University in 1981.

Mr. Kessler's extensive experience in the real estate investment, development and management industries and his senior management positions in two publicly traded REITs make him well qualified to serve as a director of the Company and strengthens our Board's collective qualifications, skills, experience and viewpoints.

Conclusion and Recommendation; Vote Required

The election of directors nominated in Proposal 1 requires the affirmative vote of a plurality of all of the votes cast on the proposal at the Annual Meeting. In the context of the election of two directors at the Annual Meeting, it will mean that the two candidates receiving the highest number of FOR votes cast will be elected.

A stockholder can vote for or withhold his or her vote from the Nominees. However, because directors are elected by a plurality vote, votes withheld from the director nominees and broker non-votes will have no effect on the outcome of the vote with respect to the election of directors. Abstentions and broker non-votes will not be counted as votes cast, although they will be considered present for the purpose of determining the presence of a quorum. THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR THE ELECTION OF THE NOMINEES NAMED ABOVE.

PROPOSAL 2: RATIFICATION OF APPOINTMENT OF INDEPENDENT ACCOUNTING FIRM

The audit committee of our Board (the "Audit Committee") has appointed Crowe Horwath LLP as our independent registered public accounting firm for the year ending December 31, 2016 and is submitting the selection of Crowe Horwath LLP to our stockholders for ratification. Crowe Horwath LLP has audited our financial statements for the fiscal years ended December 31, 2015, 2014 and 2013. Neither our bylaws nor other governing documents or applicable law require stockholder ratification of the Audit Committee's appointment of Crowe Horwath LLP as our independent registered public accounting firm. However, our Board is submitting the appointment of Crowe Horwath LLP to our stockholders for ratification as a matter of good corporate practice.

This vote is non-binding, as the Audit Committee has the sole authority to retain and dismiss the Company's independent registered public accounting firm. If our stockholders fail to ratify the appointment of Crowe Horwath LLP at the Annual Meeting, the Audit Committee will consider whether or not to appoint a different independent registered public accounting firm. Even if the appointment is ratified, the Audit Committee, in its discretion, may direct the appointment of a different independent registered public accounting firm at any time during the year if it determines that such a change would be in our best interests.

The Company expects that a representative of Crowe Horwath LLP will be present at the Annual Meeting, will have the opportunity to make a statement if so desired and will be available to respond to appropriate inquiries from stockholders.

Principal Accountant Fees and Services

The following table summarizes the aggregate fees (including related expenses) billed to us for professional services provided by Crowe Horwath LLP for the fiscal years ended December 31, 2015 and 2014:

		l Year Ended mber 31, 2015	cal Year Ended ember 31, 2014
Audit Food	¢ Decei	· ·	, and the second
Audit Fees	Ф	218,000	\$ 205,000
Audit-Related Fees		4,500	0
Tax Fees		47,075	0
All Other Fees		4,605	10,015
Total	\$	274,180	\$ 215,015

Audit Fees. Audit fees consist of fees and expenses billed by Crowe Horwath LLP related to the audit of our consolidated financial statements included in Form 10-K, the audit of our internal control over financial reporting and the reviews of the interim consolidated financial statements included in the Form 10-Q's, including services normally provided by an accountant in connection with statutory and regulatory filings or engagements.

Audit Related Fees. Audit-related fees are fees and expenses billed for assurance and related services that are reasonably related to the performance of the audit or review of the Company's financial statements and are not reported under "Audit Fees."

Tax Fees. Tax fees billed to us by Crowe Horwath LLP are for tax compliance and consulting fees, which typically consist of fees billed for professional services for tax return preparation and other tax compliance.

All Other Fees. All other fees consist of fees for products and services other than the services reported above, including consulting fees paid for a cost segregation study and accounting guidance subscriptions paid.

All services rendered by Crowe Horwath LLP were pre-approved by the Audit Committee for 2015 in accordance with its pre-approval policy, and the Audit Committee concluded that the provision of such services by Crowe Horwath LLP was compatible with the maintenance of that firm's independence in the conduct of its audit functions. The Audit Committee charter requires that the committee review and pre-approve each audit or permissible non-audit engagement or accounting project involving the independent public accountant, and the related fees or ranges of fees, prior to commencement of the engagement or project subject to certain exceptions if the services meet pre-approval policies that may be established under the charter. The Audit Committee may delegate its pre-approval authority to one or more of its members and, if such delegation occurs, then such member(s) are required to report any pre-approval decisions to the Audit Committee at its next-scheduled meeting.

Conclusion and Recommendation; Vote Required

The affirmative vote of a majority of all of the votes cast at the Annual Meeting is required to ratify the appointment of our independent public accountant. Abstentions will not be counted as votes cast and will have no effect on the result of the vote, although they will be considered present for the purpose of determining the presence of a quorum. THE BOARD OF DIRECTORS RECOMMENDS A VOTE FOR RATIFICATION OF THE APPOINTMENT OF CROWE HORWATH LLP AS OUR INDEPENDENT PUBLIC ACCOUNTANT FOR THE YEAR ENDING DECEMBER 31, 2016.

REPORT OF THE AUDIT COMMITTEE

The Board of Directors has appointed an Audit Committee presently composed of three directors, Messrs. Bates, Kessler and Schmal, who each served on the Audit Committee during 2015. Each of the committee members is independent as defined in the NYSE MKT rules, and the Board has determined that Mr. Bates is an "audit committee financial expert" (as defined in Item 407 of Regulation S-K under the Securities Act of 1933, as amended (the "Securities Act")).

The Audit Committee's responsibility is one of oversight as set forth in its charter, which is available on the Company's web site at www.owensmortgage.com. It is not the duty of the Audit Committee to prepare the Company's financial statements, to plan or conduct audits or to determine that the Company's financial statements are complete and accurate and are in accordance with generally accepted accounting principles. The Company's management is responsible for preparing the Company's financial statements and for maintaining effective internal control over financial reporting. The Company's independent registered public accounting firm is responsible for auditing the financial statements and the effectiveness of the Company's internal control over financial reporting and for expressing an opinion as to whether those audited financial statements present fairly, in all material respects, the Company's financial position, results of operations and cash flows in conformity with generally accepted accounting principles, and for expressing an opinion as to whether the Company maintained, in all material respects, effective internal control over financial reporting.

At a meeting held subsequent to December 31, 2015, the Audit Committee reviewed and discussed with management and Crowe Horwath LLP, the Company's independent registered public accounting firm, the audited financial statements of the Company as of and for the year ended December 31, 2015, and the related report prepared by Crowe Horwath LLP. The Audit Committee met with Crowe Horwath LLP, with and without management present, to discuss the results of their audit. Management represented to the Audit Committee that the Company's consolidated financial statements as of and for the year ended December 31, 2015 were prepared in accordance with accounting principles generally accepted in the United States and that the Company maintained effective internal control over financial reporting as of December 31, 2015.

The Audit Committee has discussed with Crowe Horwath LLP the matters required to be discussed by PCAOB Auditing Standard No. 16, which included among other things a discussion of Crowe Horwath LLP's judgments about the quality (not just the acceptability) of the Company's accounting principles as applied to financial reporting.

The Audit Committee has received from Crowe Horwath LLP the written statements required by PCAOB Rule No. 3526, "Communications with Audit Committees Concerning Independence," and has discussed Crowe Horwath LLP's independence with Crowe Horwath LLP, and has considered the compatibility of non-audit services with the auditor's independence. The Audit Committee also received regular updates on the amount of fees and scope of audit and other services provided by Crowe Horwath LLP.

Based on the review and discussions referred to above, and subject to the limitations on the Audit Committee's role and responsibilities referred to above and in its charter, the Audit Committee recommended to the Board of Directors that the audited consolidated financial statements as of and for the year ended December 31, 2015, be included in our Annual Report on Form 10-K for the year ended December 31, 2015, for filing with the SEC. The Audit Committee also appointed Crowe Horwath LLP as the Company's independent registered public accounting firm for the year ending December 31, 2016 and is presenting this appointment to the Company's stockholders for ratification.

By the Audit Committee: M. Lyman Bates, Jr., Chair James Matthew Kessler Dennis George Schmal

The foregoing Report of the Audit Committee shall not be deemed under the Securities Act or the Securities Exchange Act of 1934, as amended (the "Exchange Act"), to be (i) "soliciting material" or "filed" or (ii) incorporated by reference by any general statement into any filing made by us with the SEC, except to the extent that we specifically incorporate such report by reference.

GOVERNANCE OF OUR COMPANY

Pursuant to our Charter and bylaws and the MGCL our business and affairs are managed under the direction of our Board of Directors. Members of our Board keep informed of our business by participating in meetings of our Board and its committees, by reviewing analyses, reports and other materials provided to them and through discussions with our Manager and our executive officers. Our Board has developed corporate governance practices to help it fulfill its responsibility to ORM to oversee the work of management and the Manager in the conduct of our business. The governance practices are memorialized in a written set of corporate governance guidelines (the "Guidelines"), our Charter, bylaws, policies and committee charters, which form the framework for our governance and are further described below.

Board Leadership Structure

Our Board is currently comprised of five members and conducts its business through regular quarterly (or additional) meetings and through actions taken by written consent in lieu of meetings. During 2015, the Board held six meetings and each director attended at least 75% of the meetings of the Board and of the committees on which he served. Although we do not have a policy requiring director attendance at the Annual Meeting, directors are encouraged to attend the Annual Meeting.

Two of our directors, William C. Owens and Bryan H. Draper, are affiliated with the Manager. Mr. Owens serves as our Chairman and Mr. Draper serves as our Chief Executive Officer, and both are considered executive officers of the Company. While the roles are currently separated, our Board of Directors has not established a fixed policy regarding the separation of the roles of Chief Executive Officer and Chairman of the Board. Instead, the Board believes this determination is part of the succession planning process and should be considered upon the appointment or re-appointment of a Chief Executive Officer. The Board of Directors believes that the current separation of the roles of Chief Executive Officer and Chairman allows Messrs. Owens and Draper to have leadership roles on the executive management team, which our Board of Directors believes is important in light of their respective roles with our predecessor entities, their knowledge of the Company and their extensive experience in the mortgage finance and real estate industries. Our Board continues to believe that our current leadership structure, including separate positions of Chairman and Chief Executive Officer, provides an effective leadership model for the Company and the benefit of the distinct abilities and experience of both individuals. The Board also believes having a Chairman that is a member of the executive management team is useful as it ensures that Board leadership retains a close working relationship with management.

Our Board believes that its governance structure ensures a strong, independent Board even though the Board does not have an independent Chairman The Guidelines provide that a majority of the directors serving on our Board must be "independent" as defined by the NYSE MKT rules. Based upon its review of all relevant facts and circumstances, the Board has affirmatively determined that three of our five current directors, M. Lyman Bates Jr., Dennis G. Schmal and James M. Kessler, qualify as independent directors under the NYSE MKT rules. In addition, each of our Board's Audit Committee, Compensation Committee and Nominating and Corporate Governance Committee is composed entirely of independent directors as required by the charters of those committees. These independent committees of our Board also have the authority under their respective charters to hire independent advisors and consultants, at our expense, to assist them in performing their duties.

The independent Board members have decided to hold meetings periodically without the affiliated directors or other persons who are members of management present. Further, it is our policy, as a matter of good corporate governance and pursuant to the Guidelines, to have a majority of our independent directors elect a "lead independent director" to preside at these meetings. Presently, our independent directors meet during our Board's quarterly in-person meetings and may hold additional meetings at the request of the lead independent director or another independent director. The appointment of a lead independent director is generally for a one-year term or until his successor is appointed. Mr. Schmal is our current lead independent director.

The Board has established three standing committees: the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee. The three independent directors, Messrs. Bates, Kessler and Schmal, are the current members of each committee. Our Board has determined, based upon its qualitative assessment of his relevant level of knowledge and business experience that Mr. Bates (the chairman) qualifies as an "audit committee financial expert" for purposes of, and as defined by, applicable SEC rules and has the requisite accounting or related financial management expertise required by the NYSE MKT rules. In addition, our Board has determined that all of the members of the Audit Committee are financially literate as required by the NYSE MKT rules. Mr. Schmal currently serves on the audit committees of five public companies (including ours). In accordance with our Audit Committee charter and the Guidelines, these obligations have been disclosed to our Board. Our Board has determined that serving on the audit committee of four other public companies does not impair Mr. Schmal's ability to effectively serve on our Audit Committee. The primary functions of the committees are described below.

Audit Committee. This committee, which met five times during 2015, among other things acts on behalf of the Board of Directors in overseeing:

• our accounting and financial reporting processes;

- the integrity of our financial statements and the audit and reporting processes;
- our system of disclosure controls and procedures and internal control over financial reporting;
- our compliance with financial, legal and regulatory requirements;
- the qualifications and independence of our independent registered public accounting firm; and
- the Company's overall risk profile.

The Audit Committee is also responsible for engaging our independent registered public accounting firm, approving professional services provided by the independent registered public accounting firm and considering the range of audit and non-audit fees, and reviewing with management and the independent auditors our interim and audited annual financial statements as well as approving the filing of our financial statements. The Audit Committee's purpose and responsibilities are more fully set forth in the committee's charter, which was adopted by the Board of Directors on May 20, 2013, and is available on our web site at www.owensmortgage.com. The Audit Committee's meetings include, whenever appropriate, executive sessions with our independent external auditors without the presence of management.

Compensation Committee. This committee, which met five times during 2015, among other things acts on behalf of the Board of Directors to:

- discharge the Board's responsibilities relating to compensation to be paid by us to our directors which is considered on at least an annual basis;
- discharge the Board's responsibilities relating to compensation, if any, to be paid by us to our executive officers;
- oversee the compensation and fees payable to, and the performance of, the Manager under the Company's Charter and the existing management agreement dated May 20, 2013 (the "Management Agreement") with our Manager (which is described in this Proxy Statement under "Related Party Transactions"); and
- produce any report on executive compensation required to be included in our proxy statement for our annual meetings of stockholders.

The Compensation Committee's purpose and responsibilities are more fully set forth in the committee's charter, which was adopted by the Board of Directors on May 20, 2013, and amended and restated by the Board on January 4, 2016, and is available on our web site at www.owensmortgage.com.

In November 2015, as part of its regular review of the Manager's compensation under the Management Agreement and its annual review of independent director compensation, the Compensation Committee retained FTI Consulting, Inc. ("FTI") as its independent compensation consultant. FTI was retained to evaluate the terms and structure of the compensation by the Company of our Manager and of our independent directors. The Compensation Committee requested that FTI provide the Committee with a peer group analysis to compare the compensation terms for our Manager and for our independent directors against the compensation terms in place at a peer group of our competitors, such peer group to be developed by FTI. The Compensation Committee has also requested that, if appropriate based on this peer group analysis, FTI develop formal recommendations regarding the compensation of our Manager and of our independent directors. FTI's engagement is ongoing in 2016.

Nominating and Corporate Governance Committee. This committee, which met four times during 2015, among other things acts on behalf of the Board of Directors to:

- identify qualified director candidates, consistent with criteria approved by the Board and included in the committee's charter and to recommend nominees for election to the Board;
- develop and recommend to the Board corporate governance guidelines (including the Guidelines) and principles, and the implementation and monitoring of compliance with such corporate governance guidelines and principles;
- review matters involving the general operation of the Board, including Board size and composition and committee composition and structure;
- recommend the directors to serve on each committee of the Board;
- oversee the annual assessment of the Board and committee performances, and the performance of individual directors; and
- ensure the Company is in compliance with the NYSE MKT rules.

For a discussion of the consideration of diversity in the process by which candidates for director are considered for nomination by the Nominating and Corporate Governance Committee, and the process for identifying and evaluating nominees for director, including nominees recommended by stockholders, see "Governance of Our Company— Director Qualifications" in this Proxy Statement. The Nominating and Corporate Governance Committee's purpose and responsibilities are more fully set forth in the committee's charter, which was adopted by the Board of Directors on May 20, 2013, and is available on our web site at www.owensmortgage.com.

Board Role in Risk Oversight

Senior management and our Manager are responsible for the day-to-day management of risks we face. Our Board of Directors has overall responsibility for overseeing risk management with a focus on the more significant risks facing us. In this regard, the Board and its standing committees regularly meet with senior management and the Manager to review material strategic, operational, financial, compensation and compliance risks. The Board has also delegated certain risk management oversight responsibility to its Board committees.

Our Audit Committee assists the Board in fulfilling its oversight responsibilities for our accounting and financial reporting processes and the audits of our financial statements, as detailed in the discussion of its responsibilities above. The Audit Committee also is responsible for reviewing and discussing with management our major financial risk exposures and the steps management has taken to monitor and control such exposures, including our policies and practices with respect to risk assessment and risk management. The Audit Committee receives regular reports from management on the status of our internal controls and ensures that appropriate steps are being taken to mitigate any identified risk.

The Compensation Committee assists our Board by considering and addressing risk as it relates to compensation and benefit decisions, including through its review of our Manager's performance, compensation and the Management Agreement with our Manager, as detailed in the discussion of its responsibilities above. The Nominating and Corporate Governance Committee assists our Board by considering and addressing risk as it relates to corporate governance matters and other matters, as detailed in the discussion of its responsibilities above. The Compensation Committee, Nominating and Corporate Governance Committee and the Audit Committee routinely report their findings to the full Board at the next regularly scheduled Board meeting following their committee meetings and when appropriate.

The Company believes that the extent of the Board's (and its committees') role in risk oversight complements the Board's leadership structure because it allows the Company's independent directors, through the three fully independent Board committees, executive sessions with the independent auditors, and otherwise, to exercise oversight of risk without conflicts of interest that might discourage critical review.

Code of Business Conduct and Ethics

We have adopted a Code of Business Conduct and Ethics, which requires our directors, employees (if any), officers (including our Chief Executive Officer and Chief Financial Officer), and the personnel of our Manager, to abide by high standards of business conduct and ethics. The Code of Business Conduct and Ethics covers a variety of topics, including those required by the SEC and the NYSE MKT rules. Topics covered include, but are not limited to, conflicts of interest, confidentiality of information, and compliance with laws and regulations. The Code of Business Conduct and Ethics is available for viewing on our web site at www.owensmortgage, and a copy may also be obtained by stockholders, free of charge, by writing to us at Owens Realty Mortgage, Inc., 2221 Olympic Boulevard, Walnut Creek, CA 94595, Attention: Dan Worley, Secretary. The Code of Business Conduct and Ethics was adopted by the Board of Directors on May 20, 2013 and amended and restated on February 3, 2016.

Corporate Governance Guidelines

Our Board has adopted the Guidelines that address significant issues of corporate governance and set forth procedures and policies by which our Board of Directors carries out certain of its responsibilities. Among the areas addressed by the Guidelines are the composition of our Board, its functions and key responsibilities, its standing committees, director qualification standards, access to management and independent advisors, director compensation, management succession, director orientation and continuing education and the annual performance evaluation and review of our Board and its committees. The Guidelines are available for viewing on our website at www.owensmortgage.com. We will also provide the Guidelines, free of charge, to stockholders who request them by writing to us at Owens Realty Mortgage, Inc., 2221 Olympic Blvd, Walnut Creek, CA 94595, Attention: Dan Worley, Secretary. The Guidelines were adopted by the Board of Directors on May 20, 2013 and amended and restated by the Board on March 20, 2014.

Director Qualifications

The Board believes that it should be comprised of directors with complementary backgrounds, and that directors should, at a minimum, have expertise that will strengthen our Board's collective qualifications, skills, experience and viewpoints. Our Nominating and Corporate Governance Committee has not adopted a formal diversity policy in connection with the consideration of director nominations or the selection of nominees. It considers the personal and professional attributes and the business experience of each director candidate to promote diversity of expertise and experience among our directors. Additionally, directors should possess the highest personal and professional ethics and should be willing and able to devote the required amount of time to our business.

Although there is not a formal list of required qualifications governing the suitability of individual candidates (both new candidates and current Board members), the Nominating and Corporate Governance Committee, in recommending candidates for election, and the Board, in approving (and, in the case of vacancies, appointing) such candidates, will take into account many factors, including the following:

- the independence of the candidate from the Manager and whether the candidate meets the criteria for membership of the audit or other committees;
 - personal and professional integrity, ethics and values;
- experience in corporate management, such as serving as an officer or former officer of a publicly held company, and a general understanding of marketing, finance and other elements relevant to the success of a publicly traded company in today's business environment;
 - experience in the Company's industry and with relevant social policy concerns;
 - experience as a board member of another publicly held company;
 - diversity (of both background and experience);
 - academic expertise in an area of the Company's operations; and
 - practical and mature business judgment, including the ability to make independent analytical inquiries.

When seeking candidates for director, the Nominating and Corporate Governance Committee may solicit suggestions from management, incumbent directors or others. While there is no formal policy for considering candidates that are proposed by stockholders, candidates proposed in writing by stockholders entitled to vote in the election of directors will be considered and evaluated by the Nominating and Corporate Governance Committee under the same criteria applied to other candidates. The committee or its chairman will interview a candidate if it is believed the candidate might be suitable to be a director. The Nominating and Corporate Governance Committee may also ask the candidate to meet with management. If the Nominating and Corporate Governance Committee believes a candidate would be a valuable addition to the Board, it will recommend the candidate's election to the full Board.

Stockholder Communications with Directors

Our Board has adopted an Investor Relations Policy, available on its website at www.owensmortgage.com, that establishes a process by which stockholders or other interested parties may communicate in writing with our full Board, any individual member or any committee or other group of our Board, or our independent directors as a group. Any such communications should be addressed to the Board or any such individual directors or group or committee of directors by either name or title (provided that communications intended for the independent directors as a group may be addressed to the independent directors as a group or to the lead independent director, who is currently Dennis G. Schmal) and addressed to Owens Realty Mortgage, Inc., 2221 Olympic Blvd., Walnut Creek, CA 94595, Attention: Investor Relations.

All communications to directors received as set forth in the Investor Relations Policy will be opened by the office of the Company's Corporate Compliance Officer for the sole purpose of determining whether the contents represent a message to the Company's directors. Any contents that are not in the nature of advertising, promotions of a product or service or patently offensive material will be forwarded promptly to the addressee. In the case of communications to any individual director, the full Board of Directors, or any group or committee of directors, the office of the Corporate Compliance Officer will make sufficient copies of the contents to send to each addressee. Any such communications may be made anonymously. Communications regarding accounting matters may be sent in writing to the Company's main address, attention Chief Financial Officer, or by calling the Company's telephone hotline: (855) 380-8849.

Related Party Transaction Policy and Procedures

The Company has adopted a written policy for approval of transactions and arrangements between the Company and the Company's current and recent former directors, director nominees, current and recent former executive officers,

greater than five percent stockholders and their immediate family members, and entities where any of the foregoing persons is employed or serves as a general partner, principal or in a similar position (each, a "Related Party").

The policy provides that the Audit Committee reviews certain transactions subject to the policy and determines whether or not to approve or ratify those transactions. In doing so, the Audit Committee takes into account, among other things, whether the transaction is on terms comparable to those that could be obtained in arm's length dealings with an unrelated third party under similar circumstances, the extent of the Related Party's interest in the transaction, and the conflicts of interest provisions and corporate opportunity provisions of the Company's Code of Business Conduct and Ethics. The Related Party transaction must be approved or ratified by the Audit Committee in accordance with the provisions of the policy and in accordance with relevant provisions of the Company's charter, bylaws and applicable provisions of the MGCL.

The Audit Committee has considered and adopted standing pre-approvals under the policy for certain limited transactions with Related Parties that meet specific criteria. Information on transactions subject to pre-approval is to be provided to the Audit Committee at its next regularly scheduled meeting. Pre-approved transactions are limited to:

- compensation to an executive officer or director of the Company if (a) the related compensation is required to be reported in the Company's proxy statement under the SEC's compensation disclosure requirements or (b) the executive officer is not an immediate family member of another executive officer or director of the Company, the related compensation would be reported in the Company's proxy statement under the SEC's compensation disclosure requirements if the executive officer was a "named executive officer," and the Compensation Committee approved (or recommended that the Board approve and the Board has approved) such compensation;
- certain transactions in the Company's ordinary course of business where the Related Party's interest arises only from: (a) the Related Party's position as a director of another entity that is a party to the transaction; (b) from direct or indirect ownership by such Related Party and all other Related Parties, in the aggregate, of less than a 5% equity interest in another person (other than a partnership) that is a party to the transaction; (c) from both such position described in (a) and ownership described in (b); or (d) from the Related Party's position as a limited partner in a partnership in which all Related Parties in the aggregate have an interest of less than 5% and the Related Party is not a general partner of and does not have another position in such partnership;
- certain transactions in the Company's ordinary course of business where the Related Party's interest arises solely from the ownership of the Company's Common Stock and holders of the Company's Common Stock received the same benefit on a pro rata basis; and
- certain transactions involving a Related Party where the rates or charges involved are determined by competitive bids, or transactions involving the rendering of services as a common or contract carrier, or as a public utility, at rates or charges fixed in conformity with law or a governmental authority.

Compensation Committee Interlocks And Insider Participation

Messrs. Schmal, Kessler and Bates served as members of our Compensation Committee during 2015 and none of them served as an officer, former officer or employee of ours or had a relationship disclosable under Item 404 of Regulation SK. Further, during 2015, none of our executive officers served as a member of the board of directors or compensation committee (or equivalent) of any other entity, one of whose executive officers served as one of our directors or on our Compensation Committee.

INFORMATION REGARDING OUR EXECUTIVE OFFICERS

The Board of Directors generally elects officers annually following our annual meeting of stockholders to serve until the meeting of the Board following the next annual meeting. Set forth below is certain information about each executive officer as of May 6, 2016. The business address of each executive officer is c/o Owens Realty Mortgage, Inc., 2221 Olympic Blvd., Walnut Creek, CA 94595.

Name	Age	Information about Executive Officers
William C. Owens	65	Mr. Owens has been our executive Chairman of the Board since ORM's inception in 2012. Mr. Owens also served as our President from our inception in 2012 until January 2016, and as our Chief Executive Officer from January 2013 until January 2016. He has also been President and Chief Executive Officer of the Manager since April 1996 and is a member of the Board of Directors and the Loan Committee of the Manager. Further information about Mr. Owens may be found under "Proposal 1: Election of Two (2) Directors - Information about the Director Nominees and the Continuing Directors" in this Proxy Statement.
Bryan H. Draper	58	Mr. Draper has served as our President and CEO since January 2016 and as a member of our Board since the consummation of the Merger on May 20, 2013. Mr. Draper served as the Secretary and Treasurer of the Company from its inception in 2012 until January 2016, and as our Chief Financial Officer from January 2013 to January 2016. He has also been Chief Financial Officer and Corporate Secretary of the Manager since December 1987 and a member of the Board of Directors of the Manager since January 1997. Further information about Mr. Draper may be found under "Proposal 1: Election of Two (2) Directors -Information about the Director Nominees and the Continuing Directors" in this Proxy Statement.
William E. Dutra	53	Mr. Dutra has served as an Executive Vice President of the Manager since March 2014 and member of the Board of Directors and the Loan Committee of the Manager since January 1997. Mr. Dutra previously served as a Senior Vice President of the Manager and has been one of its employees since February 1986. Mr. Dutra has responsibility for loan committee review, loan underwriting and loan production.
Melina A. Platt	49	Ms. Platt was appointed as the Company's Chief Financial Officer and Treasurer in January 2016, and has been the Controller of the Manager since May 1998. Ms. Platt is a certified public accountant and is responsible for all accounting, finance, and regulatory agency filings of the Company. Ms. Platt was previously a Senior Manager with KPMG LLP.
Daniel J. Worley	45	Mr. Worley has served as a Senior Vice President of the Manager since June 2013 and is responsible for capital markets, investor relations, regulatory compliance, and governance. In addition, he is a member of the

senior management team and participates in executive management and strategy. Mr. Worley has served as Secretary of the Company since January 2016, and was Assistant Secretary of the Company from June 2013 until January 2016. Prior to joining the Company, from July 2012 to June 2013, he served as a consultant to Owens Mortgage Investment Fund. From July 2010 to October 2011, he was the Chief Strategy Officer at Mason-McDuffie Real Estate where his responsibilities included executive management, strategy, corporate development, governance, and legal matters. From May 2001 to October 2010, he co-founded and served as the Chief Operating Officer of NorthPoint Financial Group/SmartZip Analytics where his responsibilities included executive management, strategy and governance. From January 1998 to May 2001, he was the Managing Director of the Berkeley Center for Advanced Technology where his responsibilities included venture fund management, investment due diligence, investment oversight and investor relations.

Brian M. Haines

Mr. Haines has served as a Senior Vice President of the Manager since January 2014 and as a Loan Officer of the Manager since 2007. Mr. Haines' focus is on originating and structuring debt investments, developing and fostering broker relationships and asset management oversight. Prior to joining the Manager, Mr. Haines co-owned and operated a private lending firm headquartered in Northern California. He is a director of the Bay Area Mortgage Association (BAMA) and an active member of the Urban Land Institute (ULI) and California Mortgage Association (CMA).

EXECUTIVE COMPENSATION

Compensation Discussion and Analysis

38

We have no employees other than six full-time and one part-time employees (none of whom is an officer) that work for two of our subsidiaries. We are managed by our Manager pursuant to the terms of our Charter and the Management Agreement. See "Related Party Transactions" in this Proxy Statement for a description of certain relevant terms of the Charter and the Management Agreement including the management and other fees payable to our Manager thereunder and our expense reimbursement obligations to our Manager.

We do not have agreements with any of our executive officers or any employees of our Manager with respect to their compensation. Our executive officers are employees of our Manager and do not receive cash, equity or other compensation from us for serving as our executive officers. We did not pay any compensation to our executive officers nor did we make any grants of equity or other plan-based awards of any kind to them during 2015 or through the date of this Proxy Statement. None of our executive officers received any options or stock directly from us prior to the date of this Proxy Statement. We do not provide any of our executive officers with pension benefits or nonqualified deferred compensation plans. We do not have any employment agreements with any persons and are not obligated to make any payments to any of our executive officers upon termination of employment or a change in control of us.

COMPENSATION COMMITTEE REPORT

Our Compensation Committee reviewed and discussed with our management the "Compensation Discussion and Analysis" contained in this Proxy Statement. Based on that review and discussions, our Compensation Committee recommends to the Board of Directors that the "Compensation Discussion and Analysis" be included in this Proxy Statement.

By the Compensation Committee: Dennis George Schmal, Chair James Matthew Kessler M. Lyman Bates, Jr.

The foregoing Compensation Committee Report shall not be deemed under the Securities Act or the Exchange Act to be (i) "soliciting material" or "filed" or (ii) incorporated by reference by any general statement into any filing made by us with the SEC, except to the extent that we specifically incorporate such report by reference.

COMPENSATION OF INDEPENDENT DIRECTORS

Each of our independent directors, Messrs. Bates, Kessler and Schmal, were appointed to the Board of Directors on May 20, 2013 in connection with the completion of the Merger. In 2015, each of our independent directors was paid a cash retainer at an annual rate of \$20,000, payable in advance in quarterly installments of \$5,000. In addition, in 2015, Mr. Bates was paid an additional retainer at the annual rate of \$5,000 for services as the Chairman of the Audit Committee, and Mr. Schmal was paid an additional retainer at the annual rate of \$5,000 in aggregate for services as the Chairman of both the Compensation Committee and Nominating and Corporate Governance Committee, which retainers are payable in advance in semi-annual installment on the first day of January and July of each year. For 2016 we have continued to pay the same annual cash retainers that we paid in 2015 to our independent directors. We also reimburse all members of our Board for their travel expenses incurred in connection with their attendance at Board, committee and stockholder meetings. We pay directors' fees only to those directors who are independent under the NYSE MKT rules.

The following table summarizes the compensation received by our independent directors for the fiscal year ended December 31, 2015.

	Fees Earned or	
Name	Paid in Cash(\$)	
M. Lyman Bates, Jr.	\$25,000	
James M. Kessler	\$20,000	
Dennis G. Schmal	\$25,000	

SECURITY OWNERSHIP OF MANAGEMENT AND CERTAIN BENEFICIAL OWNERS

The following table sets forth certain information regarding the beneficial ownership of the Common Stock as of May 6, 2016 (unless otherwise indicated), of each current director and director nominee, each of our executive officers, our executive officers and directors as a group and each stockholder known to management to own beneficially more than 5% of the outstanding shares of our Common Stock. Unless otherwise indicated, we believe that the beneficial owner set forth in the table has sole voting and investment power.

	Number of Shares	Paraentogo of Common
Name and Address of Beneficial	Beneficially	Percentage of Common Stock Beneficially
Owner(1)	Owned	Owned
Beneficial owners of more than 5%:	Owned	Owned
Freestone Investments, LLC		
Freestone Capital Management, LLC		
Gary Furukawa(2)		
701 Fifth Avenue, Suite 7400		
Seattle, Washington 98104	850,049	8.30%
Nantahala Capital Management,	22 2,4 12	
LLC(3)		
19 Old Kings Highway South,		
Suite 200		
Darien, CT 06820	1,060,724	10.35%
Executive officers and directors:		
William C. Owens (4)(5)	179,985	1.76%
Bryan H. Draper(6)	88,292	*
James M. Kessler(7)	29,074	*
Dennis G. Schmal	3,000	*
M. Lyman Bates, Jr.	300	*
Daniel J. Worley	2,000	*
William E. Dutra(8)	20,589	*
Melina A. Platt	3,910	*
Brian M. Haines(9)	7,908	*
All executive officers and directors as a		
group (9 persons)	335,058	3.27%
Y 1		
* Less than one percent.		

⁽¹⁾ The address of each of the executive officers and directors listed above is c/o Owens Realty Mortgage, Inc., 2221 Olympic Boulevard, Walnut Creek, California 94595.

- (2) This information is based on a Schedule 13G/A filed with the SEC on February 3, 2016. An aggregate of 850,049 shares of Common Stock are beneficially owned by Freestone Investments, LLC, a Washington limited liability company ("FI"), Freestone Capital Management, LLC, a Delaware limited liability company ("FCM"), and Gary I. Furukawa ("Furukawa", and collectively, the "Reporting Persons"). FI is the general partner of Freestone Opportunity Partners LP ("FOP") and Freestone Opportunity Qualified Partner LP ("FOQP"), and reports beneficial ownership of 284,985 shares of Common Stock held directly by FOP and 396,208 shares of Common Stock held directly by FOQP. FCM is the investment manager for FOP and FOQP, as well as various managed accounts that own Common Stock and reports beneficial ownership of the 284,985 shares of Common Stock held directly by FOP, the 396,208 shares of Common Stock held directly by FOQP and an additional 168,856 shares of Common Stock held in managed accounts. Furukawa reports beneficial ownership of an additional 118,445 shares of Common Stock and has reported he may be deemed a beneficial owner of the shares of Common Stock held by FI and FCM due to his directorship of Freestone Capital Holding, LLC which directly or indirectly owns all the equity of FI and FCM.
- (3) This information is based on a Schedule 13G filed with the SEC on January 8, 2016, by Nantahala Capital Management, LLC, a Massachusetts limited liability company ("Nantahala"), Wilmot Harkey and Daniel Mack (collectively the "Reporting Persons"). The Reporting Persons report that Nantahala may be deemed to be beneficial owner of 1,060,724 shares of Common Stock held by funds and separately managed accounts under its control and, as the managing members of Nantahala, each of Messrs. Harkey and Mack may also be deemed the beneficial owner of those shares.
- (4) Mr. Owens owns 64.389% of Owens Financial Group, Inc., our Manager and shares voting power at the Manager with Mr. Draper, Mr. Dutra and Mr. Haines, each of whom owns the following percentage of the Manager: Draper 16.798%; Dutra 16.798%; and Haines 2.015%. Mr. Owens' is reporting beneficial ownership of 83,049 shares of Common Stock held by the Manager directly and 13,736 shares of Common Stock held by Investors Yield. Inc. (a wholly-owned subsidiary of the Manager), with respect to which he has shared voting and investment power. Mr. Owens disclaims beneficial ownership of these securities except to the extent of his pecuniary interest therein.
- (5) Includes: 360 shares of Common Stock held by Owens Trust dated February 24, 1998, the trustee of which is Mr. Owens. Also includes 4,637 shares of Common Stock held by Belmar, a California limited partnership of which Mr. Owens owns 49.22%. Mr. Owens disclaims beneficial ownership of these securities except to the extent of his pecuniary interest therein. Also includes 6,361 shares of Common Stock held indirectly by Mr. Owens spouse and 57,460 shares of Common Stock owned indirectly through the Owens Financial Group 401(k) Plan.
- (6) Includes 4,543 shares of Common Stock held by Draper Family Partnership of which Mr. Draper is a 50% owner. Mr. Draper disclaims beneficial ownership of these securities except to the extent of his pecuniary interest therein. Also includes: (i) 32,713 shares of Common Stock held by Draper Family Trust dated May 16, 2000 of which Mr. Draper is co-trustee; (ii) 3,275 shares of Common Stock indirectly by Mr. Draper's spouse; (iii) 47,461 shares of Common Stock held in Individual Retirement Accounts of which Mr. Draper is sole beneficiary; and (iv) 300 shares of Common Stock owned by Mr. Draper's child. Does not include 83,049 shares of Common Stock held by the Manager directly and 13,736 shares of Common Stock held by Investors Yield, Inc.
- (7) Mr. Kessler owns 29,074 shares of Common Stock through two trusts. Mr. Kessler has sole voting and investment power over 17,836 shares of Common Stock held in one of those trusts and shared voting

and investment power over shares of Common Stock held in the other trust.

- (8) Includes 2,689 shares of Common Stock held by The Dutra Trust of which Mr. Dutra is a co-trustee and 17,900 shares of Common Stock owned indirectly through the Owens Financial Group 401(k) Plan. Does not include 83,049 shares of Common Stock held by the Manager directly and 13,736 shares of Common Stock held by Investors Yield. Inc.
- (9) Includes 1,248 shares of Common Stock held in an Individual Retirement Account and 3,973 shares of Common owned indirectly through the Owens Financial Group 401(k) Plan. Does not include 83,049 shares of Common Stock held by the Manager directly and 13,736 shares of Common Stock held by Investors Yield. Inc.

RELATED PARTY TRANSACTIONS

Ownership of the Manager

The voting common stock of our Manager, Owens Financial Group, Inc., is owned as follows: 64.389% by William C. Owens, 16.798% by Bryan H. Draper, 16.798% by William E. Dutra and 2.015% by Brian M. Haines.

Management Fees and Expenses

We have entered into a Management Agreement with our Manager, Owens Financial Group, Inc., which describes the services to be provided by our Manager.

The compensation payable to the Manager for those services and certain other important provisions relating to the Manager are described in our Charter. The Manager manages our day-to-day operations and business, subject to the supervision and oversight of our Board of Directors. The Manager is required to act in accordance with policies and restrictions contained in the Company's Charter and such additional investment policies as may be adopted by our Board of Directors.

The Management Agreement continues in effect for the duration of the existence of the Company, unless earlier terminated by the affirmative vote of the holders of a majority of the outstanding shares of Common Stock, by the Manager in response to an amendment to its compensation that it does not consent to, automatically in certain circumstances relating to the assignment of the Management Agreement, or by either party for cause as defined therein.

The Management Agreement provides that the fees payable to the Manager as described in the Charter may not be changed without the approval of Board (including a majority of the independent members of the Board), the holders of a majority of the outstanding shares of Common Stock and the Manager; provided that stockholder approval is not required to adjust the Manager's compensation so long as such adjustment will not have a significant adverse impact on the stockholders of the Company. The Charter provides for the payment of the following fees by the Company and by borrowers:

Fees paid by the Company. The Manager is entitled to receive the following fees from the Company:

- Management Fee. The Manager receives a management fee payable monthly, subject to a maximum fee of 2.75% per year of the average unpaid balance of the Company's mortgage loans at the end of each month in the calendar year. The maximum payment is calculated on an annual basis; thus, the management fee in any one month could exceed .2292% (2.75% / 12 months) of the unpaid balance of Company's mortgage loans at the end of such month, provided that the maximum annual management fee will not exceed 2.75% of the average unpaid balance the Company's mortgage loans at the end of each month in the calendar year. Management fees amounted to approximately \$2,051,000 and \$1,727,000 for the years ended December 31, 2015 and 2014, respectively.
- Loan Servicing Fee. The Manager may act as servicing agent with respect to the Company's mortgage loans, in consideration for which it will be entitled to receive from ORM a monthly fee, which, when added to all other fees paid in connection with the servicing of a particular loan, does not exceed the lesser of the customary, competitive fee in the community where the loan is placed for the provision of such mortgage services on that type of loan, or up to 0.25% per year of the unpaid balance of ORM's mortgage loans at the end of each month. Servicing fees amounted to approximately \$186,000 and \$157,000 for the years ended December 31, 2015 and 2014, respectively.

Fees paid by Borrowers. The Manager is entitled to receive directly from borrowers the following fees:

• Acquisition and Origination Fees. The Manager or its affiliates is entitled to receive and retain all fees and commissions paid or payable to it by any party other than ORM and any subsidiary in connection with ORM making or investing in mortgage loans. Included in the computation of such fees or commission is any selection fee, mortgage placement fee, nonrecurring management fee and any origination fee, loan fee or points paid by borrowers to the Manager or any fee of a similar nature, however designated. Such fees earned by the Manager amounted to approximately \$1,956,000 and \$1,228,000 on loans originated or extended of approximately \$80,488,000 and \$50,440,000 for the years ended December 31, 2015 and 2014, respectively.

- Late Payment Charges. The Manager is entitled to receive and retain all additional charges paid by borrowers on delinquent loans and loans past maturity held by ORM, including additional interest and late payment fees. The amounts paid to or collected by the Manager for late payment charges totaled approximately \$30,000 and \$14,000 for the years ended December 31, 2015 and 2014, respectively.
- Other Miscellaneous Fees. ORM remits other miscellaneous fees to the Manager, which are collected from loan payments, loan payoffs or advances from loan principal (i.e. funding, demand and partial release fees). Such fees remitted to the Manager totaled approximately \$7,000 and \$4,000 for the years ended December 31, 2015 and 2014, respectively.

The Manager is entitled to be reimbursed by ORM for any expenses (subject to certain exceptions outlined in the Charter and the Management Agreement) paid by the Manager, including, without limitation, legal and accounting expenses, filing fees, printing costs, and goods, services and materials used by or for ORM. Additionally, the Manager is entitled to reimbursements for salaries for non-management and non-supervisory services. For the years ended December 31, 2015 and 2014, ORM recorded expenses totaling approximately \$590,000 and \$704,000, respectively, related to expense reimbursements to the Manager. ORM also reimbursed certain of the Manager's officers for allowed expenses in the total amount of \$1,000 and \$1,000 during the years ended December 31, 2015 and 2014, respectively..

ORM paid Investor's Yield, Inc. (a wholly owned subsidiary of the Manager) approximately \$10,000 and \$30,000 in trustee's fees related to certain foreclosure proceedings on ORM loans during the years ended December 31, 2015 and 2014, respectively.

TOTB Miami

Ownership Interest of the Manager in TOTB. TOTB Miami, LLC, a Florida limited liability company ("TOTB Miami"), was formed to own and operate a condominium complex located in Miami, Florida that was acquired through foreclosure and a series of related transactions. The condominium complex consists of three buildings on 5.41 acres of land, two of which have been previously renovated and in which 169 units remain unsold and one of which contains 160 vacant units that are currently undergoing renovation. During 2014 TOTB North, LLC, a Florida limited liability company ("TOTB North"), was formed as a wholly-owned subsidiary of TOTB Miami and the vacant 160 unit apartment building (the "Apartments") and the related 2.37 acre parcel of land were contributed by TOTB Miami to TOTB North.

The Company owns an 80.74% interest in TOTB Miami and the Manager is the owner of the other 19.26% interest. Income and loss allocations are now made based on these percentages. ORM and the Manager earned net income from TOTB Miami of approximately \$311,000 and \$74,000, respectively, during 2015.

TOTB North Construction Loan and Guaranty. On June 12, 2014, TOTB North entered into a secured Construction Loan Agreement with Bank of the Ozarks ("Ozarks") as the lender providing Borrower with a loan (the "North Loan") of up to \$21,304,000, for the purpose of renovating and improving the Apartments. As a condition to providing the North Loan, Ozarks required a joint and several completion guaranty from the Company and the Manager with respect to completion of the Apartments, a joint and several repayment guaranty from the Company and the Manager that guarantees repayment of the North Loan subject to certain limitations, and a joint and several carve-out guaranty from the Company and the Manager that provides a guaranty with respect to standard "bad-boy" carve-out provisions.

The largest outstanding principal balance of the North Loan during 2015 was \$16,010,000, the outstanding balance at December 31, 2015 was \$16,010,000 and \$19,256,000 at May 6, 2016, and the Company made no principal payments during 2015. The interest rate on the North Loan is the floating daily Three Month LIBOR rate plus four percent (4.0%) per annum, with a floor rate of four and one-half percent (4.5%). The interest rate as of May 6, 2016 was

4.63% per annum, and interest expense incurred on the loan totaled \$278,000 in 2015 (advanced from loan and capitalized to the construction project). During 2015, the Company and the Manager contributed an additional \$1,170,000 and \$279,000, respectively, due to increased construction costs for the Project.

TOTB Miami Loan and Guaranty. On November 17, 2014, TOTB Miami entered into a Loan Agreement with Ozarks providing TOTB Miami a loan (the "TOTB Loan") of \$13,000,000. As a condition to providing the TOTB Loan, Ozarks required a joint and several repayment guaranty from the Company and the Manager that guarantees repayment of the TOTB Loan subject to certain limitations, and a joint and several carve-out guaranty from the Company and the Manager that provides a guaranty with respect to standard "bad-boy" carve-out provisions.

The net cash proceeds from the TOTB Loan were distributed to the members of TOTB Miami (\$10,256,000 to ORM and \$2,446,000 to the Manager) in 2014. The largest outstanding principal balance of the TOTB Loan during 2015 was \$12,952,000, the outstanding balance at December 31, 2015 was \$12,693,000 and \$12,571,000 at May 6, 2016, and the Company made aggregate principal payments of \$282,000 during 2015. The interest rate on the TOTB Loan is the floating daily Three Month LIBOR rate plus four percent (4.0%) per annum, with a floor rate of four and one-quarter percent (4.25%). The interest rate as of May 6, 2016 was 4.63% per annum, and interest expense incurred on the loan totaled \$690,000 in 2015 (including \$129,000 in amortization of deferred financing costs).

TOTB North Construction Agreements. During 2014 and 2015, TOTB North entered into various contracts for the design, engineering, first phase demolition and concrete remediation and second phase construction for the renovation of the Apartments at TOTB North in the aggregate amount of approximately \$21,289,000 of which approximately \$18,205,000 had been incurred as of December 31, 2015. In addition, another \$1,969,000 in renovation-related costs, interest, property taxes, and amortization of deferred financing costs have been capitalized (total of \$20,174,000) as of December 31, 2015.

Agreement to Sell TOTB Properties. As previously reported, on March 16, 2016, the TOTB entities entered into a Purchase Agreement and Deposit Receipt with Interwest Capital Corporation (the "Buyer"), pursuant to which the TOTB entities have agreed to sell all real estate properties and related assets owned by the TOTB entities to the Buyer for a total sales price of \$82,000,000, subject to potential price and other adjustments. Buyer's obligation to purchase the TOTB property is subject to a number of conditions and there is no guarantee of the final purchase price or that the transaction will close. OFG will have a 19.26% interest in the final proceeds of the sale, but the amount of such proceeds will not be determined until a closing occurs and will be substantially reduced by the amount of final closing costs, required debt repayments and other items.

Guaranty of Opus Bank Line of Credit

On April 22, 2014, the Company entered into the Opus Credit Facility with Opus Bank ("Opus") to provide the Company with a revolving line of credit of up to \$20 million subject to the terms and conditions of this facility. As a condition to providing the Opus Credit Facility, Opus required the Company's Chairman of the Board, William C. Owens, to enter into a Carveout Payment Guaranty (the "Guaranty"), dated April 22, 2014, in favor of Opus. Mr. Owens' delivered the Guaranty in his individual capacity and as sole trustee of Owens Trust, a California trust controlled by Mr. Owens, to guarantee performance by the Company of certain specified obligations under the Opus Credit Facility.

The largest outstanding principal balance of the Opus Credit Facility during 2015 was \$12,626,000, the outstanding balance at December 31, 2015 was \$12,626,000 and there was \$8,594,000 outstanding at May 6, 2016. The interest rate on the Opus Credit Facility is reset every six months to the Six Month LIBOR rate of interest on such rate change date plus four percent (4.0%) per annum but in no event will the interest rate be lower than 4.5% per annum. The interest rate as of May 6, 2016 was 4.88% per annum, and interest expense incurred on the line of credit totaled \$126,000 in 2015 (including \$77,000 in amortization of deferred financing costs).

Office Lease

ORM does not have any separate offices. The Manager operates from its executive offices at 2221 Olympic Boulevard, Walnut Creek, California 94595, or the Executive Office. The lessor of the Executive Office is Olympic Blvd. Partners, a California General Partnership ("OBP"), of which the Manager is a 50% general partner. The Executive Office is the sole asset of OBP. The lease agreement, or the Office Lease, between OBP and the Manager for the Executive Office has a term extending until December 31, 2019. The Manager pays rent in the amount of \$15,975 per month to OBP under the Office Lease (increasing to the prevailing market rate beginning January 1, 2017). For each of the years ended December 31, 2015 and 2014, Owens Financial Group paid \$191,700 and \$170,400 to OBP for use of the Executive Office. The Executive Office is subject to a deed of trust in the amount of \$670,800 as of December 31, 2015 with monthly payments of interest and principal of \$4,565.88 and the remaining principal balance of \$617,013 due on February 28, 2018.

SECTION 16(a) BENEFICIAL OWNERSHIP REPORTING COMPLIANCE

Section 16(a) of the Exchange Act requires our directors, executive officers and holders of more than 10% of the outstanding shares of Common Stock ("10% Holders") to file with the SEC and the NYSE initial reports of ownership and reports of changes in ownership of Common Stock and other equity securities of ORM. Directors, executive officers and 10% Holders are required by the SEC's regulations to furnish us with copies of all Section 16(a) forms and amendments thereto filed during any given year.

Based on the review of copies of the Section 16(a) reports and amendments thereto furnished to us and/or written representations from our directors, executive officers and 10% Holders, we believe that for the year ended December 31, 2015 our directors, executive officers and 10% Holders complied with all Section 16(a) filing requirements applicable to them, except that Mr. Draper inadvertently failed to timely file a Form 4 reporting the purchase of 100 shares of Common Stock of the Company through his family trust.

PROPOSALS OF STOCKHOLDERS

Proposals received from stockholders in accordance with Rule 14a-8 under the Exchange Act will be given careful consideration by our Nominating and Corporate Governance Committee and our Board of Directors. If a stockholder intends to present a proposal at the 2017 annual meeting of stockholders pursuant to Rule 14a-8 under the Exchange Act, in order for such stockholder proposal to be included in our proxy statement for that meeting, the stockholder proposal must be received by our Secretary at Owens Realty Mortgage, Inc., on or before January 17, 2017. If such proposal is in compliance with all of the requirements of Rule 14a-8 under the Exchange Act, the proposal will be included in our proxy statement and proxy card relating to such annual meeting. Such proposals should be submitted by certified mail, return receipt requested. Nothing in this paragraph shall be deemed to require us to include any stockholder proposal that does not meet all the requirements for such inclusion established by the SEC in effect at that time.

In order for a stockholder proposal submitted outside of Rule 14a-8, including any proposal of business or nominations for the Board of Directors made by stockholders, to be considered at the 2017 annual meeting of stockholders, our bylaws require that such proposal must be made by written notice (setting forth the information required by our current bylaws) and received by our Secretary not earlier than the 150th day nor later than 5:00 p.m., Pacific Time, on the 120th day prior to May 17, 2017. Accordingly, to submit a director candidate for consideration for nomination at our 2017 annual meeting of stockholders, stockholders must submit the recommendation, in writing and in accordance with our bylaws by January 17, 2017, but in no event earlier than December 18, 2016.

DELIVERY AND HOUSEHOLDING OF PROXY MATERIALS

The rules of the SEC permit companies and intermediaries (such as brokerage firms, banks, broker-dealers or other similar organizations) to satisfy the delivery requirements for proxy statements and annual reports with respect to two or more stockholders sharing the same address by delivering a single set of proxy materials (i.e. the proxy statement and annual report) addressed to those stockholders. This process, which is commonly referred to as "householding," potentially means extra convenience for stockholders and cost savings for companies.

A number of brokers with account holders who are our stockholders will be "householding" our proxy materials. A single set of proxy materials may be delivered to multiple stockholders sharing the same address unless contrary instructions have been received from the impacted stockholders. Once a stockholder has received notice from its broker that they will be "householding" communications to such stockholder's address, "householding" will continue until such stockholder revokes consent to "householding" or is notified otherwise. If, at any time, a stockholder no longer wishes to participate in "householding" and would prefer to receive a separate set of our proxy materials, such stockholder should so notify us by directing written requests to: Owens Realty Mortgage, Inc., 2221 Olympic Blvd., Walnut Creek, CA 94595, Attn: Dan Worley, our Secretary or by calling our Investor Relations Department at (925) 935-3840. In addition, if so requested, we will also undertake to promptly deliver a separate set of proxy materials to any stockholder for whom such proxy materials were subject to "householding." Stockholders who currently receive multiple copies of our proxy materials at their address and would like to request "householding" of their communications should contact us as specified above or their respective brokers.

FINANCIAL STATEMENTS AND OTHER SEC FILINGS AVAILABLE

A COPY OF OUR 2015 ANNUAL REPORT ON FORM 10-K CONTAINING AUDITED FINANCIAL STATEMENTS WAS DELIVERED OR MADE AVAILABLE WITH THIS PROXY STATEMENT. ADDITIONAL COPIES OF OUR 2015 ANNUAL REPORT ON FORM 10-K (WITHOUT EXHIBITS, UNLESS OTHERWISE REQUESTED) ARE AVAILABLE IN PRINT, FREE OF CHARGE, TO STOCKHOLDERS REQUESTING A COPY BY WRITING TO DAN WORLEY, OUR SECRETARY, AT OWENS REALTY MORTGAGE, INC., 2221 OLYMPIC BOULEVARD, WALNUT CREEK, CALIFORNIA 94595, OR BY CALLING (925) 935-3840. YOU MAY REVIEW OUR FILINGS WITH THE SEC BY VISITING THE SEC'S HOME PAGE ON THE INTERNET AT WWW.SEC.GOV OR BY VISITING THE INVESTOR RELATIONS SECTION OF OUR WEB SITE AT WWW.OWENSMORTGAGE.COM.

OTHER MATTERS

The Board of Directors does not intend to bring other matters before the Annual Meeting except items incident to the conduct of the meeting. However, on all matters properly brought before the meeting by the Board of Directors or others, it is the intention of the persons named as proxies in the accompanying proxy card, or their substitutes, to vote in accordance with their discretion on such matters.

ckground: transparent"> (a) in relation to an Original Lender, the amount set opposite its name under the heading Commitment in Part II of Schedule 1 (*The Original Parties*) and the amount of any other Commitment transferred to it under this Agreement; and (b) in relation to any other Lender, the amount of any Commitment transferred to it under this Agreement,

to the extent not cancelled, reduced or transferred by it under this Agreement.

Compliance Certificate means a certificate substantially in the form set out in Schedule 8 (*Form of Compliance Certificate*).

Confidentiality Undertaking means a confidentiality undertaking substantially in a recommended form of the LMA or in any other form agreed between the Company and the Agent.

Default means an Event of Default or any event or circumstance specified in Clause 24 (*Events of Default*) which would (with the expiry of a grace period, the giving of notice, the making of any determination under the Finance Documents or any combination of any of the foregoing) be an Event of Default.

Delegate means any delegate, agent, attorney or co-trustee appointed by the Security Agent.

Disruption Event means either or both of:

- (a) a material disruption to those payment or communications systems or to those financial markets which are, in each case, required to operate in order for payments to be made in connection with the Facility (or otherwise in order for the transactions contemplated by the Finance Documents to be carried out) which disruption is not caused by, and is beyond the control of, any of the Parties; or
- (b) the occurrence of any other event which results in a disruption (of a technical or systems-related nature) to the treasury or payments operations of a Party preventing that, or any other Party:
 - (i) from performing its payment obligations under the Finance Documents; or

Table of Contents

(ii) from communicating with other Parties in accordance with the terms of the Finance Documents, and which (in either such case) is not caused by, and is beyond the control of, the Party whose operations are disrupted.

Dividend Account means the Euro dividend account with Citco Nederland N.V. Bank, Account Number 0635978555 charged in favour of the Finance Parties under the Dutch Pledge over Dividend Account to be granted by the Acquiror into which the Acquiror will pay all dividends and other profit distributions received in cash from the Target and all other Subsidiaries.

Dutch Parallel Debt means, in relation to an Underlying Debt (and subject to Clause 8.4(b) (*Fall-back*)), an obligation to pay to the Security Agent an amount equal to (and in the same currency as) the amount of that Underlying Debt.

Engagement Letter means the letter or letters dated on or about the date of this Agreement between the Arranger, the Borrower and the Company in relation to the arrangement of the Facility and other matters.

Environmental Claim means any claim, proceeding, formal notice or investigation by any person in respect of any Environmental Law.

Environmental Law means any applicable law or regulation which relates to the pollution or protection of the environment, harm to or the protection of human health, the condition of the workplace or any emission or substance capable of causing harm to any living organism or the environment.

Environmental Permit means any permit and other Authorisation and the filing of any notification, report or assessment required under any Environmental Law for the operation of the business of any member of the Group conducted on or from the properties owned or used by any member of the Group.

ERISA means the Employee Retirement Income Security Act of 1974 of the United States of America, as amended from time to time, and any applicable regulations promulgated thereunder.

ERISA Affiliate, with respect to any Obligor, means any person that for the purposes of Title IV of ERISA is from time to time a member of the controlled group of any Obligor or under common control with any Obligor within the meaning of section 414 of the Internal Revenue Code.

ERISA Event means:

- (a) (i) the occurrence of a reportable event, within the meaning of section 4043(c) of ERISA, with respect to any Plan unless the 30-day notice requirement with respect to such event has been waived by the PBGC; or
 - (ii) the requirements of section 4043(b) of ERISA applied with respect to a contributing sponsor, as defined in section 4001(a)(13) of ERISA, of a Plan, and an event described in paragraph (9), (10), (11), (12) or (13)

7

Table of Contents

- of section 4043(c) of ERISA is reasonably expected to occur with respect to such Plan within the following 30 days;
- (b) the application for a minimum funding waiver with respect to a Plan;
- (c) the provision by the administrator of any Plan of a notice of intent to terminate such Plan pursuant to section 4041(a)(2) of ERISA (including any such notice with respect to a plan amendment referred to in section 4041(e) of ERISA);
- (d) the cessation of operations at a facility of any Obligor or any ERISA Affiliate in the circumstances described in section 4062(e) of ERISA;
- (e) the withdrawal by any Obligor or any ERISA Affiliate from a Plan during a plan year for which it was a substantial employer, as defined in section 4001(a)(2) of ERISA;
- (f) the conditions for imposition of a lien under section 302(f) of ERISA shall have been met with respect to any Plan;
- (g) the adoption of an amendment to a Plan requiring the provision of security to such Plan pursuant to section 307 of ERISA; or
- (h) the institution by the PBGC of proceedings to terminate a Plan pursuant to section 4042 of ERISA, or the occurrence of any event or condition described in section 4042 of ERISA that constitutes grounds for the termination of, or the appointment of a trustee to administer, such Plan.

EURIBOR means, in relation to any Loan or for determining a Discount Rate:

- (a) the applicable Screen Rate; or
- (b) (if no Screen Rate is available for the Interest Period of that Loan or for determining a Discount Rate) the rate determined by the Agent as the three year Euro swap rate, as of the Specified Time on the Quotation Day.

Euro, **euro**, **EUR**, or **euros** means the single currency of Participating Member States.

Event of Default means any event or circumstance specified as such in Clause 24 (Events of Default).

Expiry Date means, for the Letter of Credit, the last day of its Term.

Facility means the term loan and letter of credit facility made available under this Agreement as described in Clause 2 (*The Facility*).

Facility Office means the office or offices notified by a Lender or the Issuing Bank to the Agent in writing on or before the date it becomes a Lender (or, following that date, by not less than five Business Days written notice) as the office or offices through which it will perform its obligations under this Agreement.

8

Table of Contents

Fee Letter means any letter or letters dated on or about the date of this Agreement between the Arranger and the Borrower or the Agent, the Security Agent and the Borrower setting out any of the fees referred to in Clause 13 (*Fees*).

Finance Document means this Agreement, the Engagement Letter, any Fee Letter, any Compliance Certificate, any Transaction Security Document, any Utilisation Request and any other document designated as such by the Agent and the Borrower.

Finance Party means the Agent, the Arranger, the Security Agent, the Issuing Bank or a Lender.

Financial Indebtedness means any indebtedness for or in respect of:

- (a) moneys borrowed;
- (b) any amount raised by acceptance under any acceptance credit facility or dematerialised equivalent;
- (c) any amount raised pursuant to any note purchase facility or the issue of bonds, notes, debentures, loan stock or any similar instrument;
- (d) the amount of any liability in respect of any lease or hire purchase contract which would, in accordance with the Accounting Principles, be treated as a finance or capital lease;
- (e) the amount of any liability in respect of any advance or deferred purchase agreement if (i) one of the primary reasons behind entering into the agreement is for the person liable under the agreement to raise finance or finance the acquisition or construction of any assets or the purchase of any services (but excluding trade credit on customary commercial terms) or (ii) the agreement is in respect of the acquisition or supply of assets or services and the payment is due more than 90 days after the date of acquisition or supply;
- (f) any agreement or option to re-acquire any asset if arranged primarily as a method of raising finance;
- (g) receivables sold or discounted (other than any receivables to the extent they are sold on a non-recourse basis);
- (h) any amount raised under any other transaction (including any forward sale or purchase agreement) having the commercial effect of a borrowing;
- (i) any Treasury Transaction entered into in connection with protection against or benefit from fluctuation in any rate or price (and, when calculating the value of any derivative transaction, only the marked to market value shall be taken into account);
- (j) any counter-indemnity obligation in respect of a guarantee, indemnity, bond, standby or documentary letter of credit or any other instrument issued by a bank or financial institution; and

9

Table of Contents

(k) the amount of any liability in respect of any guarantee or indemnity for any of the items referred to in paragraphs (a) to (j) above.

Financial Quarter and **Financial Year** each have the respective meaning given to that term in Clause 22.1 (*Financial Definitions*).

Group means the Company and its Subsidiaries for the time being (including, from the Closing Date, the Target and its Subsidiaries).

Group Structure Chart means a document in a form to be agreed with the Agent containing a complete Group structure chart as at the relevant date and indicating all loans which exist and all minority interests held by or existing in the Group.

Guarantor means an Original Guarantor and an Additional Guarantor.

Holding Company means, in relation to a company or corporation, any other company or corporation in respect of which it is a Subsidiary.

Insufficiency means, with respect to any Plan, the amount, if any, of its unfunded benefit liabilities, as defined in section 4001(a)(18) of ERISA.

Intellectual Property means any patents, trade marks, service marks, designs, business names, copyrights, design rights, moral rights, inventions, confidential information, knowhow and other intellectual property rights and interests, whether registered or unregistered and the benefit of all applications and rights to use such assets of each member of the Group.

Internal Revenue Code means the United States Internal Revenue Code of 1986, as amended, including the Treasury regulations promulgated thereunder.

Interest Period means, in relation to a Loan, each period determined in accordance with Clause 11 (*Interest Periods*) and, in relation to an Unpaid Sum, each period determined in accordance with Clause 10.3 (*Default interest*).

Interest Reserve Account means the Euro escrow account of Crown Westfallen LLC with Barclays Bank plc, Account Number 45553777, charged in favour of the Finance Parties under the US Security Agreement granted by the Borrower into which 1,925,505.40 was deposited by the Borrower on 18 August 2006 and such other Euro account as shall be designated by the Agent as the Interest Reserve Account from time to time.

Joint Venture means any joint venture entity, whether a company, unincorporated firm, undertaking, association, joint venture or partnership or any other entity.

L/C Proportion means in relation to a Lender in respect of the Letter of Credit the proportion (expressed as a percentage) borne by that Lender s Available Commitment to the Available Facility immediately prior to the issue of the Letter of Credit, adjusted to reflect any assignment or transfer under this Agreement to or by that Lender.

10

Table of Contents

Legal Due Diligence Report means the legal due diligence report prepared by White & Case LLP in the agreed form, and addressed to and/or capable of being relied on by the Arranger and the other Finance Parties.

Legal Opinion means any legal opinion delivered to the Agent under Clause 4 (*Conditions of Utilisation*) or any other provision of this Agreement.

Legal Reservations means:

- (a) the principle that equitable remedies may be granted or refused at the discretion of a court and the limitation of enforcement by laws relating to insolvency, reorganisation and other laws generally affecting the rights of creditors;
- (b) the time barring of claims under the Limitation Acts, the possibility that an undertaking to assume liability for or indemnify a person against non-payment of UK stamp duty may be void and defences of set-off or counterclaim;
- (c) similar principles, rights and defences under the laws of any Relevant Jurisdiction; and
- (d) any other matters which are set out as qualifications or reservations as to matters of law of general application in the Legal Opinions.

Lender means:

- (a) any Original Lender; and
- (b) any bank, financial institution, trust, fund or other entity which has become a Party in accordance with Clause 25 (*Changes to the Lenders and Obligors*),

which in each case has not ceased to be a Party in accordance with the terms of this Agreement.

Letter of Credit means the bank guarantee in the form set out in Schedule 10 (*Form of Bank Guarantee*).

Loan means a loan made or to be made or deemed to be made under the Facility or which is treated as a Loan in accordance with Clause 10.5(d) (*Capitalisation of Interest*) or the principal amount outstanding for the time being of that loan (including any amounts added to that loan in accordance with Clause 10.5(d) (*Capitalisation of Interest*)).

LMA means the Loan Market Association.

Majority Lenders means:

(a) if there are no Loans then outstanding, a Lender or Lenders whose Commitments aggregate more than $66^2/_3\%$ of the Total Commitments (or, if the Total Commitments have been reduced to zero, aggregated more than $66^2/_3\%$ of the Total Commitments immediately prior to the reduction); or

11

Table of Contents

(b) at any other time, a Lender or Lenders whose participations in the Loans then outstanding aggregate more than $66^2/_3\%$ of all the Loans then outstanding.

Make Whole Date means the date falling six months after the first Utilisation Date.

Mandatory Cost means the percentage rate per annum calculated by the Agent in accordance with Schedule 4 (*Mandatory Cost formula*).

Margin means 12 per cent. per annum provided that:

- (a) if an increase in the Margin takes effect in accordance with Clause 10.6 (*Increase in Margin*) then the Margin shall increase from (but excluding) the Margin Increase Date to (but including) the Margin Reduction Date, by an additional 6 per cent. per annum; and
- (b) if prepayment of not less than 15,000,000 principal amount of the Loans pursuant to Clause 9.5(c) (*Mandatory Prepayment*) does not occur by the Make Whole Date, then:
 - (i) the Margin shall increase by 0.5 per cent. per annum at the end of each consecutive period of a Month thereafter (either (1) up to a maximum rate of 15 per cent. per annum if there is no Margin Increase Date or if there has been a Margin Increase Date and a Margin Reduction Date or (2) otherwise up to a maximum rate of 21 per cent. per annum) until the earlier of the date such prepayment occurs and the first Repayment Date; and
 - (ii) the Margin shall then remain the rate in effect on such date (subject to any applicable reduction under paragraph (a) above).

Margin Increase Date and **Margin Reduction Date** shall have the meanings ascribed thereto in Clause 10.6 (*Increase in Margin*).

Margin Stock means Margin Stock or Margin Security within the meaning of Regulation U or Regulation X of the Board of Governors of the Federal Reserve System of the US (or any successor).

Material Adverse Effect means an event or circumstance which has or is reasonably likely to have a material adverse effect on:

- (a) the business, operations, property, financial condition or prospects of the Group taken as a whole; or
- (b) the ability of an Obligor to perform its payment obligations and comply with the financial covenants under the Finance Documents; or
- (c) the validity or enforceability of, or the effectiveness or ranking of any Security granted or purporting to be granted pursuant to any of, the Finance Documents or the rights or remedies of any Finance Party under any of the Finance Documents.

12

Table of Contents

Month means a period starting on one day in a calendar month and ending on the numerically corresponding day in the next calendar month, except that:

- (a) (subject to paragraph (c) below) if the numerically corresponding day is not a Business Day, that period shall end on the next Business Day in that calendar month in which that period is to end if there is one, or if there is not, on the immediately preceding Business Day;
- (b) if there is no numerically corresponding day in the calendar month in which that period is to end, that period shall end on the last Business Day in that calendar month; and
- (c) if an Interest Period begins on the last Business Day of a calendar month, that Interest Period shall end on the last Business Day in the calendar month in which that Interest Period is to end.

 The above rules will only apply to the last Month of any period.

Multiemployer Plan means a multiemployer plan, as defined in section 4001(a)(3) of ERISA, to which any Obligor or any ERISA Affiliate is making or accruing an obligation to make contributions.

Multiple Employer Plan means a single employer plan, as defined in section 4001(a)(15) of ERISA, that is subject to Title IV of ERISA and that (i) is maintained for employees of any Obligor or any ERISA Affiliate and at least one person (other than the Obligors and the ERISA Affiliates) or (ii) was so maintained and in respect of which any Obligor or any ERISA Affiliate could have liability under section 4064 or 4069 of ERISA in the event such plan has been or were to be terminated.

Non-Obligor means a member of the Group which is not an Obligor.

Obligors means the Borrower, the Company and each of the other Guarantors.

Original Audited Accounts means the consolidated accounts of the Company and its Subsidiaries audited for its Financial Year ended on 31 December 2005.

Original Financial Statements means the Original Audited Accounts and the Original Management Accounts.

Original Management Accounts means the unaudited consolidated quarterly management accounts of the Company for the period 1 January 2006 to and including 31 March 2006.

Participating Member State means any member state of the European Communities that adopts or has adopted the Euro as its lawful currency in accordance with legislation of the European Community relating to Economic and Monetary Union.

Party means a party to this Agreement.

13

Table of Contents

Patriot Act means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56 of the United States.

PBGC means the Pension Benefit Guaranty Corporation (or any successor).

Permitted Acquisition means:

- (a) the Acquisition;
- (b) any acquisition by a member of the Group of an asset sold, leased, transferred or otherwise disposed of by another member of the Group or any acquisition of assets in exchange for assets disposed of, in each case in circumstances constituting a disposal under paragraphs (b) or (c) of the definition of Permitted Disposal;
- (c) any acquisition of shares or securities pursuant to a Permitted Share Issue or otherwise permitted under Clause 23.22 (*Share capital*);
- (d) any acquisition of securities which are Cash Equivalent Investments for treasury management purposes;
- (e) any acquisition of any loan made to another member of the Group permitted under Clause 23.17 (*Loans or credit*);
- (f) any acquisition which forms part of a transaction falling within paragraphs (b) or (c) of the definition of a Permitted Transaction provided that if an asset acquired was subject to Transaction Security before the Permitted Transactions, the Transaction Security over the relevant asset is preserved or the acquiring company gives equivalent Transaction Security over the relevant asset and complies with Clause 23.32 (*Further assurance*) in respect thereof; or
- (g) any acquisition of assets in the ordinary course of trading on arm s length terms (including, without limitation, any loans or other assets made or acquired by any Regulated Entity in the ordinary course of carrying on its regulated activities).

Permitted Disposal means any sale, lease, licence, transfer or other disposal which is on arm s length terms:

- (a) of trading stock or cash (other than shares) made by any member of the Group in the ordinary course of trading of the disposing entity;
- (b) of any asset, investments or receivables or businesses (other than shares) by a member of the Group to another member of the Group, provided that:
 - (i) if the disposing entity had created Transaction Security over the relevant asset, the Transaction Security is preserved or the acquiring company gives equivalent Transaction Security over the relevant asset and complies with Clause 23.32 (*Further assurance*) in respect thereof;

14

Table of Contents

- (ii) if the disposing entity is a Guarantor, the acquiring company must be or become a Guarantor which gives a guarantee which at all times guarantees an amount that is not materially less than the amount guaranteed by the disposing company; and
- (iii) the aggregate consideration for all such disposals made in any Financial Year of the Company does not exceed 500,000 (or its equivalent in other currencies);
- (c) of assets (other than shares, investments or businesses) in exchange for other assets comparable or superior as to type, value and quality provided that if the assets disposed of are subject to Transaction Security the relevant member of the Group creates equivalent Transaction Security over the assets acquired, and complies with Clause 23.32 (*Further assurance*) in respect thereof;
- (d) of obsolete, redundant or worn-out vehicles, plant and equipment for cash;
- (e) of cash or Cash Equivalent Investments in exchange for cash or other Cash Equivalent Investments or the disposal of cash to the extent not prohibited by the Finance Documents or the making of any cash payment under the Finance Documents which does not breach any other provision of the Finance Documents;
- (f) arising as a result of the grant of any Permitted Security;
- (g) arising as a result of a transaction falling within paragraph (b) or (c) of the definition of **Permitted Transaction**:
- (h) the Rooftop sale;
- (i) the sale of the interest in the investment in the Purley Chase Golf Club in accordance with the joint venture agreement relating thereto.

provided that (subject to customary trading arrangements in relation to disposals falling within paragraph (a) of this definition) any such disposal to a third party is for consideration entirely in cash payable in full on completion of the relevant disposal and on terms that the purchaser thereof does not obtain title or possession of any asset subject to such disposal prior to completion of such disposal and payment of the whole consideration therefor (ignoring for this purpose customary purchase price adjustments or contract provisions in any relevant sale agreement) and the requirements of Clause 9.5 (*Mandatory Prepayment*) are complied with respect to any such disposal.

Permitted Financial Indebtedness means Financial Indebtedness:

- (a) arising under the Finance Documents (including the Letter of Credit) (and as varied to the extent permitted by and subject always to the terms of this Agreement);
- (b) arising under a Permitted Loan or a Permitted Guarantee or a Permitted Transaction;

15

Table of Contents

- (c) arising under any finance or capital lease of vehicles, plant, equipment, machinery and computers entered into in the ordinary course of business provided that the aggregate outstanding principal amount does not exceed 500,000 (or its equivalent in other currencies) for the Group as a whole at any time;
- (d) not permitted by any other paragraph of this definition in an aggregate outstanding principal amount which does not exceed 500,000 (or its equivalent in other currencies) for the Group as a whole at any time and which is not owed to any Restricted Person; or
- (e) of the Target existing at the Closing Date and detailed in the Target Balance Sheet.

Permitted Guarantee means:

- (a) the endorsement of negotiable instruments in the ordinary course of trade and not in respect of Financial Indebtedness:
- (b) without prejudice to the limitations in paragraph (f) below, any unsecured performance bond or similar bond guarantee or indemnity, guaranteeing performance by a member of the Group under any contract entered into by a member of the Group on arm s length terms and in the ordinary course of its trading activities and which is not in respect of Financial Indebtedness nor to or for the benefit of a Restricted Person;
- (c) any guarantee permitted under Clause 23.21 (Financial Indebtedness);
- (d) any guarantee given in respect of the netting or set-off arrangements permitted pursuant to paragraph
 (b) of the definition of Permitted Security provided that such arrangements do not require any Obligor to guarantee the liabilities of any Non-Obligor;
- (e) any guarantees contained in the Finance Documents;
- (f) guarantees granted by any member of the Group which are not permitted by the other paragraphs of this definition where the aggregate actual and contingent liability for Financial Indebtedness under such guarantees does not exceed 500,000 (or its equivalent in other currencies) in aggregate at any time;
- (g) without prejudice to the limitations imposed by paragraph (f) above, guarantees to landlords provided that the maximum amount guaranteed by all such guarantees does not exceed 500,000 (or its equivalent in other currencies) at any time; or
- (h) without prejudice to the limitations imposed by paragraph (f) above, guarantees or counter-indemnities in favour of financial institutions which have guaranteed rent obligations of a member of the Group provided that the maximum amount guaranteed by all such guarantees does not exceed 500,000 (or its equivalent in other currencies) at any time.

16

Table of Contents

Permitted Joint Venture means any investment in:

- (a) Purley Chase Golf Club;
- (b) CMBS Servicing Systems LLP;
- (c) TBE; and
- (d) Rooftop Holdings Limited;

Provided that:

- (i) in any financial year of the Company, the aggregate (the **Joint Venture Investment**) of:
 - (A) all amounts subscribed for shares in, lent to, or invested in all such Joint Ventures by any member of the Group;
 - (B) the contingent liabilities of any member of the Group under any guarantee given in respect of the liabilities of any such Joint Venture; and
 - (C) the market value of any assets transferred by any member of the Group to any such joint venture, does not exceed Euro 500,000 (or its equivalent in other currencies); and
- (ii) in the case of any such Joint Venture the Company has procured that the member of the Group holding the investment in such Joint Venture shall exercise its rights to seek to procure that such Joint Venture (or its subsidiaries) do not incur any further Financial Indebtedness or Security in excess of 500,000.

Permitted Loan means:

- (a) any trade credit extended by any member of the Group to its customers on customary commercial terms and in the ordinary course of its trading activities (including without limitation any loans made by any Regulated Entity in the ordinary course of carrying on its regulated activities so long as it does not result in a credit exposure);
- (b) any loan or credit granted by an Obligor to another Obligor, and any loan or credit granted by a Non-Obligor to an Obligor;
- (c) any loan or credit granted by an Obligor to a Non-Obligor, provided that the aggregate principal amount of Financial Indebtedness under any such loans or credit does not when aggregated with the amount of all guarantees outstanding at that time from Obligors in respect of Non-Obligors as permitted under paragraph (f) of the definition of Permitted Guarantees exceed 500,000 (or its equivalent in other currencies) at any time;

17

Table of Contents

- (d) any loan or credit granted between Non-Obligors (other than a loan or credit granted by the Acquiror);
- (e) any loan made by a member of the Group to an employee or director of any member of the Group or to any trustee of any employee share option scheme or employee unit trust scheme for any members of the Group, provided that the maximum aggregate principal amount outstanding under such loans does not exceed 500,000 (or its equivalent in other currencies) at any time;
- (f) any credit constituted by a credit balance on an account of a member of the Group with a bank or financial institution which is not otherwise in breach of this Agreement;
- (g) any loan or credit (other than a loan made by a member of the Group to another member of the Group) which is not permitted by the other paragraphs of this definition so long as the aggregate amount of Financial Indebtedness outstanding under all such loans or credit does not exceed 500,000 (or its equivalent in other currencies) at any time.

Permitted Reorganisation means the restructuring of the group comprising Crown Northcorp Limited and its subsidiaries and the sale of Crown Northcorp Limited and its subsidiaries Crown Mortgage Management Limited and Crown Mortgage Management GmbH and its interests in TBE and Crown Westfalen Credit Services GmbH to the Target on an arm s length basis by the Company for a net purchase price (after all stamp duty, stamp duty reserve, documentary, registration and other similar Taxes, costs, fees and other deductions or liabilities) of not less than 15,000,000 to be satisfied by a cash payment of such price to the Company, to be carried out in accordance with the steps set out at Transaction 1, Transaction 2 and Transaction 3 (as each are defined in the BT Treasury Letter) in the Structure Memorandum.

Permitted Reorganisation Repayment means the repayment of not less than 15,000,000 principal amount of the Loans from the net proceeds of the Permitted Reorganisation.

Permitted Security means:

- (a) any lien arising by operation of law and in the ordinary course of trading and not as a result of any default or omission by any member of the Group;
- (b) Security constituted by any netting or set-off arrangement entered into by any member of the Group in the ordinary course of its banking arrangements for the purpose of netting debit and credit balances (other than cash collateral) of members of the Group or Security arising by operation of law or under standard terms and conditions of the relevant clearing bank over bank accounts permitted to be maintained and operated under this Agreement, provided that such arrangement does not (i) permit or require credit balances of Obligors or the Acquiror to be netted or set-off against debit balances of Non-Obligors or (ii) give rise to other Security over the assets of Obligors or the Acquiror in support of the liabilities of Non-Obligors;

18

Table of Contents

- (c) any Security arising under any retention of title, hire purchase or conditional sale arrangement or arrangements having similar effect in respect of goods or assets supplied to any member of the Group in the ordinary course of the trading of such Group member and on the supplier s standard or usual terms in respect of the goods or assets supplied (and not arising as a result of any default or omission by any member of the Group);
- (d) any Security arising pursuant to an order of attachment or injunction restraining disposal of assets or similar legal process arising in connection with court proceedings which are contested by any member of the Group in good faith by appropriate proceedings with a reasonable prospect of success and which legal process does not constitute a Default where the total amount secured by all such Security does not exceed 500,000 (or its equivalent in other currencies); or
- (e) Security arising automatically by operation of law in favour of any US taxation or any government authority or organisation in respect of Taxes, assessments or governmental charges which are being contested by the relevant member of the Group in good faith and with a reasonable prospect of success, provided that the circumstances giving rise to the existence of the Security does not constitute a Default where the amount secured by such Security does not exceed 500,000 (or its equivalent in other currencies); or
- (f) any security over any rental deposits in respect of any property leased or licensed by a member of the Group in respect of amounts representing not more than twelve months rental for that property including the rent deposit deed dated 21 November 2000 granted by Crown Mortgage Management Limited in favour of Friends Provident Life Office and security deposit under the tenancy agreement dated 17 March 2000 between Crown NorthCorp Limited and Kenneth Owen and John Pyle,

but only to the extent that any Financial Indebtedness secured by such Security secures Permitted Financial Indebtedness.

Permitted Share Issue means:

- (a) an issue of ordinary shares by the Company provided that the Company promptly notifies the Agent that such shares have been issued:
- (b) an issue of shares by one wholly owned Subsidiary of the Company to its direct Holding Company (which is another wholly owned Subsidiary of the Company or the Company), provided that if the existing shares in the issuing company are the subject of Transaction Security, such new shares become the subject of equivalent Transaction Security on substantially the same terms;
- (c) the issue of shares as described in the Structure Memorandum where, if the existing shares in the issuing company are the subject of Transaction Security, such new shares become the subject of the Transaction Security in favour of the Finance Parties on substantially the same terms; or

19

Table of Contents

(d) the issue of shares by the Acquiror (or any member of the Group) under and in accordance with the Warrant Documents.

Permitted Transaction means:

- (a) any disposal required, Financial Indebtedness incurred, guarantee, indemnity, or Security given, or other transaction arising under the Finance Documents;
- (b) the Permitted Reorganisation;
- (c) the acquisitions, disposals, guarantees, loans, payments, transactions or Group reorganisations expressly and fully described in the Structure Memorandum; or
- (d) transactions other than (i) any sale, lease, licence, transfer or other disposal and (ii) the granting or creation of Security or the incurring or permitting to subsist of Financial Indebtedness, conducted in the ordinary course of trading on arm s length terms.

Plan means a Single Employer Plan or a Multiple Employer Plan.

Quarter Date means the last day of a Financial Quarter.

Quotation Day means, in relation to any period for which an interest rate is to be determined or any date in respect of which a Discount Rate is to be determined, two TARGET Days before the first day of that period or before that date (as the case may be).

Receiver means a receiver or receiver and manager or administrative receiver of the whole or any part of the Charged Property.

Register has the meaning given to it in Clause 25.11 (*The Register*).

Regulated Entity means Crown Mortgage Management Limited, Crown Mortgage Management GmbH and, following the Closing Date, the Target.

Relevant Interbank Market means the European interbank market.

Relevant Jurisdiction means, in relation to an Obligor:

- (a) its jurisdiction of incorporation;
- (b) any jurisdiction where any asset subject to or intended to be subject to the Transaction Security to be created by it is situated;
- (c) any jurisdiction where it conducts any material part of its business or carries on the administration of its business on a regular basis or has its corporate seat; and
- (d) any jurisdiction whose laws govern any of (or the perfection of any of) the Security Documents entered into by it.

20

Table of Contents

Relevant Period has the meaning given to that term in Clause 20.1 (*Financial Definitions*).

Repayment Date has the meaning given to it in Clause 8.1 (*Repayment of Loans*).

Repayment Instalment means the principal amount outstanding under the Facility due to be repaid on the relevant Repayment Date.

Repeating Representations means each of the representations set out in Clauses 20.2 (*Status*) to 20.8 (*Insolvency*), Clause 20.11 (*No default*), Clauses 20.13(d) and 20.13(h) (*Original Financial Statements*), Clause 20.14 (*No proceedings pending or threatened*), Clause 20.19 (*Ranking*), Clauses 20.20 (*Good title to asset*) to 20.21 (*Legal and beneficial ownership*) and Clause 20.30 (*US government regulations*).

Reports means the Accountants Report, the Capital and Liquidity Report, the Legal Due Diligence Report and the Structure Memorandum.

Restricted Person means:

- (a) any person with an interest (direct or indirect) in the shares in the Company or any similar equity interest;
- (b) any joint venture, consortium, partnership or similar arrangement of which any person described in paragraph (a) above is a member;
- (c) any Affiliate of the Company or any person described in paragraph (a) above; and
- (d) any person owning beneficially and/or legally (directly or indirectly) 20 per cent. or more of the equity interest in, or 20 per cent. or more of whose equity interest is owned beneficially and/or legally (directly or indirectly) by, any person described in paragraph (a) above.

Rooftop Sale means the sale of the Group s 20 per cent. interest in the share capital of Rooftop Holdings Limited on an arm s length basis for a net purchase price (after all Taxes, costs, fees and other deductions or liabilities) of not less than 10,000,000 to be satisfied by a cash payment of such net price to the Company.

Screen Rate means the three year Euro swap rate, displayed on the appropriate page of Bloomberg, as selected by the Agent. If the agreed page is replaced or service ceases to be available, the Agent may specify another page or service displaying the appropriate rate after consultation with the Borrower and the Lenders.

SEC means the Securities and Exchange Commission of the United States or any successor thereto.

Secured Parties means each Finance Party from time to time party to this Agreement and any Receiver or Delegate.

Security means a mortgage, charge, pledge, lien or other security interest securing any obligation of any person or any other agreement or arrangement having a similar effect.

21

Table of Contents

Shareholder Indebtedness means any Financial Indebtedness owed to a Restricted Person.

Single Employer Plan means a single employer plan, as defined in section 4001(a)(15) of ERISA, that is subject to title IV of ERISA that (a) is maintained for employees of any Obligor or any ERISA Affiliate and no person other than the Obligors and the ERISA Affiliates or (b) was so maintained and in respect of which any Obligor or any ERISA Affiliate could have liability under section 4069 of ERISA in the event such plan has been or were to be terminated.

Specified Time means a time determined in accordance with Schedule 9 (*Timetables*).

Structure Memorandum means the Group corporate and tax structuring information package provided by Baker Tilly, Bury St Edmonds on 21 July 2006 and 24 July 2006 as follows:

- (a) e-mail dated 21 July 2006 from Adrian Mole of Baker Tilly, Bury St Edmonds to Clifford Atkins of Shearman & Sterling (London) LLP;
- (b) powerpoint presentation entitled Crown Mortgage Management Crown NorthCorp Group Structure before and after Bank Purchase revised May 2006;
- (c) copy letter dated 14 June 2006 from Baker Tilly, Bury St Edmonds to The Secretary, HM Treasury (AP) (the **BT Treasury Letter**);
- (d) letter dated 16 June 2006 from HM Treasury to Baker Tilly, Bury St Edmonds;
- (e) memo dated 21 June 2006 entitled Project: Crown Westfalen Bank AG prepared by Baker Tilly, Frankfurt; and
- (f) memo dated 27 June 2006 to Adrian Mole of Baker Tilly from Frank Melotte of Baker Tilly, Amsterdam entitled Tax Advisers Re: Crown Mortgage Management;
- (g) e-mail dated 17 July 2006 from Stefan Kirchmann of Baker Tilly, Frankfurt to Adrian Mole of Baker Tilly, Bury St Edmonds; and
- (h) letter from Adrian Mole of Baker Tilly, Bury St Edmunds to Peter Walker of Crown Mortgage Management Limited dated 27 July, 2006,

in the case of the items described at paragraphs (a), (e), (f), (g) and (h) addressed to, and/or capable of being relied upon by the Arranger and the other Finance Parties.

Subsidiary means a subsidiary undertaking within the meaning of section 258 of the Companies Act 1985.

TARGET means Trans-European Automated Real-time Gross Settlement Express Transfer payment system.

Table of Contents

TARGET Day means any day on which TARGET is open for the settlement of payments in Euro.

Target means Westfalenbank AG registered with the commercial register of the local court in Bochum under HRB1941.

Target Balance Sheet means the balance sheet of the Target in the agreed form.

Target Shares means the entire share capital of the Target.

Tax means any tax, levy, impost, duty or other charge or withholding of a similar nature (including any penalty or interest payable in connection with any failure to pay or any delay in paying any of the same).

Tax Credit, **Tax Deduction** and **Tax Payment** each have the meanings given to it in Clause 14.1 (*Tax Definitions*).

Termination Date means 31 December, 2008.

Total Commitments means the aggregate of the Commitments being 25,000,000 at the date of this Agreement.

Transaction Documents means the Finance Documents, the Acquisition Documents and the Warrant Documents.

Transaction Security means the Security created or expressed to be created in respect of the obligations of any of the Obligors or the Acquiror under any of the Finance Documents pursuant to the Security Documents.

Transaction Security Documents means each of the documents listed as a Transaction Security Document in paragraph 3(d) of Part I of Schedule 2 (*Conditions Precedent*) or paragraph 8 of Part III of Schedule 2 (*Conditions Precedent*) and any document required to be delivered to the Agent under paragraph 13 of Part II of Schedule 2 (*Conditions Precedent*) together with any other document entered into by any Obligor or the Acquiror creating or expressed to create any Security over all or any part of its assets in respect of the obligations of any of the Obligors or the Acquiror under any of the Finance Documents.

Transfer Certificate means a certificate substantially in the form set out in Schedule 5 (*Form of Transfer Certificate*) or any other form agreed between the Agent and the Borrower.

Transfer Date means, in relation to a transfer, the later of:

- (a) the proposed Transfer Date specified in the Transfer Certificate or Assignment Agreement; and
- (b) the date on which the Agent executes the Transfer Certificate or Assignment Agreement.

Treasury Transactions means any derivative transaction entered into in connection with protection against or benefit from fluctuation in any rate or price.

23

Table of Contents

Underlying Debt means, in relation to an Obligor and at any given time, each obligation (whether present or future, actual or contingent) owing by that Obligor to a Finance Party under the Finance Documents (including, for the avoidance of doubt, any change or increase in those obligations pursuant to or in connection with any amendment or supplement or restatement or novation of any Finance Document, in each case whether or not anticipated as of the date of this Agreement) excluding that Obligor s Dutch Parallel Debts).

Unpaid Sum means any sum due and payable but unpaid by an Obligor under the Finance Documents (other than any Obligor s Dutch Parallel Debts).

US and **USA** means the United States of America or any of its states and territories and the District of Columbia.

US Obligor means an Obligor organised under any State of the US.

US Bankruptcy Code means Title 11 of the United States Code entitled Bankruptcy, as amended, or any successor thereof.

US Solvent and US Solvency mean, with respect to any person on a particular date, that on such date (a) the fair value of the property of such person is greater than the total amount of liabilities, including, without limitation, contingent liabilities, of such person, (b) such person does not intend to, and does not believe that it will, incur debts or liabilities beyond such person s ability to pay as such debts and liabilities mature and (c) such person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such person s property would constitute an unreasonably small capital. The amount of contingent liabilities at any time shall be computed as the amount that, in the light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

Utilisation means a Loan or a Letter of Credit.

Utilisation Date means the date on which a Utilisation is made.

Utilisation Request means a notice substantially in the form set out in Part I of Schedule 3 (*Requests*).

VAT means value added tax as provided for in the Value Added Tax Act 1994 and any other tax of a similar nature.

Vendor means Bayerische Hypo-und Vereinsbank A.G.

Warrant Documents means the warrant instrument entered into or to be entered into by, inter alia, the Acquiror and the Borrower in relation to the issue of warrants relating to the share capital of the Acquiror in the agreed form (the **Warrant Agreement**) and any other document or agreement entered into or executed in connection with that instrument or the rights set out in that agreement, each in the agreed form.

Warrants means the warrants issued pursuant to the Warrant Documents.

24

Table of Contents

Withdrawal Liability has the meaning specified in Part I of Subtitle E of title IV of ERISA.

1.2 **Construction**

- (a) Unless a contrary indication appears, any reference in this Agreement to:
 - (i) the **Agent**, the **Arranger**, any **Finance Party**, the **Issuing Bank**, any **Lender**, any **Obligor Party**, any **Secured Party**, the **Security Agent** or any other person shall be construed so as to include its successors in title, permitted assigns and permitted transferees and in the case of the Security Agent, any person for the time being appointed as Security Agent in accordance with the Finance Documents:
 - (ii) a document in agreed form is a document which is previously agreed in writing by or on behalf of the Company and the Agent or, if not so agreed, is in the form specified by the Agent (acting reasonably);
 - (iii) **assets** includes present and future properties, revenues and rights of every description;
 - (iv) a **director** in relation to the Acquiror, means a managing director (*bestuurder*) and **board of directors** means its managing board (*bestuur*);
 - (v) a **Finance Document** or a **Transaction Document** or any other agreement or instrument is a reference to that Finance Document or Transaction Document or other agreement or instrument as amended, novated, supplemented, extended or restated;
 - (vi) **guarantee** means (other than in Clause 17 (*Guarantee and Indemnity*)) any guarantee, letter of credit, bond, indemnity or similar assurance against loss, or any obligation, direct or indirect, actual or contingent, to purchase or assume any indebtedness of any person or to make an investment in or loan to any person or to purchase assets of any person where, in each case, such obligation is assumed in order to maintain or assist the ability of such person to meet its indebtedness;
 - (vii) a Lender s **participation** in relation to the Letter of Credit, shall be construed as a reference to the relevant amount that is or may be payable by a Lender in relation to the Letter of Credit;
 - (viii) **indebtedness** includes any obligation (whether incurred as principal or as surety) for the payment or repayment of money, whether present or future, actual or contingent;
 - (ix) a **person** includes any person, firm, company, corporation, government, state or agency of a state or any association, trust or partnership (whether or not having separate legal personality) or two or more of the foregoing;

25

Table of Contents

- (x) a **regulation** includes any regulation, rule, official directive, request or guideline (whether or not having the force of law) of any governmental, intergovernmental or supranational body, agency, department or regulatory, self-regulatory or other authority or organisation;
- (xi) a provision of law is a reference to that provision as amended or re-enacted;
- (xii) the date of this Agreement is a reference to 31 July 2006; and
- (xiii) a time of day is a reference to London time.
- (b) Section, Clause and Schedule headings are for ease of reference only.
- (c) Unless a contrary indication appears, a term used in any other Finance Document or in any notice given under or in connection with any Finance Document has the same meaning in that Finance Document or notice as in this Agreement.
- (d) A Default (other than an Event of Default) is **continuing** if it has not been remedied or waived and an Event of Default is **continuing** if it has not been remedied or waived.
- (e) The Borrower providing **cash cover** for the Letter of Credit means the Borrower paying an amount in Euro to an interest-bearing account in the name of the Borrower and the following conditions being met:
 - (i) the account is with the Agent;
 - (ii) until no amount is or may be outstanding under the Letter of Credit, withdrawals from the account may only be made to pay a Finance Party amounts due and payable to it under this Agreement in respect of the Letter of Credit; and
 - (iii) the Borrower has executed a security document over that account, in form and substance satisfactory to the Agent creating a first ranking security interest over that account.
- (f) The Borrower **repaying** or **prepaying** the Letter of Credit means:
 - (i) the Borrower providing cash cover for the Letter of Credit;
 - (ii) the maximum amount payable under the Letter of Credit being reduced or cancelled in accordance with its terms; or
 - (iii) the Issuing Bank being satisfied that it has no further liability under the Letter of Credit, and the amount by which a Letter of Credit is repaid or prepaid under paragraphs (f)(i) and (f)(ii) above is the amount of the relevant cash cover or reduction.

26

Table of Contents

- (g) An amount borrowed includes any amount utilised by way of the Letter of Credit.
- (h) A Lender funding its participation in a Utilisation includes a Lender participating in the Letter of Credit.
- (i) The outstanding amount of the Letter of Credit at any time is the maximum amount that is or may be payable by the Borrower in respect of the Letter of Credit at that time.

1.3 Third party rights

- (a) Unless expressly provided to the contrary in a Finance Document a person who is not a Party has no right under the Contracts (Rights of Third Parties) Act 1999 (the **Third Parties Act**) to enforce or to enjoy the benefit of any term of this Agreement.
- (b) Notwithstanding any term of any Finance Document the consent of any person who is not a Party is not required to rescind or vary this Agreement at any time.

27

SECTION 2 THE FACILITY

2. THE FACILITY

2.1 The Facility

Subject to the terms of this Agreement, the Lenders make available to the Borrower a Euro term loan and letter of credit facility in an aggregate amount equal to the Total Commitments.

2.2 Finance Parties rights and obligations

- (a) The obligations of each Finance Party under the Finance Documents are several. Failure by a Finance Party to perform its obligations under the Finance Documents does not affect the obligations of any other Party under the Finance Documents. No Finance Party is responsible for the obligations of any other Finance Party under the Finance Documents.
- (b) The rights of each Finance Party under or in connection with the Finance Documents are separate and independent rights and any debt arising under the Finance Documents to a Finance Party from an Obligor shall be a separate and independent debt.
- (c) A Finance Party may, except as otherwise stated in the Finance Documents, separately enforce its rights under the Finance Documents.

2.3 **Obligors Agent**

- (a) Each Obligor (other than the Company) by its execution of this Agreement or an Accession Letter irrevocably appoints the Company to act on its behalf as its agent in relation to the Finance Documents and irrevocably authorises:
 - (i) the Company on its behalf to supply all information concerning itself contemplated by this Agreement to the Finance Parties and to give all notices and instructions (including, in the case of the Borrower, Utilisation Requests), to execute on its behalf any Accession Letter, to make such agreements and to effect the relevant amendments, supplements and variations capable of being given, made or effected by any Obligor notwithstanding that they may affect the Obligor, without further reference to or the consent of that Obligor; and
 - (ii) each Finance Party to give any notice, demand or other communication to that Obligor pursuant to the Finance Documents to the Company,

and in each case the Obligor shall be bound as though the Obligor itself had given the notices and instructions (including, without limitation, any Utilisation Requests) or executed or made the agreements or effected the amendments, supplements or variations, or received the relevant notice, demand or other communication.

28

Table of Contents

(b) Every act, omission, agreement, undertaking, settlement, waiver, amendment, supplement, variation, notice or other communication given or made by the Obligors Agent or given to the Obligors Agent under any Finance Document on behalf of another Obligor or in connection with any Finance Document (whether or not known to any other Obligor and whether occurring before or after such other Obligor became an Obligor under any Finance Document) shall be binding for all purposes on that Obligor as if that Obligor had expressly made, given or concurred with it. In the event of any conflict between any notices or other communications of the Obligors Agent and any other Obligor, those of the Obligors Agent shall prevail.

3. **PURPOSE**

3.1 **Purpose**

The Borrower shall apply all amounts borrowed by it under the Facility and the Letter of Credit only in making a share premium payment (*onbedongen agio storting*) to the Acquiror which shall fund payment to the Vendor of the purchase price for the Target Shares in accordance with the terms of the Acquisition Agreement.

3.2 **Monitoring**

No Finance Party is bound to monitor or verify the application of any amount borrowed pursuant to this Agreement.

4. CONDITIONS OF UTILISATION

4.1 **Initial conditions precedent**

The Borrower may not deliver a Utilisation Request unless, and a Lender shall be obliged to comply with Clause 5.3 (*Lenders Participation*) (but subject to the conditions specified therein) if, the Agent has received all of the documents and other evidence listed in Schedule 2 (*Conditions Precedent*) in form and substance satisfactory to the Agent. The Agent shall notify the Borrower and the Lenders promptly upon being so satisfied.

4.2 Further conditions precedent

The Lenders will only be obliged to comply with Clause 5.3 (*Lenders participation*) if on the date of a Utilisation Request and on the proposed Utilisation Date:

- (a) no Default is continuing or would result from the proposed Utilisation; and
- (b) the Repeating Representations to be made by each Obligor are true in all material respects.

20

SECTION 3 UTILISATION

5. UTILISATION LOANS

5.1 **Delivery of a Utilisation Request**

The Borrower shall deliver to the Agent a duly completed Utilisation Request not later than the Specified Time in relation to any Loan to be made under paragraph (a) of Clause 7.5 (*Deemed Loan*).

5.2 Completion of a Utilisation Request for a Loan

A Utilisation Request in respect of a Loan to be made under paragraph (a) of Clause 7.5 (*Deemed Loan*) is irrevocable and will not be regarded as having been duly completed unless:

- (a) the proposed Utilisation Date is a Business Day within the Availability Period, being the Closing Date;
- (b) the currency of the Utilisation is Euros and the amount of the Loan to be made under paragraph (a) above is the relevant amount to be demanded under the Letter of Credit; and
- (c) the Borrower has delivered to the Agent evidence that the German Federal Agency for Supervision of Financial Services (*Bundesanstalt für Finanzdienstleistungsaufsicht*) does not and will not raise objections to the Acquisition nor the entry into and performance of the Acquisition Documents and/or the Finance Documents nor the validity and enforceability of the Acquisition Documents and/or the Finance Documents.

5.3 Lenders participation

- (a) If the conditions set out in this Agreement have been met, each Lender shall make its participation in a Loan to be made under paragraph (a) of Clause 7.5 (*Deemed Loan*) available by the Utilisation Date through its Facility Office.
- (b) The amount of each Lender s participation in a Loan to be made under paragraph (a) of Clause 7.5 (*Deemed Loan*) will be equal to the proportion borne by its Available Commitment to the Available Facility immediately prior to making that Loan.
- (c) The Agent shall notify each Lender of the amount of a Loan to be made under paragraph (a) of Clause 7.5 (*Deemed Loan*) and the amount of its participation in that Loan by the Specified Time.

6. UTILISATION THE LETTER OF CREDIT

6.1 Currency and amount

The maximum amount of the Letter of Credit must be a Euro amount of 25,000,000.

30

6.2 Delivery of a Utilisation Request for the Letter of Credit

The Borrower may request the Letter of Credit to be issued by delivery to the Agent of a duly completed Utilisation Request not later than the Specified Time.

6.3 Completion of the Utilisation Request for the Letter of Credit

The Utilisation Request for a Letter of Credit is irrevocable and will not be regarded as having been duly completed unless:

- (a) it specifies that it is for the Letter of Credit;
- (b) the proposed Utilisation Date is a Business Day within the Availability Period and is the Acquisition Signing Date;
- (c) the currency and amount of the Letter of Credit comply with Clause 6.1 (Currency and amount);
- (d) the form of the Letter of Credit is attached;
- (e) the Expiry Date of the Letter of Credit falls on or before the Closing Date;
- (f) the delivery instructions for the Letter of Credit are specified; and
- (g) the identity of the beneficiary of the Letter of Credit is the Vendor.

6.4 Issue of the Letter of Credit

- (a) If the conditions set out in this Agreement have been met, the Issuing Bank shall issue the Letter of Credit on the Utilisation Date.
- (b) Subject to Clause 4.1 (*Initial conditions precedent*), the Issuing Bank will only be obliged to comply with paragraph (a) above, if on the date of the Utilisation Request and on the proposed Utilisation Date:
 - (i) no Default is continuing or would result from the proposed Utilisation; and
 - (ii) all the representations and warranties in Clause 20 (Representations) are true.
- (c) The amount of each Lender s participation in the Letter of Credit will be equal to the proportion borne by its Available Commitment to the Available Facility immediately prior to the issue of the Letter of Credit.
- (d) The Agent shall notify each Lender of the details of its participation in the Letter of Credit by the Specified Time.

31

7. **LETTER OF CREDIT**

7.1 **Immediately payable**

Subject to Clause 7.5 (*Deemed Loan*), the Borrower shall repay or prepay any amount outstanding under a Letter of Credit immediately.

7.2 Claims under the Letter of Credit

- (a) The Borrower irrevocably and unconditionally authorises the Issuing Bank to pay any claim made or purported to be made under the Letter of Credit and which appears on its face to be in order (in this Clause 7, a claim).
- (b) Subject to Clause 7.5 (*Deemed Loan*), the Borrower shall immediately on demand pay to the Agent for the Issuing Bank an amount equal to the amount of any claim.
- (c) The Borrower acknowledges that the Issuing Bank:
 - (i) is not obliged to carry out any investigation or seek any confirmation from any other person before paying a claim; and
 - (ii) deals in documents only and will not be concerned with the legality of a claim or any underlying transaction or any available set-off, counterclaim or other defence of any person.
- (d) The obligations of the Borrower under this Clause will not be affected by:
 - (i) the sufficiency, accuracy or genuineness of any claim or any other document; or
 - (ii) any incapacity of, or limitation on the powers of, any person signing a claim or other document.

7.3 **Indemnities**

- (a) The Borrower shall immediately on demand indemnify the Issuing Bank against any cost, loss or liability incurred by the Issuing Bank (otherwise than by reason of the Issuing Bank s gross negligence or wilful misconduct) in acting as the Issuing Bank under the Letter of Credit.
- (b) Each Lender shall (according to its L/C Proportion) immediately on demand indemnify the Issuing Bank against any cost, loss or liability incurred by the Issuing Bank (otherwise than by reason of the Issuing Bank s gross negligence or wilful misconduct) in acting as the Issuing Bank under the Letter of Credit (unless the Issuing Bank has been reimbursed by an Obligor pursuant to a Finance Document) including in relation to a Loan made by the Issuing Bank under Clause 7.5 (*Deemed Loan*).
- (c) Subject to Clause 7.5 (*Deemed Loan*), the Borrower shall immediately on demand reimburse any Lender for any payment it makes to the Issuing Bank under this Clause 7.3 in respect of the Letter of Credit.

32

Table of Contents

- (d) The obligations of each Lender under this Clause are continuing obligations and will extend to the ultimate balance of sums payable by that Lender in respect of the Letter of Credit, regardless of any intermediate payment or discharge in whole or in part.
- (e) The obligations of any Lender or the Borrower under this Clause will not be affected by any act, omission, matter or thing which, but for this Clause, would reduce, release or prejudice any of its obligations under this Clause (without limitation and whether or not known to it or any other person) including:
 - (i) any time, waiver or consent granted to, or composition with, any Obligor, any beneficiary under the Letter of Credit or any other person;
 - (ii) the release of any other Obligor or any other person under the terms of any composition or arrangement with any creditor or any member of the Group;
 - (iii) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor, any beneficiary under the Letter of Credit or other person or any non-presentation or non-observance of any formality or other requirement in respect of any instrument or any failure to realise the full value of any security;
 - (iv) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor, any beneficiary under the Letter of Credit or any other person;
 - (v) any amendment (however fundamental) or replacement of a Finance Document, the Letter of Credit or any other document or security;
 - (vi) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document, the Letter of Credit or any other document or security; or
 - (vii) any insolvency or similar proceedings.

7.4 **Rights of contribution**

No Obligor will be entitled to any right of contribution or indemnity from any Finance Party in respect of any payment it may make under this Clause 7.

7.5 **Deemed Loan**

(a) Subject to paragraph (b) below any amount demanded from and paid by the Issuing Bank under the Letter of Credit shall be deemed to constitute a Loan made on the due date for payment under the Letter of Credit by the Issuing Bank to the extent that there remains a sufficient amount of the Commitments undrawn at the time (for this purpose the relevant Utilisation by the Letter of Credit being deemed not to be outstanding to the extent of the amount demanded).

33

Table of Contents

(b) All provisions of this Agreement relating to Loans and the making of them including the provisions of Clause 4.2 (*Further conditions precedent*) and all provisions relating to the payment of interest, the repayment and prepayment of principal and other amounts shall apply to any Loan made under paragraph (a) above. The proceeds of such Loan shall be applied in satisfaction of the Borrower's obligations to Lenders under the Letter of Credit under this Agreement.

34

SECTION 4 REPAYMENT, PREPAYMENT AND CANCELLATION

8. **REPAYMENT**

8.1 **Repayment of Loans**

The Borrower shall repay the aggregate Loans on the following dates (each a **Repayment Date**) and by the following instalments:

Repayment Date	Principal Amount to be Repaid
31 March, 2007	15,000,000 (provided that if the Permitted Reorganisation has completed on or prior to that date, then the principal amount on that date shall be the higher of (i) zero and (ii) 15,000,000 less the principal amount prepaid in accordance with Clause 9.5(c) (<i>Mandatory Prepayment</i>) from the Net Proceeds of the Permitted Reorganisation)

Termination Date The balance of the Loans outstanding

If on the first Repayment Date, the principal amount of the Loans outstanding is less than the principal amount specified to be repaid on that Repayment Date, then the principal amount to be repaid shall be instead the balance of the Loans remaining outstanding on that Repayment Date.

8.2 **Reborrowing**

The Borrower may not reborrow any part of the Facility which is repaid.

8.3 **Accrued Interest**

On any repayment under Clause 8.1 (*Repayment of Loans*) accrued interest on the principal amount of the Loans repaid on that date must be paid.

8.4 **Dutch Parallel Debts**

(a) Dutch Parallel Debts

- (i) Each Obligor undertakes with the Security Agent to pay to the Security Agent its Dutch Parallel Debts.
- (ii) Subclause 8.4(a)(i) above is:
 - (A) for the purpose of ensuring the validity and effect of any security right governed by Dutch law and granted or to be

35

Table of Contents

granted by any Obligor or the Acquiror pursuant to the Finance Documents; and

- (B) without prejudice to the other provisions of the Finance Document.
- (iii) Each Dutch Parallel Debt is a separate and independent obligation and shall not constitute the Security Agent and any Finance Party as joint creditors (*hoofdelijk schuldeisers*) of any Underlying Debt.

(b) Fall-back

- (i) If (notwithstanding subclause 8.4(a)(a)(iii) (*Dutch Parallel Debts*) above) any Dutch Parallel Debt constitutes the Security Agent as a joint creditor with any Finance Party, the Security Agent may determine (at its discretion) that that Dutch Parallel Debt and one or more other Dutch Parallel Debts shall be combined into one single Dutch Parallel Debt (a **Combined Dutch Parallel Debt**), whereupon those Dutch Parallel Debts shall be combined into a Combined Dutch Parallel Debt:
 - (A) the amount of which shall be equal to the aggregate of the amounts of the Underlying Debts combined into it:
 - (B) which shall, if the Underlying Debts are expressed in different currencies, be expressed in such of those currencies or Euro as the Security Agent may determine (and, for this purpose, each Underlying Debt shall be converted into the currency of the Combined Dutch Parallel Debt at the rate of exchange reasonably selected by the Security Agent);
 - (C) which shall, if the Underlying Debts combined into it fall due at different times, fall due in parts corresponding to those Underlying Debts (but otherwise in accordance with Clause 8.4(c) (*Payment*)); and
 - (D) to which this Agreement shall otherwise apply as if the Dutch Combined Parallel Debt were a Dutch Parallel Debt.
- (ii) If any Underlying Debt is avoided or reduced other than:
 - (A) as a result of payment to, or recovery or discharge by, the Finance Party to which the Underlying Debt is owed: or
 - (B) otherwise with the consent of that Finance Party, the amount of the Dutch Parallel Debt corresponding to that Underlying Debt shall be equal to the amount which the Underlying Debt would have had if the avoidance or reduction had not occurred.

36

Table of Contents

(c) Payment

- (i) No Obligor may pay any Dutch Parallel Debt other than at the instruction of, and in the manner determined by, the Security Agent.
- (ii) Without prejudice to subclause 8.4(c)(i) no Obligor shall be obliged to pay any Dutch Parallel Debt before the corresponding Underlying Debt has fallen due.
- (iii) All payments to be made by an Obligor in respect of its Dutch Parallel Debts shall be calculated and be made without (and clear of any deduction for) set-off or counterclaim.

(d) Application

Any payment made, or amount recovered, in respect of an Obligor s Dutch Parallel Debts shall reduce the Underlying Debts owed to a Finance Party by the amount which that Finance Party has received out of that payment or recovery under the Finance Documents.

9. PREPAYMENT AND CANCELLATION

9.1 **Illegality**

If it becomes unlawful in any applicable jurisdiction for a Lender to perform any of its obligations as contemplated by this Agreement or to fund or maintain its participation in any Utilisation:

- (a) that Lender shall promptly notify the Agent upon becoming aware of that event;
- (b) upon the Agent notifying the Borrower, the Commitment of that Lender will be immediately cancelled; and
- (c) the Borrower shall repay that Lender s participation in the Utilisations made to the Borrower on the last day of the Interest Period for each Utilisation occurring after the Agent has notified the Borrower or, if earlier, the date specified by the Lender in the notice delivered to the Agent (being no earlier than the last day of any applicable grace period permitted by law).

9.2 Illegality in relation to Issuing Bank

If it becomes unlawful for the Issuing Bank to issue or leave outstanding the Letter of Credit, then:

- (a) the Issuing Bank shall promptly notify the Agent upon becoming aware of that event;
- (b) upon the Agent notifying the Company, the Issuing Bank shall not be obliged to issue the Letter of Credit; and

37

(c) the Company shall procure that the Borrower shall use its best endeavours to procure the release of the Letter of Credit issued by the Issuing Bank and outstanding at such time.

9.3 Voluntary prepayment of Loans and Letters of Credit

Subject to Clause 9.6 (*Make Whole Payments*), the Borrower may at any time after the Closing Date, if it gives the Agent not less than 5 Business Days (or such shorter period as the Majority Lenders may agree) prior notice, prepay the whole or any part of any Loan (or, if only the Letter of Credit is outstanding, the Letter of Credit is outstanding, the Letter of Credit is outstanding, the Letter of Credit) by a minimum amount of 500,000).

9.4 Right of repayment and cancellation in relation to a single Lender

- (a) If:
 - (i) any sum payable to any Lender by an Obligor is required to be increased under paragraph (c) of Clause 14.3 (*Tax gross-up*); or
 - (ii) any Lender claims indemnification from the Borrower under Clause 14.4 (*Tax indemnity*) or Clause 15.1 (*Increased costs*),

the Borrower may, whilst the circumstance giving rise to the requirement for indemnification continues, give the Agent notice of cancellation of the Commitment of that Lender and its intention to procure the repayment of that Lender s participation in the Utilisation.

- (b) On receipt of a notice referred to in paragraph (a) above, the Commitment of that Lender shall immediately be reduced to zero.
- (c) On the last day of each Interest Period which ends after the Borrower has given notice under paragraph (a) above (or, if earlier, the date specified by the Borrower in that notice), the Borrower shall repay that Lender s participation in each Utilisation and together with all interest and other amounts accrued under the Finance Documents.

9.5 **Mandatory Prepayment**

- (a) For the purposes of this Clause 9.5:
 - (i) Acquisition Proceeds means the proceeds of a claim (a Recovery Claim) against the Vendor or any of its Affiliates in relation to the Acquisition Documents or against the provider of a Report (in its capacity as a provider of that Report) and after deducting any reasonable expenses payable and any stamp duty, stamp duty reserve, documentary, registration and other similar Taxes (if applicable) which are incurred by a member of the Group to persons who are not members of the Group in each case in relation to that Recovery Claim;

38

- (ii) **Disposal** means a sale, lease, licence, transfer, loan or other disposal by a person of any asset, undertaking or business (whether by a voluntary or involuntary single transaction or series of transactions);
- (iii) **Disposal Proceeds** means the consideration receivable by any member of the Group (including any amount receivable in repayment of debt owed to other members of the Group remaining after the relevant Disposal) for the Permitted Reorganisation, the Rooftop Sale or any other Disposal made by any member of the Group after deducting all amounts due to the Agent under the Finance Documents in respect of all costs, expenses, liabilities and any stamp duty, stamp duty reserve, documentary, registration and other similar Taxes (if applicable) incurred by the Agent in relation to any release or amendment of any Transaction Security relating to that Disposal and all reasonable expenses which are incurred by any member of the Group with respect to that Disposal to persons who are not members of the Group;
- (iv) **Excluded Disposal Proceeds** means the proceeds of any Disposal which the Company notifies the Agent are the proceeds of any Disposal referred to in paragraph (a), (b), (c), (e) or (f) of the definition of **Permitted Disposal**; and
- (v) **Insurance Proceeds** means the proceeds of an insurance claim received by any member of the Group (other than to meet a third party claim) and after deducting any reasonable expenses in relation to that claim which are incurred by any member of the Group to persons who are not members of the Group.
- (b) Upon the occurrence of:
 - (i) a Change of Control;
 - (ii) Ronald E. Roark resigns as Vice Chairman and/or Group Chief Executive of the Company or ceases to be employed by the Company or to devote the time and attention to the business, trade and offices of the Group or perform the functions required under the terms of is/her service contract; or
 - (iii) the sale of all or substantially all of the assets of the Group, whether in a single transaction or a series of related transactions,

the Facility will be cancelled, the outstanding Utilisations, together with accrued interest, and all other amounts accrued under the Finance Documents shall become immediately due and payable (together with any amount payable under Clause 9.6 (*Make Whole Payments*)) and the Commitments shall be reduced to zero.

(c) Upon the occurrence of the Permitted Reorganisation and upon the occurrence of the Rooftop Sale, the Company shall ensure that the Borrower prepays the Loans (or, if only the Letter of Credit is outstanding, the Letter of Credit) in the amount of the Disposal Proceeds of the Permitted Reorganisation and in

39

the amount of the Disposal Proceeds of the Rooftop Sale immediately upon receipt of such Disposal Proceeds (together with any amount payable under Clause 9.6 (*Make Whole Payments*)).

- (d) Upon the occurrence of any Disposal, the Company shall ensure that the Borrower prepays the Loans (or, if only the Letter of Credit is outstanding, the Letter of Credit) in the amount of the Disposal Proceeds (except for Excluded Disposal Proceeds) within 3 Business Days of receipt of the Disposal Proceeds (together with any amount payable under Clause 9.6 (*Make Whole Payments*)) and pending such prepayment are credited to an account charged in favour of the Secured Parties.
- (e) The Company shall ensure that Acquisition Proceeds and Insurance Proceeds are applied in prepayment of the Loans (or, if only the Letter of Credit is outstanding, the Letter of Credit) within 3 Business Days of their receipt (together with any amount payable under Clause 9.6 (*Make Whole Payments*)) and pending such application are credited to an account charged in favour of the Secured Parties.

9.6 Make Whole Payments

If the Borrower prepays any Loan (or, if only the Letter of Credit is outstanding, the Letter of Credit) (in whole or in part) pursuant to Clause 9.3 (*Voluntary prepayment of loans*) or Clause 9.5 (*Mandatory Prepayment*) at any time up to and including the Make Whole Date, the Borrower shall, on the date of such prepayment, pay to the Agent for the account of the Lenders a sum calculated using the formula:

$$VP \times (M \times \frac{N}{360})$$

$$\frac{1 + (Discount N)}{Rate \times 360}$$

where

VP = aggregate of the principal amount of the Loan(s) (or, if only the Letter of Credit is outstanding, the Letter of Credit) prepaid.

M =the Margin on the date of such prepayment.

N = number of days from (and including) the actual date of prepayment to (but excluding) the Make Whole Date.

Discount Rate = EURIBOR in respect of the prepayment date.

9.7 **Restrictions**

(a) Any notice of cancellation or prepayment given by any Party under this Clause 9.7 shall be irrevocable and, unless a contrary indication appears in this Agreement, shall specify the date or dates upon which the relevant cancellation or prepayment is to be made and the amount of that cancellation or prepayment.

40

Table of Contents

- (b) Any prepayment under this Agreement shall be made together with accrued interest on the principal amount of a Loan prepaid, Break Costs (if any) and any amounts payable under Clause 9.6 (*Make Whole Payments*), but without any other premium or penalty.
- (c) The Borrower may not reborrow any part of the Facility which is prepaid.
- (d) The Borrower shall not repay or prepay all or any part of the Loans or cancel all or any part of the Commitments except at the times and in the manner expressly provided for in this Agreement.
- (e) No amount of the Total Commitments cancelled under this Agreement may be subsequently reinstated.
- (f) If the Agent receives a notice under this Clause 9.7 it shall promptly forward a copy of that notice to either the Borrower or the affected Lender, as appropriate.
- (g) The Agent shall notify the Lenders as soon as possible of any proposed prepayment under this Clause 8.4 and the Company and the Lenders of any calculation pursuant to Clause 9.6 (*Make Whole Payments*).

41

SECTION 5 COSTS OF UTILISATION

10. **INTEREST**

10.1 Calculation of interest

The rate of interest on the Loan for each Interest Period is the percentage rate per annum which is the aggregate of the applicable:

- (a) Margin;
- (b) EURIBOR; and
- (c) Mandatory Cost, if any.

10.2 **Payment of interest**

Subject to Clause 10.5 (*Capitalisation of interest*), the Borrower shall pay accrued interest on the Loan on the last day of each Interest Period.

10.3 **Default interest**

- (a) If an Obligor fails to pay any amount (other than any amount payable in respect of its Dutch Parallel Debts) payable by it under a Finance Document on its due date, interest shall accrue on the overdue amount from the due date up to the date of actual payment (both before and after judgment) at a rate which, subject to paragraph (b) below, is 2 per cent higher than the rate which would have been payable if the overdue amount had, during the period of non-payment, constituted the Loan in the currency of the overdue amount for successive Interest Periods, each of a duration selected by the Agent (acting reasonably). Any interest accruing under this Clause 10.3 shall be immediately payable by the Obligor on demand by the Agent.
- (b) If any overdue amount consists of all or part of the Loan which became due on a day which was not the last day of an Interest Period relating to the Loan:
 - (i) the first Interest Period for that overdue amount shall have a duration equal to the unexpired portion of the current Interest Period relating to the Loan; and
 - (ii) the rate of interest applying to the overdue amount during that first Interest Period shall be 2 per cent. higher than the rate which would have applied if the overdue amount had not become due.
- (c) Default interest (if unpaid) arising on an overdue amount will be compounded with the overdue amount at the end of each Interest Period applicable to that overdue amount but will remain immediately due and payable.

42

10.4 Notification of rates of interest

The Agent shall promptly notify the Lenders and the Borrower of the determination of a rate of interest under this Agreement.

10.5 Capitalisation of interest

- (a) The Borrower may elect to capitalise a portion of the accrued interest on the Loan due and payable on the last day of each Interest Period pursuant to Clause 10.2 (*Payment of interest*) or a portion of the accrued Letter of Credit fee due and payable on any date pursuant to Clause 13.3(b) (*Fees payable in respect of the Letter of Credit*) by serving a Capitalisation Notice on the Agent not later than the Specified Time.
- (b) The portion of the accrued interest or accrued Letter of Credit fee which may be capitalised pursuant to paragraph (a) above shall be an amount calculated using the formula:

8% x O N

x 360

where

- O = in the case of accrued interest, the amount of the outstanding Loan on the last day of the relevant Interest Period and, in the case of accrued Letter of Credit fee, the outstanding amount of the Letter of Credit on the relevant date for payment of such accrued Letter of Credit Fee; and
- N = in the case of accrued interest, the number of days which have elapsed during the relevant Interest Period and in the case of accrued Letter of Credit fee, the number of days which have elapsed since the last payment date for the Letter of Credit fee,
- (the Capitalised Interest).
- (c) Each Capitalisation Notice delivered to the Agent by the Borrower is irrevocable.
- (d) On the last day of each Interest Period in the case of accrued interest and on the relevant date for payment in the case of accrued Letter of Credit fee, the relevant Capitalised Interest (if any) shall be added to the amount of the outstanding Loan or if there is none, shall be treated as a Loan made on that date and payable in accordance with Clause 8.1 (*Repayment of Loans*).

10.6 Increase in Margin

If the Warrants are not issued or any other matter required under Clause 23.36 (a) to (c) (inclusive) (*Conditions Subsequent in relation to Warrants*) is not satisfied prior to the earlier of the Closing Date and the date falling 68 days after the date of this Agreement (the **Margin Increase Date**), the Margin shall increase by 6.00 per cent. per annum with effect from (and excluding) the Margin Increase Date until (and including) the date on which the Warrants are issued and such other matters specified

43

Table of Contents

in Clause 23.36 (Conditions Subsequent in relation to Warrants) are satisfied (the Margin Reduction Date).

11. INTEREST PERIODS

11.1 Selection of Interest Periods

- (a) Subject to the following paragraphs of this Clause 11.1, each Interest Period shall be three Months, ending on a Quarter Date provided that the first Interest Period for a Loan shall end on the first Quarter Date occurring after its Utilisation Date.
- (b) An Interest Period for a Loan shall not extend beyond the Termination Date.
- (c) Each Interest Period for a Loan shall start on its Utilisation Date or (if already made) on the last day of its preceding Interest Period. If another Loan is already outstanding on the Utilisation Date of the first referred to Loan, then the first Interest Period for the first referred to Loan shall end on the same day as the then current Interest Period for such outstanding Loan(s).

11.2 Non-Business Days

If an Interest Period would otherwise end on a day which is not a Business Day, that Interest Period will instead end on the next Business Day in that Month (if there is one) or the preceding Business Day (if there is not).

11.3 Consolidation

If two or more Interest Periods for different Loans end on the same date, these Loans will be consolidated into and treated as a single Loan on the last day of such Interest Periods.

12. CHANGES TO THE CALCULATION OF INTEREST

12.1 Market disruption

- (a) If a Market Disruption Event occurs in relation to a Loan for any Interest Period, then the rate of interest on each Lender s share of the Loan for the Interest Period shall be the percentage rate per annum which is the sum of:
 - (i) the Margin;
 - (ii) the rate notified to the Agent by that Lender as soon as practicable and in any event before interest is due to be paid in respect of that Interest Period, to be that which expresses as a percentage rate per annum the cost to that Lender of funding its participation in the Loan from whatever source it may reasonably select; and
 - (iii) the Mandatory Cost, if any, applicable to that Lender s participation in that Loan.

44

Table of Contents

(b) In this Agreement **Market Disruption Event** means at or about noon on the Quotation Day for the relevant Interest Period the Screen Rate is not available and the Agent is unable to determine EURIBOR for the relevant Interest Period.

12.2 Alternative basis of interest or funding

- (a) If a Market Disruption Event occurs and the Agent or the Borrower so requires, the Agent and the Borrower shall enter into negotiations (for a period of not more than thirty days) with a view to agreeing a substitute basis for determining the rate of interest.
- (b) Any alternative basis agreed pursuant to paragraph (a) above shall, with the prior consent of all the Lenders and the Borrower, be binding on all Parties.

12.3 Break Costs

- (a) The Borrower shall, within three Business Days of demand by a Finance Party, pay to that Finance Party its Break Costs attributable to all or any part of a Loan or Unpaid Sum being paid by the Borrower on a day other than the last day of an Interest Period for that Loan or Unpaid Sum.
- (b) Each Lender shall, as soon as reasonably practicable after a demand by the Agent, provide a certificate confirming the amount of its Break Costs for any Interest Period in which they accrue.

13. **FEES**

13.1 Arrangement fee

The Borrower shall pay to the Arranger (for its own account) an arrangement fee in the amount agreed in a Fee Letter on the earlier of the Closing Date and the last day of the Availability Period.

13.2 Agency fee

The Borrower shall pay to the Agent and the Security Agent (for their own respective accounts) agency fees in the amounts and at the times agreed in a Fee Letter.

13.3 Fees payable in respect of the Letter of Credit

- (a) The Borrower shall pay to the Agent (for the account of each Lender) a Letter of Credit fee in Euro (computed at the rate equal to the Margin) on the outstanding amount of the Letter of Credit requested by it for the period from the issue of the Letter of Credit until its Expiry Date or, if earlier, the date upon which payment of a drawing is made under the Letter of Credit and there is then no further amount to be drawn. This fee shall be distributed according to each Lender s L/C Proportion of the Letter of Credit.
- (b) The accrued Letter of Credit fee on a Letter of Credit shall be payable quarterly in arrears and on the Closing Date. The accrued Letter of Credit fee is also payable to the Agent on the cancelled amount of any Lender s

45

Table of Contents

Commitment at the time the cancellation is effective if that Commitment is cancelled in full and the Letter of Credit is prepaid or repaid in full.

46

SECTION 6 ADDITIONAL PAYMENT OBLIGATIONS

14. TAX GROSS UP AND INDEMNITIES

14.1 Tax definitions

In this Agreement:

Tax Credit means a credit against, relief or remission for, or repayment of any Tax;

Tax Deduction means a deduction or withholding for or on account of Tax from a payment under a Finance Document; and

Tax Payment means either the increase in a payment made by an Obligor to a Finance Party under Clause 14.3(d) (*Tax gross-up*) or a payment made under Clause 14.4(a) (*Tax indemnity*).

14.2 Payments to be free and clear

All payments to be made by each Obligor under each Finance Document shall be paid free and clear of and without any Tax Deduction (in each case except to the extent required by law).

14.3 Tax gross-up

- (a) The Borrower shall, promptly upon it or an Obligor becoming aware that an Obligor is required by law to make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction), notify the Agent of such requirement or change.
- (b) A Lender, upon becoming aware that an Obligor is required by law to make a Tax Deduction (or that there is any change in the rate or the basis of a Tax Deduction in respect of a payment payable to that Lender), shall notify the Agent of this. If the Agent receives such notification from a Lender it shall notify the Borrower.
- (c) If an Obligor is required by law to make a Tax Deduction it shall make that Tax Deduction and any payment required in connection with any Tax Deduction within the time period and in the minimum amount required by law.
- (d) Except as provided by Clause 14.8 (*US filings*) below, if a Tax Deduction is required by law to be made by an Obligor or the Agent the amount of the payment due from the Obligor shall be increased to an amount which ensures that, after the making of that Tax Deduction, each relevant Party receives on the due date and retains (free from any liability in respect of such Tax Deduction) a net sum equal to the amount of the payment which it would have received and so retained had no such Tax Deduction been required.
- (e) Within 30 days after making any Tax Deduction or any payment required in connection with any Tax Deduction, the Obligor making that Tax Deduction

47

Table of Contents

or payment shall deliver to the Agent for the relevant Party evidence satisfactory to that Party (acting reasonably) that the Tax Deduction has been made and that any payment which is required in connection with any Tax Deduction has been made to the relevant Tax authority or other person.

14.4 Tax indemnity

- (a) Except as provided by paragraph (b) and Clause 14.8 (*US filings*) below, the Company shall, or shall procure that an Obligor shall, on demand by the Agent, indemnify a Finance Party against any loss, liability or cost which that Finance Party (in its absolute discretion) determines will be or has been (directly or indirectly) suffered for or on account of Tax by that Finance Party in respect of a Finance Document.
- (b) Paragraph (a) above shall not apply:
 - (i) with respect to any Tax assessed on a Finance Party under the laws of the jurisdiction in which:
 - (A) that Finance Party is incorporated or, if different, the jurisdiction (or jurisdictions) in which that Finance Party is treated as resident for Tax purposes; or
 - (B) that Finance Party s Facility Office is located in respect of amounts received or receivable in that jurisdiction,
 - if that Tax is imposed on or calculated by reference to the overall net income received or receivable by that Finance Party. Any sum deemed to be received or receivable, including, for the avoidance of doubt, any amount treated as income but not actually received by the Finance Party (such as a Tax Deduction), is not income received or receivable for this purpose; or
 - (ii) if and to the extent that a loss, liability or cost is compensated for by an increased payment pursuant to Clause 14.3(d) (*Tax gross up*) or would have been so compensated but was not so compensated solely because one or more of the exclusions contained in paragraph (f) of Clause 14.3 (*Tax gross up*) applied.
- (c) A Finance Party making, or intending to make, a claim under paragraph (a) above shall promptly notify the Agent of the event which will give, or has given, rise to the claim, following which the Agent shall notify the Borrower.
- (d) A Finance Party shall, on receiving a payment from an Obligor under paragraph (a) above, notify the Agent.

14.5 Tax credits

If an Obligor makes a Tax Payment and the relevant Finance Party determines, in its sole opinion, that it has obtained, utilised and retained a Tax Credit which is attributable either to an increased payment of which that Tax Payment forms part or to that Tax Payment, that Finance Party shall pay to the relevant Obligor such amount as

48

Table of Contents

that Finance Party determines, in its sole opinion, to be attributable to the relevant Tax Payment and as will leave that Finance Party (after that payment) in the same after Tax position as it would have been in if the Tax Payment had not been required to be made by that Obligor.

14.6 **Stamp Taxes**

The Borrower shall, or shall procure that an Obligor shall, pay and, within five Business Days of demand by the Agent, indemnify each Finance Party against any cost, loss or liability that Finance Party incurs in relation to all stamp duty, stamp duty reserve, documentary, registration and other similar Taxes payable in connection with or in relation to any Finance Document.

14.7 **VAT**

- (a) All amounts set out in, or expressed to be payable under, a Finance Document by any Party to a Finance Party which (in whole or in part) constitute the consideration for VAT purposes shall be deemed to be exclusive of any VAT which is chargeable on such supply, and accordingly, subject to paragraph
 (c) below, if VAT is chargeable on any supply made by any Finance Party to any Party under a Finance Document, that Party shall pay to the Finance Party (in addition to and at the same time as paying the consideration) an amount equal to the amount of the VAT (and such Finance Party shall promptly provide an appropriate VAT invoice to such Party).
- (b) If VAT is chargeable on any supply made by any Finance Party (the **Recipier**) to any other Finance Party (the **Recipient**) under a Finance Document, and any Party (the **Relevant Party**) is required by the terms of any Finance Document to pay an amount equal to the consideration for such supply to the Supplier (rather than being required to reimburse the Recipient in respect of that consideration), such Party shall also pay to the Supplier (in addition to and at the same time as paying such amount) an amount equal to the amount of such VAT. The Recipient shall promptly pay to the Relevant Party an amount equal to any credit or repayment from the relevant Tax authority which it reasonably determines relates to the VAT chargeable on that supply.
- (c) Where a Finance Document requires any Party to reimburse a Finance Party for any costs or expenses, that Party shall also at the same time pay and indemnify the Finance Party against all VAT incurred by the Finance Party in respect of the costs or expenses to the extent that the Finance Party determines in its sole discretion that neither it nor any other member of any group of which it is a member for VAT purposes is entitled to credit or repayment from the relevant Tax authority in respect of the VAT.

14.8 US filings

(a) Each Lender that is a United States person shall, on or prior to the date of the execution and delivery of this Agreement in the case of an Original Lender and on the date of the Transfer Certificate or Assignment Agreement pursuant to which it becomes a Lender in the case of any other Lender, and from time

Table of Contents

to time thereafter as required by law and as reasonably requested in writing by the Agent or the relevant Obligor (but only so long as such Lender remains lawfully able to do so), provide the Agent or the relevant Obligor with two properly completed and duly executed Internal Revenue Service Forms W-9, as appropriate, or any successor or other form prescribed by the Internal Revenue Service, certifying that such Lender is a United States person and is not subject to United States backup withholding tax on payments made by an Obligor that is a United States person pursuant to this Agreement. For purposes of this Clause 14.8, the term **United States person** shall have the meaning specified in Section 7701(a)(30) of the Internal Revenue Code.

- (b) Each Lender that is not a United States person, on or prior to the date of the execution and delivery of this Agreement in the case of the Original Lender and on the date of the Transfer Certificate or Assignment Agreement pursuant to which it becomes a Lender in the case of any other Lender, and from time to time thereafter as required by law and as reasonably requested in writing by the Agent or the relevant Obligor (but only so long as such Lender remains lawfully able to do so):
 - (i) in the case of a Lender claiming the benefits of an exemption from or reduction in United States federal withholding tax pursuant to a double taxation agreement between the United States and the jurisdiction of which such Lender is or is treated as a resident, provide the Agent or the relevant Obligor with two properly completed and duly executed Internal Revenue Service Forms W-8BEN, or any successor or other form prescribed by the Internal Revenue Service, certifying that such Lender is exempt from or entitled to a reduced rate of United States federal withholding tax under an applicable double taxation agreement on payments made by an Obligor that is a United States person pursuant to this Agreement; or
 - (ii) in the case of a Lender claiming the benefits of an exemption from United States federal withholding tax because the payments otherwise subject to such withholding tax are effectively connected with the Lender s conduct of a trade or business within the United States, provide the Agent or the relevant Obligor with two properly completed and duly executed Internal Revenue Service Forms W-8ECI, or any successor or other form prescribed by the Internal Revenue Service, certifying that such payments are effectively connected with the conduct of a trade or business within the United States; or
 - (iii) in the case of a Lender claiming the benefits of the exemption from United States federal withholding tax pursuant to Section 881(c) of the Internal Revenue Code with respect to payments of portfolio interest made by an Obligor that is a United States person pursuant to this Agreement, provide the Agent or the relevant Obligor with (x) a certificate to the effect that such Lender is neither (i) a bank (within the meaning of Section 881(c)(3)(A) of the Internal Revenue Code), (ii) a 10-percent shareholder of the relevant Obligor (within the meaning of Section 881(c)(3)(B) of the Internal Revenue Code) nor (iii) a controlled foreign corporation related to the relevant Obligor (as

- such term is described in Section 881(c)(3)(C) of the Internal Revenue Code), and (y) two properly completed and duly executed copies of Internal Revenue Service Forms W-8BEN or any successor or other form prescribed by the Internal Revenue Service, certifying that the Lender is a non-United States person; or
- (iv) in the case of a Lender that is a foreign intermediary or foreign flow-through entity for United States federal income tax purposes, provide the Agent or the relevant Obligor with respect to such Lender two properly completed and duly executed Internal Revenue Service Forms W-8IMY or any successor or other form prescribed by the Internal Revenue Service, as a basis for claiming exemption from or reduction in United States federal withholding tax on payments made pursuant to this Agreement by an Obligor that is a United States person, together with any supplementary information such Lender is required to transmit with such form and, in the case of a nonqualified intermediary that is a Lender or a non-withholding Lender that is a foreign flow-through entity, with respect to each beneficiary or member of such Lender, two forms or certificates described in sub-clause (i), (ii) or (iii) above of this Clause 14.8, as applicable.
- (c) If a Lender fails to provide the Agent or the relevant Obligor with the appropriate Internal Revenue Service form or, if applicable, the certificate, each as described above and each being properly completed and duly executed, or to update them as appropriate (other than if the failure to furnish such form or certificate is due to a change in law, or in the interpretation or application thereof, occurring after the date on which the form or certificate originally was required to be provided or if such form, certificate or other document otherwise is not required under subsection (a) or (b) of this Clause 14.8), United States backup withholding tax and United States federal withholding tax, in each case, imposed on any amount paid by an Obligor that is a United States person under this Agreement, shall be considered excluded from the gross-up at Clause 14.3 (Tax gross up) and the indemnity at Clause 14.4 (Tax indemnity) under this Agreement by reason of such failure unless and until the Lender provides the appropriate Internal Revenue Service form or certificate that is properly completed and duly executed and establishing (x) an exemption from United States backup withholding tax and (y) a complete exemption from, or a reduction of, United States federal withholding tax on any amount paid by the relevant Obligor under this Agreement, whereupon United States federal withholding tax at such reduced rate only (to the extent a complete exemption is not available to such Lender) shall be considered excluded from such gross-up and indemnity for periods governed by such form and certificate. If any Internal Revenue Service form provided by a Lender pursuant to this Clause 14.8 at the time such Lender first becomes a Lender hereunder or when it first provides such form indicates a United States federal withholding tax rate in excess of zero in respect of interest paid by the relevant Obligor under this Agreement, United States federal withholding tax at such rate shall be considered excluded from the gross up at Clause 14.3(d) (Tax gross up) and the indemnity at Clause 14.4(a) (Tax indemnity) under this Agreement unless and until such Lender provides the appropriate form

51

certifying that a lesser rate applies, whereupon United States federal withholding tax at such lesser rate only shall be considered excluded from such gross up or indemnity for periods governed by such form; provided, however, that if at the date a Lender transferee under a Transfer Certificate or assignee under an Assignment Agreement becomes a party to this Agreement, the Lender transferor was entitled to payments under Clauses 14.3 (Tax gross up) or 14.4 (Tax indemnity) in respect of United States federal withholding tax in connection with interest paid at such date, then, to that extent, the payments under Clause 14.3 (Tax gross up) or indemnity under Clause 14.4 (Tax indemnity) shall include (in addition to withholding taxes that may be imposed in the future or other amounts otherwise included therein) an amount of United States federal withholding tax applicable with respect to the Lender transferor on such date. If any form, certificate or document referred to in this Clause 14.8 requires the disclosure of information, other than information necessary to compute the Tax payable and information required on the date hereof by Internal Revenue Service Forms W-9, W-8BEN, W-8ECI, or W-8IMY (as applicable) or the related certificate described above, that a Lender reasonably considers to be confidential, such Lender shall give notice thereof to the Agent or the relevant Obligor and shall not be obligated to include in any such form or document any such confidential information.

(d) Without prejudice to the survival of any other agreement of the Obligors hereunder, the agreements and obligations of the Obligor that is a United States person contained in this Clause 14.8 shall survive the payment in full of principal, interest and all other amounts payable hereunder and the termination of this Agreement.

15. INCREASED COSTS

15.1 Increased costs

- (a) Subject to Clause 15.3 (*Exceptions*) the Borrower shall, within five Business Days of a demand by the Agent, pay for the account of a Finance Party the amount of any Increased Costs incurred by that Finance Party or any of its Affiliates as a result of (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation or (ii) compliance with any law or regulation made after the date of this Agreement.
- (b) In this Agreement Increased Costs means:
 - (i) a reduction in the rate of return from the Facility or on a Finance Party s (or its Affiliate s) overall capital;
 - (ii) an additional or increased cost; or
 - (iii) a reduction of any amount due and payable under any Finance Document, which is incurred or suffered by a Finance Party or any of its Affiliates to the extent that it is attributable to that Finance Party having entered into its

52

Table of Contents

Commitment or funding or performing its obligations under any Finance Document.

15.2 Increased cost claims

- (a) A Finance Party intending to make a claim pursuant to Clause 15.1 (*Increased costs*) shall notify the Agent of the event giving rise to the claim, following which the Agent shall promptly notify the Borrower.
- (b) Each Finance Party shall, as soon as practicable after a demand by the Agent, provide a certificate confirming the amount of its Increased Costs.

15.3 Exceptions

- (a) Clause 15.1 (*Increased costs*) does not apply to the extent any Increased Cost is:
 - (i) attributable to a Tax Deduction required by law to be made by an Obligor;
 - (ii) compensated for by Clause 14.4 (*Tax indemnity*) (or would have been compensated for under Clause 14.4 (*Tax indemnity*) but was not so compensated solely because any of the exclusions in paragraph (b) of Clause 14.4 (*Tax indemnity*) applied);
 - (iii) compensated for by the payment of the Mandatory Cost; or
 - (iv) attributable to the wilful breach by the relevant Finance Party or its Affiliates of any law or regulation.
- (b) In this Clause 15.3, a reference to a **Tax Deduction** has the same meaning given to the term in Clause 14.1 (*Tax definitions*).

16. **OTHER INDEMNITIES**

16.1 **Currency indemnity**

- (a) If any sum due from an Obligor or the Acquiror under the Finance Documents (a **Sum**), or any order, judgment or award given or made in relation to a Sum, has to be converted from the currency (the **First Currency**) in which that Sum is payable into another currency (the **Second Currency**) for the purpose of:
 - (i) making or filing a claim or proof against that Obligor;
 - (ii) obtaining or enforcing an order, judgment or award in relation to any litigation or arbitration proceedings,

that Obligor shall and the Company shall procure that the Acquior shall, as an independent obligation, within three Business Days of demand, indemnify each Finance Party to whom that Sum is due against any cost, loss or liability arising out of or as a result of the conversion including any discrepancy

53

Table of Contents

- between (A) the rate of exchange used to convert that Sum from the First Currency into the Second Currency and (B) the rate or rates of exchange available to that person at the time of its receipt of that Sum.
- (b) Each Obligor waives and the Company shall procure that the Acquiror waives, any right it may have in any jurisdiction to pay any amount under the Finance Documents in a currency or currency unit other than that in which it is expressed to be payable.

16.2 Other indemnities

The Borrower shall (or shall procure that an Obligor will), within three Business Days of demand, indemnify each Finance Party against any cost, loss or liability incurred by that Finance Party as a result of:

- (a) the occurrence of any Event of Default;
- (b) a failure by an Obligor to pay any amount due under a Finance Document on its due date, including without limitation, any cost, loss or liability arising as a result of Clause 28 (*Sharing among the Finance Parties*);
- (c) funding, or making arrangements to fund, its participation in a Utilisation requested by the Borrower in a Utilisation Request but not made by reason of the operation of any one or more of the provisions of this Agreement (other than by reason of default or negligence by that Lender alone);
- (d) a Utilisation (or part of a Utilisation) not being prepaid in accordance with a notice of prepayment given by the Borrower; or
- (e) issuing or making arrangements to issue the Letter of Credit requested by the Borrower in a Utilisation Request but not issued by reason of the operation of one or more of the provisions of this Agreement.

16.3 **Indemnity to the Agent**

The Borrower shall promptly indemnify the Agent against any cost, loss or liability incurred by the Agent (acting reasonably) as a result of:

- (a) investigating any event which it reasonably believes is a Default; or
- (b) acting or relying on any notice, request or instruction which it reasonably believes to be genuine, correct and appropriately authorised.

16.4 Indemnity to the Security Agent

- (a) Each Obligor shall promptly indemnify the Security Agent and every Receiver and Delegate against any cost, loss or liability incurred by any of them as a result of:
 - (i) the taking, holding, protection or enforcement of the Transaction Security,

54

Table of Contents

- (ii) the exercise of any of the rights, powers, discretions and remedies vested in the Security Agent and each Receiver and Delegate by the Finance Documents or by law; and
- (iii) any default by any Obligor or the Acquiror in the performance of any of the obligations expressed to be assumed by it in the Finance Documents.
- (b) The Security Agent may, in priority to any payment to the Secured Parties, indemnify itself out of the Charged Property in respect of, and pay and retain, all sums necessary to give effect to the indemnity in this Clause 16.4 and shall have a lien on the Transaction Security and the proceeds of the enforcement of the Transaction Security for all monies payable to it.

17. MITIGATION BY THE LENDERS

17.1 **Mitigation**

- (a) Each Finance Party shall, in consultation with the Borrower, take all reasonable steps to mitigate any circumstances which arise and which would result in any amount becoming payable under or pursuant to, or cancelled pursuant to, any of Clause 9.1 (*Illegality*), Clause 14 (*Tax gross-up and indemnities*), Clause 15.1 (*Increased costs*) or paragraph 3 of Schedule 4 (*Mandatory Cost formulae*) including (but not limited to) transferring its rights and obligations under the Finance Documents to another Affiliate or Facility Office.
- (b) Paragraph (a) above does not in any way limit the obligations of any Obligor under the Finance Documents.

17.2 Limitation of liability

- (a) The Borrower shall indemnify each Finance Party for all costs and expenses reasonably incurred by that Finance Party as a result of steps taken by it under Clause 17.1 (*Mitigation*).
- (b) A Finance Party is not obliged to take any steps under Clause 17.1 (*Mitigation*) if, in the opinion of that Finance Party (acting reasonably), to do so might be prejudicial to it.

18. COSTS AND EXPENSES

18.1 Transaction expenses

The Borrower shall promptly on demand pay the Agent, the Arranger, the Issuing Bank and the Security Agent the amount of all costs and expenses (including legal fees) reasonably incurred by any of them in connection with the negotiation, preparation, printing, execution and syndication of:

(a) this Agreement and any other documents referred to in this Agreement and the Transaction Security; and

55

Table of Contents

(b) any other Finance Documents executed after the date of this Agreement.

18.2 Amendment costs

If (a) an Obligor requests an amendment, waiver or consent or (b) an amendment is required pursuant to Clause 29.9 (*Change of currency*), the Borrower shall, within five Business Days of demand, reimburse the Agent and the Security Agent for the amount of all costs and expenses (including legal fees) reasonably incurred by the Agent and the Security Agent (and, in the case of the Security Agent, by any Receiver or Delegate) in responding to, evaluating, negotiating or complying with that request or requirement.

18.3 Enforcement and preservation costs

The Company shall, within five Business Days of demand, pay to the Arranger and each other Secured Party the amount of all costs and expenses (including legal fees) incurred by it in connection with the enforcement of or the preservation of any rights under any Finance Document and the Transaction Security and any proceedings instituted by or against the Security Agent as a consequence of taking or holding the Transaction Security or enforcing these rights.

56

SECTION 7 GUARANTEE

19. GUARANTEE AND INDEMNITY

19.1 Guarantee and indemnity

Each Guarantor irrevocably and unconditionally jointly and severally:

- (a) guarantees to each Finance Party punctual performance by the Borrower of all its obligations under the Finance Documents:
- (b) undertakes with each Finance Party that whenever the Borrower does not pay any amount when due under or in connection with any Finance Document, that Guarantor shall promptly within three Business Days of demand pay that amount as if it was the principal obligor; and
- (c) indemnifies each Finance Party promptly within three Business Days of demand against any cost, loss or liability suffered by that Finance Party if any obligation guaranteed by it is or becomes unenforceable, invalid or illegal. The amount of the cost, loss or liability shall be equal to the amount which that Finance Party would otherwise have been entitled to recover.

19.2 Continuing guarantee

This guarantee is a continuing guarantee and will extend to the ultimate balance of sums payable by any Obligor under the Finance Documents, regardless of any intermediate payment or discharge in whole or in part.

19.3 Reinstatement

If any payment by an Obligor or any discharge given by a Finance Party (whether in respect of the obligations of any Obligor or any security for those obligations or otherwise) is avoided or reduced as a result of insolvency or any similar event:

- (a) the liability of each Obligor shall continue as if the payment, discharge, avoidance or reduction had not occurred; and
- (b) each Finance Party shall be entitled to recover the value or amount of that security or payment from each Obligor, as if the payment, discharge, avoidance or reduction had not occurred.

19.4 Waiver of defences

The obligations of each Guarantor under this Clause 19 will not be affected by an act, omission, matter or thing which, but for this Clause, would reduce, release or prejudice any of its obligations under this Clause 19 (without limitation and whether or not known to it or any Finance Party) including:

(a) any time, waiver or consent granted to, or composition with, any Obligor or the Acquiror or other person;

57

Table of Contents

- (b) the release of any other Obligor or the Acquiror or any other person under the terms of any composition or arrangement with any creditor of any member of the Group;
- (c) the taking, variation, compromise, exchange, renewal or release of, or refusal or neglect to perfect, take up or enforce, any rights against, or security over assets of, any Obligor or the Acquiror or other person or any non-presentation or non-observance of any formality or other requirement in respect of any instrument or any failure to realise the full value of any security;
- (d) any incapacity or lack of power, authority or legal personality of or dissolution or change in the members or status of an Obligor or the Acquiror or any other person;
- (e) any amendment, novation, supplement, extension, restatement (however fundamental and whether or not more onerous) or replacement of any Finance Document or any other document or security including without limitation any change in the purpose of, any extension of or any increase in any facility or the addition of any new facility under any Finance Document or other document or security;
- (f) any unenforceability, illegality or invalidity of any obligation of any person under any Finance Document or any other document or security; or
- (g) any insolvency or similar proceedings.

19.5 **Guarantor intent**

Without prejudice to the generality of Clause 19.4 (*Waiver of defences*), each Guarantor expressly confirms that it intends that this guarantee shall extend from time to time to any (however fundamental) variation, increase, extension or addition of or to any of the Finance Documents and/or any facility or amount made available under any of the Finance Documents for the purposes of or in connection with any of the following: (i) acquisitions of any nature; (ii) increasing working capital enabling investor distributions to be made; (iii) carrying out restructurings; (iv) refinancing existing facilities; (v) refinancing any other indebtedness; (vi) making facilities available to new borrowers; (vii) any other variation or extension of the purposes for which any such facility or amount might be made available from time to time; and (viii) any fees, costs and/or expenses associated with any of the foregoing.

19.6 Immediate recourse

Each Guarantor waives any right it may have of first requiring any Finance Party (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from that Guarantor under this Clause 19. This waiver applies irrespective of any law or any provision of a Finance Document to the contrary.

19.7 **Appropriations**

58

Table of Contents

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full, each Finance Party (or any trustee or agent on its behalf) may:

- (a) refrain from applying or enforcing any other moneys, security or rights held or received by that Finance Party (or any trustee or agent on its behalf) in respect of those amounts, or apply and enforce the same in such manner and order as it sees fit (whether against those amounts or otherwise) and no Guarantor shall be entitled to the benefit of the same; and
- (b) hold in an interest-bearing suspense account any moneys received from any Guarantor or on account of any Guarantor s liability under this Clause 19.

19.8 Deferral of Guarantors rights

Until all amounts which may be or become payable by the Obligors under or in connection with the Finance Documents have been irrevocably paid in full and unless the Agent otherwise directs, no Guarantor will exercise any rights which it may have by reason of performance by it of its obligations under the Finance Documents:

- (a) to be indemnified by an Obligor or the Acquiror;
- (b) to claim any contribution from the Acquiror or any other guarantor of any Obligor s obligations under the Finance Documents; and/or
- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under the Finance Documents or of any other guarantee or security taken pursuant to, or in connection with, the Finance Documents by any Finance Party.

19.9 Release of Guarantors right of contribution

If any Guarantor (a **Retiring Guarantor**) ceases to be a Guarantor in accordance with the terms of the Finance Documents for the purpose of any sale or other disposal of that Retiring Guarantor then on the date such Retiring Guarantor ceases to be a Guarantor:

- (a) that Retiring Guarantor is released by each other Guarantor from any liability (whether past, present or future and whether actual or contingent) to make a contribution to any other Guarantor arising by reason of the performance by any other Guarantor of its obligations under the Finance Documents; and
- (b) each other Guarantor waives any rights it may have by reason of the performance of its obligations under the Finance Documents to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of the Finance Parties under any Finance Document or of any other security taken pursuant to, or in connection with, any Finance Document where such rights or security are granted by or in relation to the assets of the Retiring Guarantor.

19.10 Additional security

59

Table of Contents

This guarantee is in addition to and is not in any way prejudiced by any other guarantee or security now or subsequently held by any Finance Party.

19.11 Other Guarantors

It is specifically acknowledged and agreed that the Finance Parties may from time to time make any arrangement, compromise, waiver or other dealing with any Obligor in relation to any guarantee or other obligations under the Finance Documents which such Finance Parties may think fit and no such arrangement, compromise, waiver or other dealing shall exonerate or discharge any other Obligor from its obligations under the Finance Documents.

19.12 Guarantee Limitations

- (a) The guarantee and indemnity obligations of each Guarantor incorporated in England and Wales under this Clause 19 do not apply to any liability to the extent that it would result in the guarantee and indemnity constituting unlawful financial assistance within the meaning of Section 151 of the Companies Act 1985.
- (b) Each of the Parties hereby confirms that it is the intention of all such persons that the obligations of each US Guarantor organised under the laws of any state of the United States of America (a US Guarantor) under this Clause 19 do not constitute a fraudulent transfer or conveyance for the purposes of any proceeding of the type referred to in Clause 24.6 (Insolvency) or 24.7 (Insolvency proceedings) or Title 11, US Bankruptcy Code, the United States Uniform Fraudulent Conveyance Act, the United States Uniform Fraudulent Transfer Act or any similar foreign or state law, to the extent applicable to the obligations of a US Guarantor under this Clause 19. To effect the foregoing intention, the Agent, the Arranger, the Issuing Bank, the Security Agent, the Lenders and the Guarantors hereby irrevocably agree that the obligations of each US Guarantor at any time shall be limited to the maximum amount as will result in the obligations of such US Guarantor under this Clause 19 not constituting a fraudulent transfer or conveyance.

60

SECTION 8 REPRESENTATIONS, UNDERTAKINGS AND EVENTS OF DEFAULT

20. REPRESENTATIONS

20.1 General

- (a) Each Obligor makes the representations and warranties set out in this Clause 20 to each Finance Party.
- (b) In relation to the representations and warranties made on the date of this Agreement and any other date on or before the Closing Date, it is assumed that Closing has occurred.

20.2 Status

- (a) It and each of its Subsidiaries is a limited liability corporation, or, in the case of Regal Mortgages LLP, a limited liability partnership, duly incorporated or established and validly existing under the law of its jurisdiction of incorporation or establishment. In the case of the Company, the Acquior is a private company with limited liability (besloten vennootschap met beperkte aansprakelijkheid) or a limited liability company (naamloze vennootschap).
- (b) It and each of its Subsidiaries has the power to own its assets and carry on its business as it is being conducted.

20.3 **Binding obligations**

Subject to the Legal Reservations:

- (a) the obligations expressed to be assumed by it or any of its relevant Subsidiaries in each Transaction Document to which it is, or such Subsidiary is, a party are legal, valid, binding and enforceable obligations; and
- (b) without limiting the generality of paragraph (a) above, each Transaction Security Document to which it is or any of the Subsidiaries is a party creates the security interests which that Transaction Security Document purports to create and those security interests are valid and effective.

20.4 Non-conflict with other obligations

The entry into and performance by it and, in the case of the Borrower, the Acquiror, of, and the transactions contemplated by, the Transaction Documents and the granting of the Transaction Security do not and will not conflict with:

- (a) any law or regulation applicable to it;
- (b) the constitutional documents of any member of the Group; or
- (c) any agreement or instrument binding upon it or any member of the Group or any of its or any member of the Group s assets or constitute a default or

61

Table of Contents

termination event (however described) under any such agreement or instrument.

20.5 Power and authority

Subject to the Legal Reservations:

- (a) it and each of its relevant Subsidiaries has the power to enter into, perform and deliver, and has taken all necessary action to authorise its or such Subsidiary s entry into, performance and delivery of, the Transaction Documents to which it or such Subsidiary is or will be a party and the transactions contemplated by those Transaction Documents; and
- (b) no limit on its powers or, in the case of the Company, the Acquiror s powers, will be exceeded as a result of the borrowing, grant of security or giving of guarantees or indemnities contemplated by the Transaction Documents to which it is a party.

20.6 Validity and admissibility in evidence

- (a) All Authorisations required or desirable:
 - (i) to enable it and each of its relevant Subsidiaries lawfully to enter into, exercise its or such relevant Subsidiary s rights and comply with its or such Subsidiary s obligations in the Transaction Documents to which it or such Subsidiary is a party; and
 - (ii) to make the Transaction Documents to which it is or, in the case of the Company, the Acquiror is a party admissible in evidence in its Relevant Jurisdictions,
 - have been obtained or effected and are in full force and effect except any Authorisation referred to in paragraph (b) of Clause 20.9 (*No filing or stamp taxes*), which Authorisations will be promptly obtained or effected after the date of this Agreement.
- (b) All Authorisations necessary for the conduct of the business, trade and ordinary activities of members of the Group have been obtained or effected and are in full force and effect and each Regulated Entity is in compliance with all, and no regulatory body is investigating any of, the capital and liquidity requirements applicable to it.

20.7 Governing law and enforcement

- (a) The choice of governing law of the Finance Documents will be recognised and enforced in its Relevant Jurisdictions.
- (b) Any judgment obtained in relation to a Finance Document in the jurisdiction of the governing law of that Finance Document will be recognised and enforced in its Relevant Jurisdictions.

62

Table of Contents

20.8 Insolvency

No:

- (a) corporate action, legal proceeding or other procedure or step described in paragraph (a) of Clause 24.6 (*Insolvency*); or
- (b) creditors process described in Clause 24.8 (*Creditors process*), has been taken or, to the knowledge of the Company, threatened in relation to a member of the Group; and none of the circumstances described in Clause 24.6 (*Insolvency*) applies to a member of the Group.

20.9 No filing or stamp taxes

Under the laws of its Relevant Jurisdiction it is not necessary that the Finance Documents be filed, recorded or enrolled with any court or other authority in that jurisdiction or that any stamp, registration, notarial or similar Taxes or fees be paid on or in relation to the Finance Documents or the transactions contemplated by the Finance Documents:

- (a) except:
 - (i) registration of particulars of the English Law Debenture and English Law Share Charge (referred to in paragraph 3(d) of Part 1 of Schedule 2) at the Companies Registration Office in England and Wales under Section 395 of The Companies Act 1985 and payment of associated fees; and
 - (ii) registration of particulars of the English Law Debenture (referred to in paragraph 3(d) of Part 1 of Schedule 2) at the Trade Marks Registry at the Patent Office in England and Wales and payment of associated fees;
 - which registrations will be made promptly after the date of the relevant Finance Document;
- (b) except any filing, recording or enrolling or any tax or fee payable in relation to the US Security Agreement referred to in paragraph 3(d) of Part 1 of Schedule 2 which is referred to in any Legal Opinion and which will be made or paid promptly after the date of the relevant Finance Document; and
- (c) the filing of a copy of the Facility Agreement with the Securities and Exchange Commission of the US which filing will be made promptly if so required.

20.10 **Deduction of Tax**

It is not required and, in the case of the Company, the Acquiror is not required to make any deduction for or on account of Tax from any payment it may make under any Finance Document.

63

Table of Contents

20.11 No default

- (a) No Event of Default and, on the date of this Agreement and the Closing Date, no Default is continuing or is reasonably likely to result from the making of any Utilisation or the entry into, the performance of, or any transaction contemplated by, any Transaction Document.
- (b) No other event or circumstance is outstanding which constitutes (or, with the expiry of a grace period, the giving of notice, the making of any determination or any combination of any of the foregoing, would constitute) a default or termination event (however described) under any other agreement or instrument which is binding on it or any of its Subsidiaries or to which its (or any of its Subsidiaries) assets are subject which has or is reasonably likely to have a Material Adverse Effect.

20.12 No misleading information

Save as disclosed in writing to the Agent and the Arranger prior to the date of this Agreement:

- (a) any factual information contained in the Reports and the Base Case Model (the **Information Package**) was true and accurate in all material respects as at the date of the relevant report or document containing the information or (as the case may be) as at the date the information is expressed to be given.
- (b) the Base Case Model has been prepared in accordance with the Accounting Principles as applied to the Original Financial Statements, and the financial projections contained in the Base Case Model have been prepared on the basis of recent historical information, are fair and based on reasonable assumptions and have been approved by the board of directors of the Company.
- (c) any financial projection or forecast contained in the Information Package has been prepared on the basis of recent historical information and on the basis of reasonable assumptions and was fair (as at the date of the relevant report or document containing the projection or forecast) and arrived at after careful consideration.
- (d) the expressions of opinion or intention provided by or on behalf of an Obligor or the Acquiror for the purposes of the Information Package were made after careful consideration and (as at the date of the relevant report or document containing the expression of opinion or intention) were fair and based on reasonable grounds.
- (e) no event or circumstance has occurred or arisen and no information has been omitted from the Information Package and no information has been given or withheld that results in the information, opinions, intentions, forecasts or projections contained in the Information Package being untrue or misleading in any material respect.

64

Table of Contents

- (f) all material information provided to a Finance Party by or on behalf of the Company in connection with the Acquisition and/or the Target Group on or before the date of this Agreement and not superseded before that date (whether or not contained in the Information Package) is accurate and not misleading in any material respect and all projections provided to any Finance Party on or before the date of this Agreement have been prepared in good faith on the basis of assumptions which were reasonable at the time at which they were prepared and supplied.
- (g) all other written information provided by any member of the Group (including its advisers) to a Finance Party or the provider of any Report was true, complete and accurate in all material respects as at the date it was provided and is not misleading in any respect.

The representations and warranties made with respect to the Reports are made by each Obligor in this Clause 20.12 only so far as it is aware after making due and careful enquiries.

20.13 Original Financial Statements

- (a) Its Original Financial Statements were prepared in accordance with the Accounting Principles consistently applied.
- (b) Its unaudited Original Financial Statements fairly represent its financial condition and results of operations (consolidated in the case of Target) for the relevant month.
- (c) Its audited Original Financial Statements give a true and fair view of its financial condition and results of operations (consolidated in the case of Target) during the relevant financial year.
- (d) There has been no material adverse change in its or the Target s assets, business or financial condition (or the assets, business or consolidated financial condition of the Group, in the case of the Company) since 31 December, 2005.
- (e) The Original Financial Statements of the Target do not consolidate the results, assets or liabilities of any person or business which does not form part of the Target Shares.
- (f) Its (and in the case of the Company, the Acquiror s) most recent financial statements delivered pursuant to Clause 21.1 (*Financial statements*):
 - (i) have been prepared in accordance with the Accounting Principles as applied to the Original Financial Statements and the Base Case Model; and
 - (ii) give a true and fair view of (if audited) or fairly present (if unaudited) its consolidated financial condition as at the end of, and consolidated results of operations for, the period to which they relate.

65

Table of Contents

- (g) The budgets and forecasts supplied under this Agreement were arrived at after careful consideration and have been prepared in good faith on the basis of recent historical information and on the basis of assumptions which were reasonable as at the date they were prepared and supplied.
- (h) Since the date of the most recent financial statements delivered pursuant to Clause 21.1 (*Financial statements*) there has been no material adverse change in the business, assets or financial condition of the Group.

20.14 No proceedings pending or threatened

No litigation, arbitration or administrative proceedings or investigations of, or before, any court, arbitral body or agency which, if adversely determined, are reasonably likely to have a Material Adverse Effect have (to the best of its knowledge and belief (having made all reasonable enquiries)) been started or threatened against it or any of its Subsidiaries.

20.15 No breach of laws

- (a) It has not (and none of its Subsidiaries has) breached any law or regulation to which it or its Subsidiaries (as applicable) is subject which breach has or is reasonably likely to have a Material Adverse Effect.
- (b) No labour disputes are current or, to the best of its knowledge and belief (having made all reasonable enquiries), threatened against any member of the Group which have or are reasonably likely to have a Material Adverse Effect.

20.16 Environmental laws

- (a) Each member of the Group is in compliance with Clause 23.3 (*Environmental compliance*) and to the best of its knowledge and belief (having made all reasonable enquiries) no circumstances have occurred which would prevent such compliance in a manner or to an extent which has or is reasonably likely to have a Material Adverse Effect.
- (b) No Environmental Claim has been commenced or (to the best of its knowledge and belief (having made all reasonable enquiries)) is threatened against any member of the Group where that claim has or is reasonably likely, if determined against that member of the Group, to have a Material Adverse Effect.
- (c) The cost to the Group of compliance with Environmental Laws (including Environmental Permits) is (to the best of its knowledge and belief, having made all reasonable enquiries) adequately provided for in the Base Case Model.

20.17 Taxation

(a) It is not (and none of its Subsidiaries is) materially overdue in the filing of any Tax reports or returns or liable to pay any material tax penalties and it is not (and none of its Subsidiaries is) overdue in the payment of any amount in respect of Tax save where the payment of such Taxes is contested in good

Table of Contents

faith by a member of the Group and where full provision has been made in the accounts of such Group member for the relevant Tax liabilities and such non-payment will not result in a Security ranking prior to the Transaction Security.

- (b) No claims, tax audits or investigations are being, nor, to the best of its knowledge and belief (having made all reasonable enquiries), are reasonably likely to be, made or conducted against it or any of its Subsidiaries with respect to Tax which would result, or be reasonably likely to result, in liabilities of or claims against any members of the Group which would have a Material Adverse Effect or result in the creation of a Security ranking prior to the Transaction Security.
- (c) Each of the Obligors and, in the case of the Company, the Acquiror is resident for Tax purposes only in the jurisdiction of its incorporation.

20.18 Security and Financial Indebtedness

- (a) No Security exists over all or any of its present or future assets or shares or the present or future assets or shares of any of its Subsidiaries other than a Permitted Security.
- (b) Neither it nor any of its Subsidiaries has any outstanding Financial Indebtedness or guarantees other than a Permitted Financial Indebtedness.

20.19 Ranking

The Transaction Security has or will have the ranking in priority which it is expressed to have in the Transaction Security Documents and it is not subject to any prior ranking or pari passu ranking Security.

20.20 Good title to assets

It and, so far as it is aware, each of its Subsidiaries has a good, valid and marketable title to, or valid leases or licences of, and all appropriate Authorisations to use, the assets necessary to carry on its business as presently conducted.

20.21 Legal and beneficial ownership

- (a) It and, so far as it is aware, each of its Subsidiaries is the sole legal and beneficial owner of the respective assets over which it purports to grant Security.
- (b) All the Target Shares are or will be on the Closing Date legally and beneficially owned by the Acquiror free from any claims, third party rights or competing interests other than Permitted Security permitted under Clause 23.14 (*Negative pledge*).

20.22 Shares

The shares and, if applicable, limited partnership interests of any member of the Group which are (or are required by this Agreement to become) subject to the Transaction Security are (or will on the Closing Date be) issued, fully paid (and, in

67

Table of Contents

the case of limited partnership interests, have not been repaid), non-assessable and are not subject to any option to purchase or similar rights. The constitutional documents of companies whose shares are (or are required by this Agreement to become) subject to the Transaction Security do not and could not restrict or inhibit any transfer of those shares on creation or enforcement of the Transaction Security, save as required by law and specified in the Legal Opinions required to be delivered in connection with the relevant Transaction Security pursuant to this Agreement. Except as provided in the Warrant Documents, there are no agreements in force which provide for the issue or allotment of, or grant any person the right to call for the issue or allotment of, any share or loan capital of any member of the Group or the Target (including any conditional right and any option or right of pre-emption, exchange or conversion).

20.23 Structure Memorandum

- (a) The Structure Memorandum accurately records the shareholding structure of the Group as at the date of this Agreement and as it will be immediately following Closing and immediately following completion of Transaction 1, Transaction 2 and Transaction 3 (as each are defined in the BT Treasury Letter), and includes the following information:
 - (i) accurate details of each member of the Group, including its current name and company registration number, its jurisdiction of incorporation and/or establishment, a list of its shareholders and their percentage ownership or is not a company with limited liability; and
 - (ii) accurate details of any minority interests in any member of the Group held by any person who is not a member of the Group and accurate details of all companies, joint ventures, partnerships and other persons in which any member of the Group has an interest or participation.
- (b) All necessary intra-Group loans, transfers, share exchanges and other steps resulting in the final Group structure are set out in the Structure Memorandum and have been or will be made or taken in compliance with all relevant laws and regulations and all requirements of relevant regulatory authorities.
- (c) Crown Intressenter AB is in the course of being liquidated and dissolved.

20.24 Intellectual Property

It and each of its Subsidiaries:

- (a) is the sole legal and beneficial owner of or has licensed to it on normal commercial terms all the Intellectual Property which is material in the context of its business and which is required by it in order to carry on its business as it is being conducted and as contemplated in the Base Case Model;
- (b) does not (nor does any of its Subsidiaries), in carrying on its businesses, infringe any Intellectual Property of any third party in any respect which has or is reasonably likely to have a Material Adverse Effect; and

Table of Contents

(c) has taken all formal or procedural actions (including payment of fees) required to maintain any material Intellectual Property owned by it.

20.25 Accounting reference date

The accounting reference date of each member of the Group is 31 December.

20.26 Acquisition Documents, disclosures and other Documents

- (a) The Acquisition Documents contain all the terms of the Acquisition.
- (b) There is no disclosure made to the Acquisition Documents which has or may have a material adverse effect on any of the information, opinions, intentions, forecasts and projections contained or referred to in the Information Package.
- (c) To the best of its knowledge no representation or warranty given by any party to the Acquisition Documents is untrue or misleading in any material respect.

20.27 Pensions

- (a) Neither it nor any of its Subsidiaries is or has at any time been an employer (for the purposes of Sections 38 to 51 of the Pensions Act 2004) of an occupational pension scheme which is not a money purchase scheme (both terms as defined in the Pensions Schemes Act 1993); and
- (b) Neither it nor any of its Subsidiaries is or has at any time been connected with or an associate of (as those terms are used in Sections 39 and 43 of the Pensions Act 2004) such an employer.
- (c) The pension schemes of each member of the Group are funded to the extent required by law or otherwise comply with the requirements of any material law applicable in the jurisdiction in which the relevant pension scheme is maintained, in each case, where failure to do so could reasonably be expected to have a Material Adverse Effect.

20.28 No adverse consequences

- (a) It is not necessary under the laws of its Relevant Jurisdictions:
 - (i) in order to enable any Finance Party to enforce its rights under any Finance Document; or
 - (ii) by reason of the execution of any Finance Document or the performance by it of its obligations under any Finance Document,

that any Finance Party should be licensed, qualified or otherwise entitled to carry on business in any of its Relevant Jurisdictions; and

(b) No Finance Party is or will be deemed to be resident, domiciled or carrying on business in its Relevant Jurisdictions by reason only of the execution, performance and/or enforcement of any Finance Document.

69

Table of Contents

20.29 Holding Company

Except as may arise under the Transaction Documents, before the Closing Date the Borrower has not traded or incurred any liabilities or commitments (actual or contingent, present or future) other than acting as a Holding Company of the Acquiror.

20.30 US government regulations

- (a) Neither it, nor any of its Subsidiaries, is an investment company, or an affiliated person of, or promoter or principal underwriter for, an investment company, as such terms are defined in the Investment Company Act of 1940, as amended. Neither the making of any Utilisation, nor the application of the proceeds or repayment thereof by any Obligor, nor the consummation of the other transactions contemplated hereby, will violate any provision of such Act or any rule, regulation or order of the Securities and Exchange Commission thereunder.
- (b) No Obligor is and, in the case of the Company, the Acquiror is not engaged in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulation U of the Board of Governors of the Federal Reserve System of the United States of America) as in effect from time to time, and no proceeds of any Loan will be used to purchase or carry any margin stock or to extend credit to others for the purpose of purchasing or carrying any margin stock.

20.31 Employee Benefit Plans

- (a) No ERISA Event has occurred or is reasonably expected to occur that has resulted in or is reasonably expected to result in a material liability of it or its Subsidiaries or any ERISA Affiliate.
- (b) Schedule B (*Actuarial Information*) to the most recent annual report (Form 5500 Series) for each Plan that has been filed with the Employee Benefits Security Administration of the United States of America, is complete and accurate and fairly presents the funding status of such Plan, and since the date of such Schedule B there has been no material adverse change in such funding status.
- (c) Neither it nor any ERISA Affiliate has incurred or is reasonably expected to incur any Withdrawal Liability to any Multiemployer Plan to the extent such incurrence would have or be reasonably likely to have a Material Adverse Effect.
- (d) Neither it nor any ERISA Affiliate has been notified in writing by the sponsor of a Multiemployer Plan that such Multiemployer Plan is in reorganisation or has been terminated, within the meaning of Title IV of ERISA, and no such Multiemployer Plan is reasonably expected to be in reorganisation or to be terminated, within the meaning of Title IV of ERISA.

70

Table of Contents

20.32 Solvency

If it is an Obligor organised under the laws of any state of the United States of America, it and each of its Subsidiaries which is organised under the laws of any state of the United States of America is US Solvent.

20.33 Times when representations made

- (a) All the representations and warranties in this Clause 20 are made by each Original Obligor on the date of this Agreement, the first Utilisation Date, the Closing Date and the date of payment of a claim under the Letter of Credit.
- (b) All the representations and warranties in this Clause 20 are deemed to be made by each Obligor on the Closing Date.
- (c) The Repeating Representations are deemed to be made by each Obligor on the date of each Utilisation Request, on each Utilisation Date and on the first day of each Interest Period (except that those contained in paragraphs (a) (e) of Clause 20.13 (*Original Financial Statements*) will cease to be so made once subsequent financial statements have been delivered under this Agreement).
- (d) All the representations and warranties in this Clause 20 except Clause 20.12 (*No misleading information*), Clause 20.23 (*Structure Memorandum*), Clause 20.26 (*Acquisition Documents, disclosures and Other Documents*) and Clause 20.29 (*Holding Companies*) are deemed to be made by each Additional Obligor on the day on which it becomes (or it is proposed that it becomes) an Additional Obligor.
- (e) Each representation or warranty deemed to be made after the date of this Agreement shall be deemed to be made by reference to the facts and circumstances existing at the date the representation or warranty is deemed to be made.

21. INFORMATION UNDERTAKINGS

The undertakings in this Clause 21 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

In this Clause 21:

Annual Financial Statements means the financial statements for a Financial Year delivered pursuant to paragraph (a) of Clause 21.1 (*Financial statements*).

Monthly Financial Information means the financial information delivered pursuant to paragraph (d) of Clause 21.1 (*Financial statements*).

Quarterly Financial Statements means consolidated quarterly financial statements for a Financial Quarter.

71

Table of Contents

21.1 Financial statements

The Company shall supply to the Agent in sufficient copies for all the Lenders:

- (a) as soon as they are available, but in any event within 90 days after the end of each of its Financial Years:
 - (i) its audited consolidated financial statements for that Financial Year;
 - (ii) the audited financial statements (consolidated if appropriate) of each Obligor and the Acquiror for that Financial Year; and
 - (iii) the audited financial statements of any other Subsidiary for that Financial Year if requested by the Agent;
- (b) as soon as they become available, but in any event within 90 days after the end of the first two Financial Quarters of each Financial Year, its consolidated financial statements for that financial half-year;
- (c) as soon as they are available, but in any event within 45 days after the end of each Financial Quarter of each of its Financial Years its consolidated financial statements for that Financial Quarter as filed with the Securities and Exchange Commission of the US together with, if the Agent so requests, the related working papers and, in relation to the Acquiror, a consolidated balance sheet for the Acquiror Group as at the end of such Financial Quarter;
- (d) as soon as they are available, but in any event within 30 days after the end of each month the financial information package, comprising such monthly management information as the Company and its Subsidiaries prepare together with such further information as the Agent may reasonably request.

21.2 Provision and contents of Compliance Certificate

- (a) The Company shall supply a Compliance Certificate to the Agent with each set of its audited consolidated Annual Financial Statements and each set of its consolidated Quarterly Financial Statements.
- (b) The Compliance Certificate shall, amongst other things, set out (in reasonable detail) computations as to compliance with Clause 22 (*Financial Covenants*).
- (c) Each Compliance Certificate shall be signed by two directors of the Company and, if required to be delivered with the consolidated Annual Financial Statements of the Company, shall be reported on by the Company s Auditors in the form required by the Majority Lenders (acting reasonably).

21.3 Requirements as to financial statements

(a) The Company shall procure that each set of Annual Financial Statements and Quarterly Financial Statements includes a balance sheet, profit and loss account and cashflow statement. In addition the Company shall procure that:

72

Table of Contents

- (i) each set of Annual Financial Statements shall be audited by the Auditors; and
- (ii) each set of Quarterly Financial Statements includes a cashflow forecast in respect of the Group relating to the three month period commencing at the end of the relevant Financial Quarter.
- (b) Each set of financial statements delivered pursuant to Clause 21.1 (Financial statements):
 - (i) shall be certified by a director of the relevant company as giving a true and fair view of (in the case of Annual Financial Statements for any Financial Year), or fairly representing (in other cases), its financial condition and operations as at the date as at which those financial statements were drawn up and, in the case of the Annual Financial Statements, shall be accompanied by any letter addressed to the management of the relevant company by the Auditors and accompanying those Annual Financial Statements;
 - (ii) in the case of consolidated financial statements of the Group, shall be accompanied by a statement by the directors of the Company comparing actual performance for the period to which the financial statements relate to:
 - (A) the projected performance for that period set out in the Budget; and
 - (B) the actual performance for the corresponding period in the preceding Financial Year of the Group; and
 - (iii) shall be prepared using the Accounting Principles, accounting practices and financial reference periods consistent with those applied in the preparation of the Base Case Model unless, in relation to any set of financial statements, the Company notifies the Agent that there has been a change in the Accounting Principles or the accounting practices and its Auditors (or, if appropriate, the Auditors of the Obligor or the Acquiror) deliver to the Agent:
 - (A) a description of any change necessary for those financial statements to reflect the Accounting Principles or accounting practices upon which the Base Case Model was prepared; and
 - (B) sufficient information, in form and substance as may be reasonably required by the Agent, to enable the Lenders to determine whether Clause 22 (*Financial Covenants*) has been complied with and to make an accurate comparison between the financial position indicated in those financial statements and the Base Case Model.

73

Table of Contents

Any reference in this Agreement to any financial statements shall be construed as a reference to those financial statements as adjusted to reflect the basis upon which the Base Case Model was prepared.

- (c) If the Agent wishes to discuss the financial position of any member of the Group with the Auditors, the Agent may notify the Company, stating the questions or issues which the Agent wishes to discuss with the Auditors. In this event, the Company must ensure that the Auditors are authorised (at the expense of the Company):
 - (i) to discuss the financial position of each member of the Group with the Agent on request from the Agent; and
 - (ii) to disclose to the Agent for the Finance Parties any information which the Agent may reasonably request.

21.4 **Budget**

- (a) The Company shall supply to the Agent in sufficient copies for all the Lenders, as soon as the same become available but in any event within 90 days before the start of each of its Financial Years, an annual Budget for that financial year.
- (b) The Company shall ensure that each Budget:
 - (i) is in a form and has a level of information reasonably acceptable to the Agent and shall include:
 - (A) a projected consolidated profit and loss account and cashflow statement of the Group for that Financial Year and for each quarter of that Financial Year;
 - (B) a projected consolidated balance sheet of the Group as at the end of each quarter of that Financial Year;
 - (C) Capital Expenditure, investments, acquisitions and disposals projected to be made by the Group during each quarter of that Financial Year;
 - (D) projected covenant calculations in respect of each financial covenant undertaking contained in Clause 22 (*Financial Covenants*) as at the end of or, as the case may be, in respect of the 12 Months ending at the end of each quarter of that Financial Year; and
 - (E) a management commentary on such budget (commenting on proposed activities of the Group during that Financial Year and the principal assumptions upon which the Budget was prepared);

/4

Table of Contents

- (ii) is prepared in accordance with the Accounting Principles and the accounting practices and financial reference periods applied to financial statements under Clause 21.1 (*Financial statements*); and
- (iii) has been approved by the board of directors of the Company.
- (c) If the Company updates or changes the Budget, it shall within five Business Days of the update or change being made deliver to the Agent, in sufficient copies for each of the Lenders, such updated or changed Budget together with a written explanation of the main changes in that Budget.

21.5 Management calls

Once in each Month and at any other time if requested to do so by the Agent if the Agent reasonably suspects that a Default is continuing or may have occurred or may occur, at least two directors (one of whom shall be the Group chief executive officer or the chief finance director shall participate on a conference call with the Finance Parties at a time and venue agreed with the Agent concerning the ongoing business and financial performance of the Group and such other matters related thereto as the Agent shall reasonably request (acting on the instructions of any Lender).

21.6 Year-end

- (a) The Company shall not change its Accounting Reference Date.
- (b) The Company shall procure that each quarterly accounting period and each Financial Quarter of each member of the Group ends on an accounting date.

21.7 Information: miscellaneous

The Company shall supply to the Agent (in sufficient copies for all the Lenders, if the Agent so requests):

- (a) at the same time as they are dispatched, copies of all documents dispatched by the Company to its shareholders generally (or any class of them) or dispatched by the Company, the Acquiror or any Obligors to its creditors generally (or any class of them) or otherwise filed with the Securities and Exchange Commission of the US;
- (b) promptly upon becoming aware of them, the details of any litigation, arbitration or administrative proceedings which are current, threatened or pending against any member of the Group, and which, if adversely determined, are reasonably likely to have a Material Adverse Effect or any material regulatory claim, investigation or proceeding involving any member of the Group;
- (c) promptly upon becoming aware of the relevant claim, the details of any claim which is current, threatened or pending against the Vendor or any other person in respect of the Acquisition Documents and details of any disposal which will require a prepayment under Clause 9.5 (*Mandatory Prepayment*);

75

Table of Contents

- (d) promptly, such information as the Security Agent may reasonably require about the Charged Property and compliance of the Obligors and the Acquiror with the terms of any Transaction Security Documents;
- (e) promptly following any change to the membership or structure of the Group or the ownership of any member of the Group, a copy of the Group Structure Chart amended to reflect such change;
- (f) promptly following the Closing Date, a calculation of the Total Net Worth immediately following the Closing Date certified by two directors of the Company and, promptly following the occurrence of the Permitted Reorganisation, a calculation of the Total Net Worth immediately following the occurrence of the Permitted Reorganisation certified by two directors of the Company; and
- (g) promptly on request, such further information regarding the financial condition, assets and operations of the Group and/or any member of the Group (including any requested amplification or explanation of any item in the financial statements, budgets or other material provided by any Obligor under this Agreement, any changes to management of the Group and an up to date copy of its Shareholders register (or equivalent in its jurisdiction of incorporation)) as any Finance Party through the Agent may reasonably request.

21.8 Notification of default

- (a) Each Obligor shall notify the Agent of any Default (and the steps, if any, being taken to remedy it) promptly upon becoming aware of its occurrence (unless that Obligor is aware that a notification has already been provided by another Obligor).
- (b) Promptly upon a request by the Agent, the Company shall supply to the Agent a certificate signed by two of its directors or senior officers on its behalf certifying that no Default is continuing (or if a Default is continuing, specifying the Default and the steps, if any, being taken to remedy it).

21.9 **Investigations**

- (a) If an Event of Default has occurred and is continuing or if the Majority Lenders believe in good faith and on reasonable grounds that any financial statements or calculations provided by any member of the Group are inaccurate or incomplete in any material respect, the Agent (acting on the instructions of the Majority Lenders) may (following consultation with the Company as to the scope of the investigation):
 - (i) instruct (or require the Company to instruct) the Auditors or another firm of accountants selected by the Agent to carry out an investigation into the affairs of the Group and/or the financial performance of the Group and/or the accounting and other reporting procedures and standards of the Group; and/or

76

Table of Contents

(ii) instigate such other investigations and commission such other reports (including, without limitation, legal and valuation reports) as the Agent (acting on the instructions of the Majority Lenders) shall reasonably require into the affairs of the Group,

in each case, to the extent that the Agent (acting reasonably) considers them to be necessary to establishing the accuracy of such financial statements. The reasonable expense of any such investigations or reports shall be borne by the Company, unless the Company provides all information necessary to show that such financial statements or calculations are not materially inaccurate or incomplete prior to the exercise by the Agent of the rights specified in paragraph (i) or (ii) above following a request from the Agent made on reasonable notice. The Company shall take all reasonable steps to ensure that each of its Subsidiaries co-operates fully with any person conducting an investigation or preparing a report referred to in this paragraph (a).

- (b) If the Agent wishes to discuss the financial position of any member of the Group with the Auditors, the Agent may notify the Company, stating the questions or issues which the Agent wishes to discuss with the Auditors. In this event, the Company must ensure that the Auditors are authorised (at the expense of the Company):
 - (i) to discuss the financial position of each member of the Group with the Agent on request from the Agent; and
 - (ii) to disclose to the Agent for distribution to the Finance Parties any information which the Agent may reasonably request.

21.10 Use of websites

- (a) The Company and the Borrower may satisfy its obligation under this Agreement to deliver any information in relation to those Lenders (the **Website Lenders**) who accept this method of communication by posting this information onto an electronic website designated by the Borrower and the Agent (the **Designated Website**) if:
 - (i) the Agent expressly agrees (after consultation with each of the Lenders) that it will accept communication of the information by this method;
 - (ii) each of the Company, Borrower and the Agent are aware of the address of and any relevant password specifications for the Designated Website; and
 - (iii) the information is in a format previously agreed between the Company and the Agent and capable of being posted on the Designated Website.

77

Table of Contents

If any Lender (a **Paper Form Lender**) does not agree to the delivery of information electronically then the Agent shall notify the Borrower accordingly and the Company and the Borrower shall supply the information to the Agent (in sufficient copies for each Paper Form Lender) in paper form. In any event the Company and the Borrower shall supply the Agent with at least one copy in paper form of any information required to be provided by it.

- (b) The Agent shall supply each Website Lender with the address of and any relevant password specifications for the Designated Website following designation of that website by the Borrower and the Agent.
- (c) The Company shall promptly upon becoming aware of its occurrence notify the Agent if:
 - (i) the Designated Website cannot be accessed due to technical failure;
 - (ii) the password specifications for the Designated Website change;
 - (iii) any new information which is required to be provided under this Agreement is posted onto the Designated Website;
 - (iv) any existing information which has been provided under this Agreement and posted onto the Designated Website is amended; or
 - (v) the Company becomes aware that the Designated Website or any information posted onto the Designated Website is or has been infected by any electronic virus or similar software.
 If the Company notifies the Agent under paragraph (c)(i) or paragraph (c)(v) above, all information to be provided by the Company and the Borrower under this Agreement after the date of that notice shall be supplied in paper form unless and until the Agent and each Website Lender is satisfied that the circumstances giving rise to the notification are no longer continuing.
- (d) Any Website Lender may request, through the Agent, one paper copy of any information required to be provided under this Agreement which is posted onto the Designated Website. The Company and the Borrower shall comply with any such request within ten Business Days.

21.11 Know your customer checks

- (a) If:
 - (i) the introduction of or any change in (or in the interpretation, administration or application of) any law or regulation made after the date of this Agreement;
 - (ii) any change in the status of an Obligor or the Acquiror after the date of this Agreement; or

Table of Contents 125

78

Table of Contents

- (iii) a proposed assignment or transfer by a Lender of any of its rights and obligations under this Agreement to a party that is not a Lender prior to such assignment or transfer, obliges the Agent or any Lender (or, in the case of paragraph (iii) above, any prospective new Lender) to comply with know your customer or similar identification procedures in circumstances where the necessary information is not already available to it, each Obligor shall promptly upon the request of the Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself or on behalf of any Lender) or any Lender (for itself or, in the case of the event described in paragraph (iii) above, on behalf of any prospective new Lender) in order for the Agent, such Lender or, in the case of the event described in paragraph (iii) above, any prospective new Lender to carry out and be satisfied it has complied with all necessary know your customer or other similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents. Without prejudice to the foregoing, each Obligor will provide to the Finance Parties its name, address and such other information as will allow each of the Finance Parties to identify such Obligor and its Subsidiaries to the extent necessary under the Financial Services Act 1996, the Patriot Act and any applicable Anti-Terrorism Laws.
- (b) Each Lender shall promptly upon the request of the Agent supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself) in order for the Agent to carry out and be satisfied it has complied with all necessary know your customer or other similar checks under all applicable laws and regulations pursuant to the transactions contemplated in the Finance Documents.
- (c) If the accession of such Additional Guarantor obliges the Agent or any Lender to comply with know your customer or similar identification procedures in circumstances where the necessary information is not already available to it, the Company shall promptly upon the request of the Agent or any Lender supply, or procure the supply of, such documentation and other evidence as is reasonably requested by the Agent (for itself or on behalf of any Lender) or any Lender (for itself or on behalf of any prospective new Lender) in order for the Agent or such Lender or any prospective new Lender to carry out and be satisfied it has complied with all necessary know your customer or other checks in relation to any relevant person pursuant to the accession of such Subsidiary to this Agreement as an Additional Guarantor.

22. FINANCIAL COVENANTS

22.1 **Financial definitions**: In this Agreement, unless the context requires otherwise, the following expressions shall have the following meanings, and each of the expressions shall refer to the position of the Group on a consolidated basis, unless the context otherwise requires and amounts shall not be double-counted:

Borrowings means Financial Indebtedness save for any Financial Indebtedness for or in respect of the items set out in paragraphs (j) or (l) (to the extent it relates to a

79

Table of Contents

guarantee of any of the items referred to in paragraph (j)) of the definition of Financial Indebtedness.

Capital Expenditure means any expenditure or obligation in respect of expenditure of the Group which should be treated as capital expenditure in the consolidated accounts of the Group in accordance with Accounting Principles and, to the extent any capital expenditure is financed by any finance lease, hire purchase or similar arrangement, including the capital element of any expenditure or obligation incurred in connection with such finance lease or similar arrangement.

Cashflow means, in respect of any Financial Quarter and without double counting any amount, EBITDA for that period:

- (a) less the aggregate of all Capital Expenditure paid by members of the Group during that period;
- (b) less any increase in the amount of Working Capital during that period;
- (c) less any extraordinary or exceptional items paid in cash (including any post-closing adjustments paid in cash by any member of the Group pursuant to the terms of the Acquisition Agreement) during that period;
- (d) less any Tax paid during that period by any member of the Group;
- (e) less any decrease in provisions and other non-cash credits, which are not Current Assets or Current Liabilities, taken into account in establishing EBITDA;
- (f) less (to the extent already included) the amount of any Disposal Proceeds and Acquisition Proceeds which are applied in prepayment of the Loans or provision of cash cover in accordance with this Agreement;
- (g) plus any decrease in Working Capital during that period;
- (h) plus any extraordinary or exceptional items received in cash during that period other than, but only to the extent not applied to mandatory prepayment of the Facility in accordance with this Agreement, any post closing adjustment payments pursuant to the Acquisition Agreement during the relevant period;
- (i) plus any Tax rebate received in cash during that period;
- (j) plus any increase in provisions and other non-cash debits and non-cash charges (which are not Current Assets or Current Liabilities) taken into account in establishing EBITDA; and
- (k) plus the amount of any dividends or other profit distributions (net of Tax) received in cash by any member of the Group during that period from companies which are not members of the Group and less any dividends or other distributions paid to minority shareholders of any member of the Group (other than the Company) or paid to any Restricted Person.

80

Table of Contents

Current Assets means the aggregate of all inventory, trade and other receivables in respect of operating items of each member of the Group including sundry debtors (excluding cash at bank, Cash Equivalent Investments, cash on hand and Interest Receivable due to mature within 12 months of the date of computation) and excluding:

- (a) receivables in relation to Tax:
- (b) extraordinary items, exceptional items and other non-operating items;
- (c) insurance claims; and
- (d) amounts due from the Vendor in connection with the Acquisition.

Current Liabilities means the aggregate of all liabilities of each member of the Group falling due within 12 months of the date of computation (including trade and other creditors, accruals, provisions and prepayments) falling due within such 12 month period excluding:

- (a) interest payable and liabilities for borrowings;
- (b) liabilities for Tax;
- (c) extraordinary items, exceptional items and non-operating items;
- (d) insurance claims;
- (e) liabilities for dividends declared but not paid by the Company; and
- (f) amounts due to the Vendor in connection with the Acquisition.

EBIT means, in respect of any Financial Quarter, the consolidated profit on ordinary activities of the Group for such period:

- (a) before any deduction for corporation Tax or other Taxes on income or gains;
- (b) before any deduction for Interest Payable and any addition for Interest Receivable;
- (c) before taking into account any items which are extraordinary or exceptional items;
- (d) after deducting the amount of profit of any member of the Group (other than the Company) which is attributable to any third party (not being a member of the Group) which is a shareholder or partner in such member of the Group;
- (e) after deducting the amount of any profit of any investment or entity (which is not itself a member of the Group) in which any member of the Group has an ownership interest to the extent that the amount of such profit included in the financial statements of the Group exceeds the amount (net of applicable withholding tax) received in cash by members of the Group through distributions by such investment or entity;

81

Table of Contents

- (f) after deducting any gain over book value arising in favour of a member of the Group on the disposal of any business or asset (not being any disposals made in the ordinary course of trading) during such period and any gain arising on any fair value revaluation of any business, asset or liability during such period;
- (g) after adding back any non-cash loss against book value incurred by a member of the Group on the disposal of any business or asset (not being any disposals made in the ordinary course of trading) during such period and any non-cash loss arising on any fair value revaluation of any business, asset or liability during such period;
- (h) before taking account of any realised and unrealised exchange gains and losses including those arising on the translation of Borrowings in a foreign currency; and
- (i) adding back (to the extent otherwise deducted) restructuring costs and transaction costs contemplated in the Structure Memorandum up to an amount not exceeding 100,000;
 in each case, to the extent added, deducted or taken into account, as the case may be, for the purposes of determining profits of the Group from ordinary activities before Taxes.

EBITDA means, in respect of any Relevant Period or Financial Quarter, EBIT for such period adding back depreciation of tangible assets and amortisation of goodwill or intangible assets during that period, to the extent deducted in arriving at EBIT.

Financial Month means each monthly period by reference to which management accounts of the Group Companies are prepared.

Financial Quarter means each of the quarterly periods ending on 31 March, 30 June, 30 September and 31 December in each Financial Year by reference to which the quarterly management accounts of members of the Group are prepared.

Financial Year means each period ending on its Accounting Reference Date in respect of which annual audited consolidated financial statements of the Group are required to be prepared.

Fixed Charge Cover means the ratio of Cashflow to Total Debt Service.

Interest Cover means the ratio of EBITDA to Net Interest Payable.

Interest Payable means, in respect of any Relevant Period or Financial Quarter, interest, commission, fees, discounts, repayment and prepayment premia and other finance payments accrued in respect of any Borrowings of any member of the Group for that period whether paid or payable or capitalised:

- (a) including the amount of the discount element of any Borrowings amortised during that period;
- (b) excluding any obligation owed by any member of the Group to any other member of the Group;

82

Table of Contents

whether or not paid.

- (c) including the interest element of payments on leases and other arrangements of the type referred to in paragraph (e) of the definition of Financial Indebtedness;
- (d) including any commission, discounts, repayment and prepayment premia and other finance payments payable by members of the Group under interest rate hedging arrangement in relation to that period;
- (e) excluding any accrued commission, discounts, repayment and prepayment premia and other finance payments payable to members of the Group under any interest rate hedging arrangement;
- (f) excluding amortisation of agreed restructuring costs or transaction costs, to the extent otherwise included; and
- (g) including the amount of cash dividends or other dividends paid by the Parent in that period.

 Interest Receivable means, in respect of any Relevant Period or Financial Quarter, the amount of accrued interest, commission, fees, discounts, repayment and prepayment premia and other finance payments due and payable in cash to members of the Group (other than by other members of the Group) during such period

Model means the financial model set out in Schedule 11 (*Model*).

Net Interest Payable means, in respect of any Relevant Period or Financial Quarter, the amount of Interest Payable less the amount of Interest Receivable during such period (excluding any such Interest Receivable which is capitalised or rolled-up or otherwise not currently payable in respect of the period and any Interest Receivable to the extent deemed irrecoverable).

Relevant Cash Equivalent Investments means Cash Equivalent Investments where securities or certificates would, if exchanged into cash, be freely and immediately available to be applied in repayment or prepayment of the Facility or in repayment or prepayment of Financial Indebtedness incurred by such member of the Group in the jurisdiction in which such Cash Equivalent Investments are held.

Relevant Period means a twelve month covenant testing period.

Total Debt means, at any time, the aggregate outstanding principal or capital amount of all Borrowings of the Group but:

- (a) excluding any Borrowings owed by one member of the Group to any other member of the Group; and
- (b) including in the case of finance leases and other items referred to in paragraph (d) of the definition of Financial Indebtedness only the capitalised value,

and so that guarantees shall be excluded to the extent the guaranteed Financial Indebtedness has been taken into account and no amount shall be included or excluded more than once.

83

Table of Contents

Total Debt Service means, in respect of any Relevant Period or Financial Quarter, the aggregate of:

- (a) Net Interest Payable for that period;
- (b) the aggregate of all scheduled repayments of the Facility falling due during that period (as adjusted as a result of any voluntary or mandatory prepayments); and
- (c) the aggregate of all scheduled repayments and all prepayments (whether voluntary or mandatory) of principal under the terms of any other Borrowings of any member of the Group (excluding any Borrowings between any member of the Group and any other member of the Group) falling due during that period:
 - (i) including, without limitation, the capital element of any payments falling due in respect of any Borrowings falling within paragraph (e) of the definition of Financial Indebtedness; and
 - (ii) excluding any repayment or prepayment of any overdraft, ancillary or revolving credit facility falling due during that period and capable of being simultaneously redrawn under the terms of the relevant facility,

and so that no amount shall be included more than once.

Total Leverage means the ratio of Total Debt to EBITDA.

Total Net Debt means, at any time, Total Debt but deducting the aggregate amount of Cash and Relevant Cash Equivalent Investments held by any member of the Group at the relevant time.

Total Net Leverage means the ratio of Total Net Debt to EBITDA.

Total Net Worth means at a particular time:

- (a) the amount (including any share premium) for the time being paid up or credited as paid up on the issued share capital of the Acquiror (other than shares which are expressed to be redeemable);
- (b) plus the aggregate amount standing to the credit or minus the amount standing to the debit of the consolidated capital and revenue reserves of the Acquiror Group; and
- (c) plus the aggregate amount of accrued profits or minus the amount of accrued losses for previous Financial Years which have been carried forward plus (to the extent not otherwise included) the amount standing to the credit (or minus any amount standing to the debit) of the profit and loss account of the Acquiror Group;

less (but without double counting) any amount included in the above which is attributable to:

84

Table of Contents

- (i) any dividends or other distributions declared, recommended or made by any member of the Acquiror Group (other than to another member of the Acquiror Group);
- (ii) interests of (if any) third party shareholders in any member of the Acquiror Group (other than the Borrower);
- (iii) any provisions made for Tax;
- (iv) the amount by which the net book value of any asset has been written up since 31 December, 2005 (or, in the case of a person becoming a member of the Acquiror Group after that date, the date on which that person became or becomes a member of the Acquiror Group) by way of revaluation or on its transfer from one member of the Acquiror Group to another; and
- (v) goodwill or other intangible assets;

but ignoring any variation in the credit or debit balance of the Acquiror Group consolidated profit and loss account since the date of the then latest audited consolidated balance sheet of the Acquiror Group except to the extent reflected in any later Acquiror Group consolidated profit and loss statement delivered to the Agent under Clause 21 (*Information Undertakings*).

Working Capital means on any date Current Assets less Current Liabilities on such date.

22.2 Financial condition

The Company shall ensure that:

(a) **Fixed Charge Cover**: Fixed Charge Cover in respect of any three month period specified in column 1 below shall not be less than the ratio specified in column 2 below in respect of that three month period.

Column 1 Three month period	Column 2 Ratio
Period expiring:	
30 September 2006	2.4:1
31 December 2006	0.0:1
31 March 2007	0.0:1
30 June 2007	1.4:1
30 September 2007	1.2:1
31 December 2007	4.8:1
31 March 2008	3.4:1
30 June 2008	3.6:1
30 September 2008	5.7:1
8.	5

Table of Contents

(b) **Interest Cover**: Interest Cover in respect of any Relevant Period specified in column 1 below shall be or shall exceed the ratio specified in column 2 below in respect of that Relevant Period.

Column 1	Column 2
Relevant Period	Ratio
Period expiring:	
30 September 2006	3.0:1
31 December 2006	1.6:1
31 March 2007	1.8:1
30 June 2007	2.0:1
30 September 2007	1.9:1
31 December 2007	3.8:1
31 March 2008	4.9:1
30 June 2008	5.8:1
30 September 2008	7.5:1

(c) **Total Leverage**: Total Leverage in respect of any Relevant Period specified in column 1 below (being Total Debt as at the last day of such Relevant Period to EBITDA for that Relevant Period) shall not exceed the ratio specified in column 2 below in respect of that Relevant Period.

Column 1		Column 2
Relevant Period		Ratio
Period expiring:		
30 September 2006		6.2:1
31 December 2006		9.9:1
31 March 2007		6.5:1
30 June 2007		6.2:1
30 September 2007		6.9:1
31 December 2007		4.0:1
31 March 2008		3.3:1
30 June 2008		2.8:1
30 September 2008		2.2:1
	86	

Table of Contents

- (d) **Minimum Net Worth**: The Total Net Worth shall not on 31st December 2006 and any subsequent Quarter Date be less than 95% of the lower of:
 - (i) 27 million; and
 - (ii) the aggregate of 25 million and the Final Amount (as defined in the Warrant Document) provided that if, when the Minimum Net Worth for any test date falls to be calculated, such Final Amount has not been determined, paragraph (i) shall be applicable.
- (e) **Total Net Leverage**: Total Net Leverage in respect of any Relevant Period specified in column 1 below (being Total Net Debt as at the last day of such Relevant Period to EBITDA for that Relevant Period) shall not exceed the ratio specified in column 2 below in respect of that Relevant Period.

Column 1	Column 2
Relevant Period	Ratio
Period expiring	
30 September 2006	1.0:1
31 December 2006	1.8:1
31 March 2007	1.8:1
30 June 2007	1.6:1
30 September 2007	1.7:1
31 December 2007	0.7:1
31 March 2008	0.4:1
30 June 2008	0.4:1
30 September 2008	0.1:1

(f) Capital Expenditure:

- (i) The aggregate Capital Expenditure of the Group in respect of:
 - (A) the period beginning on the date of this Agreement and ending on the expiry of the first Financial Year of the Company specified in column 1 below; and
 - (B) each other Financial Year of the Company specified in column 1 below, shall not exceed the amount set out in column 2 below opposite that Financial Year.

87

Table of Contents

 Column 1
 Column 2

 Maximum
 Maximum

 Financial Year Ending
 Expenditure

 31st December 2006
 1,328,750

 31st December 2007
 891,250

 31st December 2008
 980,000

If in any Financial Year (the **Original Financial Year**) the amount of the Capital Expenditure is less than the maximum amount permitted for that Original Financial Year (the difference being referred to below as the **Unused Amount**), then the maximum expenditure amount set out in column 2 above for the immediately following Financial Year (the **Carry Forward Year**) shall be increased by an amount equal to the lowest of (1) the Unused Amount and (2) 50% of the maximum amount permitted for the Original Financial Year and (3) the amount which could have been utilised by way of additional Capital Expenditure in the Original Financial Year without causing a breach of the Cash Cover test during that Original Financial Year.

In any Carry Forward Year, the original amount specified in column 2 above shall be treated as having been incurred prior to any Unused Amount carried forward into such Carry Forward Year.

(ii) The aggregate amount of the capital value of all assets of the Group in respect of which there is any liability under a finance lease or capital lease shall not at any time exceed 500,000.

22.3 Financial testing

The financial covenants set out in Clause 22.2 (Financial condition) shall be calculated in accordance with the Accounting Principles and tested by reference to each of the financial statements and/or each Compliance Certificate delivered pursuant to Clause 21.2 (*Provision and contents of Compliance Certificate*) provided that:

- (a) in the case of Interest Cover, for the first three test dates falling after the date of this Agreement, EBITDA and Net Interest Payable shall be calculated on the basis of EBITDA and Net Interest Payable for the period from 1st July 2006 until such test date; and
- (b) in the case of Total Leverage and Total Net Leverage, for the first three test dates falling after the date of this Agreement, EBITDA shall be calculated on the basis of EBITDA for the period from 1st July 2006 to such test date, annualised.

23. GENERAL UNDERTAKINGS

The undertakings in this Clause 23 remain in force from the date of this Agreement for so long as any amount is outstanding under the Finance Documents or any Commitment is in force.

88

Table of Contents

Authorisations and compliance with laws

23.1 Authorisations

Each Obligor shall (and the Company shall procure that the Acquiror shall) promptly:

- (a) obtain, comply with and do all that is necessary to maintain in full force and effect; and
- (b) supply certified copies to the Agent of,
- any Authorisation required under any law or regulation of a Relevant Jurisdiction to:
 - (i) enable it to perform its obligations under the Finance Documents and the Acquisition Documents;
 - (ii) ensure the legality, validity, enforceability or admissibility in evidence of any Finance Document or Acquisition Document; and
 - (iii) carry on its business where failure to do so has or is reasonably likely to have a Material Adverse Effect.

23.2 Compliance with laws

- (a) Each Obligor shall (and the Company shall ensure that each member of the Group will) comply in all respects with all laws and regulations to which it may be subject where failure so to comply has or is reasonably likely to have a Material Adverse Effect.
- (b) Each Obligor shall (and the Company shall ensure that each member of the Group will) comply with all Anti-Terrorism Laws to which it may be subject.
- (c) Each Obligor shall (and the Company shall ensure that each member of the Group will) comply with the applicable laws or regulations relating to Margin Stock.
- (d) Each Obligor shall not (and the Company shall ensure that no member of the Group will) by act or omission become subject to any categories, laws or regulations described in Clause 20.30 (*US government regulations*).

23.3 Environmental compliance

Each Obligor shall (and the Company shall ensure that each member of the Group will):

- (a) comply with all Environmental Law to which it may be subject;
- (b) obtain, maintain and ensure compliance with all requisite Environmental Permits;
- (c) implement procedures to monitor compliance with and to prevent liability under any Environmental Law to which it may be subject,

89

Table of Contents

where failure to do so has or is reasonably likely to have a Material Adverse Effect.

23.4 Environmental claims

Each Obligor shall (through the Company), promptly upon becoming aware of the same, inform the Agent in writing of:

- (a) any Environmental Claim against any member of the Group which is current, pending or threatened; and
- (b) any facts or circumstances which are reasonably likely to result in any Environmental Claim being commenced or threatened against any member of the Group,

where the claim, if determined against that member of the Group, has or is reasonably likely to have a Material Adverse Effect.

23.5 Taxation

- (a) Each Obligor shall (and the Company shall ensure that each member of the Group will) pay and discharge all Taxes imposed upon it or its assets within the time period allowed without incurring penalties unless and only to the extent that:
 - (i) such payment is being contested in good faith;
 - (ii) adequate reserves are being maintained for those Taxes and the costs required to contest them which have been disclosed in its latest financial statements delivered to the Agent under Clause 21.1 (*Financial statements*); and
 - (iii) such payment can be lawfully withheld and failure to pay those Taxes does not have or is not reasonably likely to have a Material Adverse Effect or to result in the creation of any Security ranking in priority to the Transaction Security.
- (b) No member of the Group may change its residence for Tax purposes.

Restrictions on business focus

23.6 Merger

No Obligor shall (and the Company shall ensure that no other member of the Group will) enter into any amalgamation, demerger, merger, consolidation or corporate reconstruction (other than the Permitted Reorganisation).

23.7 Change of business

The Company shall procure that no substantial change is made to the general nature of the business of the Company, the Acquiror, the Obligors or the Group taken as a whole from that carried on at the date of this Agreement.

90

Table of Contents

23.8 Acquisitions

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no other member of the Group will):
 - (i) acquire a company or any shares, other equity-related investments or securities or a business or undertaking (or, in each case, any interest in any of them); or
 - (ii) incorporate a company.
- (b) Paragraph (a) above does not apply to an acquisition of a company, of shares, securities or a business or undertaking (or, in each case, any interest in any of them) or the incorporation of a company which is:
 - (i) a Permitted Acquisition; or
 - (ii) a Permitted Transaction.

23.9 **Joint ventures**

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no member of the Group will):
 - (i) enter into, invest in or acquire (or agree to acquire) any shares, stocks, securities or other interest in any Joint Venture; or
 - (ii) transfer any assets or lend to or guarantee or give an indemnity for or give Security for the obligations of a Joint Venture or maintain the solvency of or provide working capital to any Joint Venture (or agree to do any of the foregoing).
- (b) Paragraph (a) above does not apply to any acquisition of (or agreement to acquire) any interest in a Joint Venture or transfer of assets (or agreement to transfer assets) to a Joint Venture or loan made to or guarantee given in respect of the obligations of a Joint Venture if such transaction is a Permitted Acquisition, a Permitted Disposal, a Permitted Loan or a Permitted Joint Venture.

23.10 Holding Companies

The Borrower shall not and the Company shall procure that the Acquiror shall not trade, carry on any business, own any assets or incur any liabilities except for:

- (a) the provision of administrative services (excluding treasury services) to its Subsidiaries of a type customarily provided by a holding company to its Subsidiaries;
- (b) ownership of shares in its Subsidiaries (and in the case of the Acquiror, its only direct Subsidiary shall be the Target), intra-Group debit balances, intra-Group credit balances and other credit balances in bank accounts, cash and Cash Equivalent Investments but only if those shares, credit balances, cash

91

Table of Contents

and Cash Equivalent Investments are subject to the Transaction Security provided that the Company shall procure that the Acquiror shall at all times after the date of this Agreement maintain a credit balance in the Dividend Account in an amount not less than 500,000;

(c) any liabilities under the Transaction Documents to which it is a party and professional fees and administration costs in the ordinary course of business as a holding company.

Restrictions on dealing with assets and Security

23.11 Preservation of assets

- (a) Each Obligor shall (and the Company shall ensure that each member of the Group will) maintain in good working order and condition (ordinary wear and tear excepted) all of its assets necessary or desirable in the conduct of its business.
- (b) The Company shall procure that the Target does not undertake or incur any business, activities, acquisitions or liabilities which will materially reduce its cash or Cash Equivalent Investments except for the Permitted Reorganisation and the payment of dividends to the Acquiror for the purpose of making payments to the Lenders under this Agreement.
- (c) The Company shall ensure that each Regulated Entity maintains capital and liquidity sufficient at all times to comply with its Authorisations and the requirements of its regulators (including, without limitation, in relation to the Permitted Reorganisation).

23.12 Pari passu ranking

Each Obligor shall (and the Company shall procure that the Acquiror shall) ensure that at all times any claims of a Finance Party against it under the Finance Documents rank at least pari passu with the claims of all its other unsecured and unsubordinated creditors except those creditors whose claims are mandatorily preferred by laws of general application to companies in the jurisdiction of incorporation or organisation of such Obligor or the Acquiror as the case may be.

23.13 Acquisition Documents

- (a) The Company shall procure that the Acquiror shall promptly pay all amounts payable to the Vendor under the Acquisition Documents as and when they become due (except to the extent that any such amounts are being contested in good faith by a member of the Group and where adequate reserves are set aside for any such payment).
- (b) The Company shall procure that the Acquiror and each relevant member of the Group shall, take all reasonable and practical steps to preserve and enforce its rights (or the rights of any other member of the Group) and pursue any claims and remedies arising under any Acquisition Documents.

92

Table of Contents

- (c) The Company shall:
 - (i) notify the Agent promptly of any material claim made by a Group Company under an Acquisition Document;
 - (ii) promptly on request provide the Agent with reasonable details of that claim and its progress; and
 - (iii) notify the Agent promptly upon that claim being resolved.
- (d) The Company shall keep the Agent informed as to the status and progress of the Acquisition.
- (e) Each Obligor shall (and the Company shall ensure that each member of the Group will) comply in all respects with all laws and regulations to which it may be subject under the Acquisition.
- (f) The Company shall keep the Agent informed and consult with it as to:
 - (i) the terms and conditions of any assurance or undertaking proposed to be given by or on behalf of any member of the Group to any person for the purpose of obtaining any regulatory, competition and/or anti-trust Authorisation or clearance necessary or desirable in connection with the Acquisition; and
 - (ii) any terms or conditions proposed in connection with any regulatory, competition and/or anti-trust Authorisation or clearance necessary or desirable in connection with the Acquisition.

23.14 Negative pledge

In this Clause 23.14, **Quasi-Security** means a transaction described in paragraph (b) below.

Except as permitted under paragraph (c) below:

- (a) No Obligor shall (and the Company shall ensure that no other member of the Group will) create or permit to subsist any Security over any of its assets.
- (b) No Obligor shall (and the Company shall ensure that no other member of the Group will):
 - (i) sell, transfer or otherwise dispose of any of its assets on terms whereby they are or may be leased to or re-acquired by an Obligor or any other member of the Group;
 - (ii) sell, transfer or otherwise dispose of any of its receivables on recourse terms;
 - (iii) enter into any arrangement under which money or the benefit of a bank or other account may be applied, set-off or made subject to a combination of accounts; or

93

Table of Contents

- (iv) enter into any other preferential arrangement having a similar effect, in circumstances where the arrangement or transaction is entered into primarily as a method of raising Financial Indebtedness or of financing the acquisition of an asset.
- (c) Paragraphs (a) and (b) above do not apply to any Security or (as the case may be) Quasi-Security, which is:
 - (i) Permitted Security; or
 - (ii) a Permitted Transaction.

23.15 Disposals

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no member of the Group will) enter into a single transaction or a series of transactions (whether related or not) and whether voluntary or involuntary to sell, lease, transfer or otherwise dispose of any asset.
- (b) Paragraph (a) above does not apply to any sale, lease, transfer or other disposal which is:
 - (i) a Permitted Disposal; or
 - (ii) a Permitted Transaction.

23.16 Arm s length basis

- (a) Except as permitted by paragraph (b) below, no Obligor shall (and the Company shall ensure no member of the Group will) enter into any transaction with:
 - (i) any person except on arm s length terms and for full market value; and
 - (ii) any Restricted Person, which has a value in excess of 500,000 or which results in the aggregate value of all such transactions exceeding 2 million.
- (b) The following transactions shall not be a breach of this Clause 23.16:
 - (i) intra-Group loans permitted under Clause 23.17 (Loans or credit);
 - (ii) fees, costs and expenses payable under the Transaction Documents in the amounts set out in the Transaction Documents delivered to the Agent under Clause 4.1 (*Initial conditions precedent*) or agreed by the Agent; and
 - (iii) any Permitted Transactions.

Restrictions on movement of cash cash out

94

Table of Contents

23.17 Loans or credit

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no member of the Group will) be a creditor in respect of any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to:
 - (i) a Permitted Loan; or
 - (ii) a Permitted Transaction.

23.18 No Guarantees or indemnities

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no member of the Group will) incur or allow to remain outstanding any guarantee in respect of any obligation of any person.
- (b) Paragraph (a) does not apply to a guarantee which is:
 - (i) a Permitted Guarantee; or
 - (ii) a Permitted Transaction.

23.19 Dividends and share redemption

- (a) Except as permitted under paragraph (b) below, the Company shall not (and will ensure that no other member of the Group will):
 - (i) declare, make or pay any dividend, charge, fee or other distribution (or interest on any unpaid dividend, charge, fee or other distribution) (whether in cash or in kind) on or in respect of its share capital (or any class of its share capital);
 - (ii) repay or distribute any dividend or share premium reserve;
 - (iii) pay or allow any member of the Group to pay any management, advisory or other fee to or to the order of any of the shareholders of the Company; or
 - (iv) redeem, repurchase, defease, retire or repay any of its share capital or resolve to do so.
- (b) Paragraph (a) above does not apply to a Permitted Transaction (other than one referred to in paragraph (c) of the definition of that term).

23.20 Shareholder Indebtedness

The Company shall ensure that at all times all Shareholder Indebtedness is subordinated to all amounts payable under the Finance Documents, on a basis approved by the Agent, and shall not (and will ensure that no other member of the Group will):

95

Table of Contents

- (a) repay or prepay any principal amount (or capitalised interest) outstanding under any Shareholder Indebtedness:
- (b) pay any interest, fee or charge accrued or due under any Shareholder Indebtedness;
- (c) purchase, redeem, defease or discharge any Shareholder Indebtedness; or
- (d) take any action which would prejudice the subordination of any Shareholder Indebtedness.

Restrictions on movement of cash cash in

23.21 Financial Indebtedness

- (a) Except as permitted under paragraph (b) below, no Obligor shall (and the Company shall ensure that no member of the Group will) incur or allow to remain outstanding any Financial Indebtedness.
- (b) Paragraph (a) above does not apply to Financial Indebtedness which is:
 - (i) Permitted Financial Indebtedness; or
 - (ii) a Permitted Transaction.

23.22 Share capital

No Obligor shall (and the Company shall ensure no member of the Group will) issue any shares except pursuant to:

- (a) a Permitted Share Issue; or
- (b) a Permitted Transaction.

Miscellaneous

23.23 Insurance

- (a) Each Obligor shall (and the Company shall ensure that each member of the Group will) maintain at its own expense fidelity and other insurances on and in relation to its business and assets against such risks, in such amounts and to such extent as would be prudent for companies owning, possessing or leasing similar assets and carrying on a business of a substantially similar size and nature and operating in the same locations as those of such members of the Group or which are required by law or regulation. Such insurances must:
 - (i) provide coverage against business interruption and loss of profits, fidelity and other matters customary for Regulated Entities, fire, terrorism (to the extent that such cover is available on commercially viable terms), public liability, professional indemnity, pollution and third-party property damage liability but subject to any exclusions typical for insurances of this type; and

96

Table of Contents

- (ii) be in compliance with other insurance obligations binding on such members of the Group such as under any leases of material property to which such members of the Group are party where failure to do so could lead to forfeiture of such lease.
- (b) All insurances must be with reputable independent insurance companies or underwriters.
- (c) Each Obligor will, and will procure that each of its Subsidiaries will:
 - (i) supply to the Agent promptly following its request copies of each policy or certificate for insurance required to be maintained in accordance with paragraph (a) above, together with copies of the current premium receipts relating thereto, and use all reasonable endeavours to procure that the insurer undertakes to the Agent to notify the Agent within 30 days (or such other period as may be agreed between the Agent and the insurer) should any sum payable by a member of the Group under such insurance not be paid or give no less than 30 days notice (or such other period as may be agreed between the Agent and the insurer) should the insurer intend to cancel, lapse or avoid such policy;
 - (ii) promptly notify the Agent in writing of any material change to its insurance coverage from time to time; and
 - (iii) promptly notify the Agent in writing of any claim or notification made to its insurers under any of its insurance policies which is for, or is reasonably likely to result in, a claim under such policy for an amount in excess of 500,000 (or its equivalent in other currencies).
- (d) If any Obligor (or any other member of the Group) fails to purchase or maintain any insurance required by this Clause 23.23 (*Insurance*), the Agent may purchase such insurance as may be necessary to remedy any such failure and each Obligor shall (and the Company shall ensure that each member of the Group will) indemnify the Agent on demand against any costs or expenses properly incurred by it in purchasing any such insurance.

23.24 Pensions

- (a) The Company shall ensure that all pension schemes operated by or maintained for the benefit of members of the Group and/or any of their employees are fully funded to the extent required by law based on reasonable actuarial assumptions applicable in the jurisdiction in which the relevant pension scheme is maintained.
- (b) The Company shall ensure that no member of the Group is or has been at any time an employer (for the purposes of Sections 38 to 51 of the Pensions Act 2004) of an occupational pension scheme which is not a money purchase scheme (both terms as defined in the Pension Schemes Act 1993) or connected with or an associate of (as those terms are under in Sections 39 or 43 of the Pensions Act 2004) such an employer.

9/

Table of Contents

- (c) The Company shall deliver to the Agent at such times as those reports are prepared in order to comply with the then current statutory or auditing requirements (as applicable either to the trustees of any relevant schemes or to the Company), actuarial reports in relation to all pension schemes mentioned in (a) above.
- (d) The Company shall promptly notify the Agent of any material change in the rate of contributions to any pension schemes mentioned in (a) above paid or recommended to be paid (whether by the scheme actuary or otherwise) or required (by law or otherwise).

23.25 Access

If a Default is continuing or the Agent reasonably suspects a Default is continuing or may occur, each Obligor shall, and the Company shall ensure that each member of the Group will, (not more than once in every Financial Year unless the Agent reasonably suspects a Default is continuing or may occur) permit the Agent and/or the Security Agent and/or accountants or other professional advisers and contractors of the Agent or Security Agent free access at all reasonable times and on reasonable notice at the risk and cost of the Obligor or Company to (a) the premises, assets, books, accounts and records of each member of the Group and (b) meet and discuss matters with management of the Company.

23.26 Intellectual Property

Each Obligor shall and the Company shall procure that each Group member will:

- (a) preserve and maintain the subsistence and validity of the Intellectual Property necessary for the business of the relevant Group member;
- (b) use reasonable endeavours to prevent any infringement in any material respect of the Intellectual Property;
- (c) make registrations and pay all registration fees and taxes necessary to maintain the Intellectual Property in full force and effect and record its interest in that Intellectual Property;
- (d) not use or permit the Intellectual Property to be used in a way or take any step or omit to take any step in respect of that Intellectual Property which may materially and adversely affect the existence or value of the Intellectual Property or imperil the right of any member of the Group to use such property; and
- (e) not discontinue the use of the Intellectual Property.

23.27 Amendments

No Obligor shall (and the Company shall ensure that no member of the Group will) amend, vary, novate, supplement, supersede, waive or terminate any term of a Transaction Document (save for a non-material term of an Acquisition Document) or any other document delivered to the Agent pursuant to Clauses 4.1 (*Initial conditions precedent*) (including without limitation the constitutional documents of the

Table of Contents

Borrower, the Acquiror, the Target and each of the other Obligors) or Clause 25 (*Changes to the Lenders and the Obligors*) or enter into any agreement with any shareholders of the Company or any of their Affiliates which is not a member of the Group except in writing:

- (a) in accordance with the provisions of Clause 35 (Amendments and Waivers); or
- (b) with the prior written consent of the Lenders; or
- (c) after the Closing Date in the case of a document other than a Finance Document, in a way which could not be reasonably expected materially and adversely to affect the interests of the Lenders.
- (d) The Company shall promptly supply to the Agent a copy of any document relating to any of the matters referred to in paragraphs (a) to (c) above.

23.28 Financial assistance

Each Obligor shall (and the Company shall procure each member of the Group will) comply in all respects with Sections 151 to 158 of the United Kingdom Companies Act 1985 and any equivalent legislation in other jurisdictions, in each case as applicable, including in relation to the execution of the Transaction Security Documents and payment of amounts due under this Agreement.

23.29 Group bank accounts

If the Majority Lenders so require by notice in writing, the Company shall ensure that within 30 days of the date of any such notice all bank accounts of the Obligors and such other members of the Group as the Majority Lenders may require shall be opened and maintained with a Finance Party or an Affiliate of a Finance Party and, in the case of an Obligor, are subject to valid Security under the Transaction Security Documents.

23.30 Treasury Transactions

No Obligor shall (and the Company will procure that no members of the Group will) enter into any Treasury Transaction, other than:

- (a) spot and forward delivery foreign exchange contracts entered into in the ordinary course of business and not for speculative purposes; and
- (b) any Treasury Transaction entered into for the hedging of actual or projected real exposures arising in the ordinary course of trading activities of a member of the Group for a period of not more than six months and not for speculative purposes.

23.31 Guarantors

(a) The Company shall ensure that at all times after the Acquisition Signing Date, the aggregate assets, the aggregate net assets and aggregate turnover (after Tax) of the Guarantors and the Acquiror (in each case calculated on an

99

unconsolidated basis and excluding all intra-group items) represents not less than 80 per cent. of consolidated turnover (after Tax) of the Group.

(b) The Company need only perform its obligations under paragraph (a) above if it is not unlawful for the relevant person to become a Guarantor and that person becoming a Guarantor would not result in personal liability for that person s directors or other management. Each Obligor must use, and must procure that the relevant person uses, all reasonable endeavours lawfully available to avoid any such unlawfulness or personal liability. This includes agreeing to a limit on the amount guaranteed. The Agent may (but shall not be obliged to) agree to such a limit if, in its opinion, to do so would avoid the relevant unlawfulness or personal liability.

23.32 Further assurance

- (a) Each Obligor shall (and the Company shall procure that each member of the Group will) promptly do all such acts or execute all such documents (including assignments, transfers, mortgages, charges, notices and instructions) as the Security Agent may reasonably specify (and in such form as the Security Agent may reasonably require in favour of the Security Agent or its nominee(s)):
 - (i) to perfect the Security created or intended to be created under or evidenced by the Transaction Security Documents (which may include the execution of a mortgage, charge, assignment or other Security over all or any of the assets which are, or are intended to be, the subject of the Transaction Security) or for the exercise of any rights, powers and remedies of the Security Agent or the Finance Parties provided by or pursuant to the Finance Documents or by law;
 - (ii) to confer on the Security Agent or confer on the Finance Parties Security over any property and assets of that Obligor or the Acquiror located in any jurisdiction equivalent or similar to the Security intended to be conferred by or pursuant to the Transaction Security Documents; and/or
 - (iii) to facilitate the realisation of the assets which are, or are intended to be, the subject of the Transaction Security.
- (b) Each Obligor shall (and the Company shall procure that each member of the Group shall) take all such action as is available to it (including making all filings and registrations) as may be necessary for the purpose of the creation, perfection, protection or maintenance of any Security conferred or intended to be conferred on the Security Agent or the Finance Parties by or pursuant to the Finance Documents.

23.33 Permitted Reorganisation and Rooftop Sale

(a) The Company shall ensure that the Permitted Reorganisation and the Rooftop Sale are completed as soon as practicable and in any event in the case of the Permitted Reorganisation prior to 31st March 2007 and in the case of the

100

Table of Contents

Rooftop Sale prior to 31st December, 2008. The Permitted Reorganisation shall be completed in accordance with the steps set out at Transaction 1, Transaction 2 and Transaction 3 (as each are defined in the BT Treasury Letter) in the Structure Memorandum which Structure Memorandum may not be amended without the prior written consent of the Agent.

- (b) Prior to the completion of the Permitted Reorganisation and of the Rooftop Sale:
 - (i) the Company shall keep the Agent informed as to the progress of such matters at such intervals as the Agent may specify (or upon request from the Agent); and
 - (ii) the Company shall take all such steps as may be necessary or appropriate to preserve or enhance the value of the Crown Northcorp Limited and its relevant subsidiaries and investments to be sold to the Target in accordance with the Permitted Reorganisations and Rooftop Holding Limited and to minimise the tax and other liabilities arising in connection therewith.
- (c) The Company shall promptly obtain all such Authorisations as may be required for the purposes of the Permitted Reorganisation and the Rooftop Sale.

23.34 ERISA reporting requirements

Each Obligor shall (and the Company shall ensure that each relevant member of the Group will):

- (a) **ERISA Events and ERISA Reports**: (i) promptly and in any event within 10 days after such Obligor or any ERISA Affiliate knows or has reason to know that any ERISA Event has occurred, deliver to the Agent a statement of the finance director of the Company describing such ERISA Event and the action, if any, that such Obligor or such ERISA Affiliate has taken and proposes to take with respect thereto, and (ii) on the date any records, documents or other information must be furnished to the PBGC with respect to any Plan pursuant to section 4010 of ERISA, deliver to the Agent a copy of such records, documents and information;
- (b) **Plan Terminations**: promptly and in any event within five Business Days after receipt thereof by any Obligor or any ERISA Affiliate, deliver to the Agent copies of each written notice from the PBGC stating its intention to terminate any Plan or to have a trustee appointed to administer any Plan;
- (c) **Plan Annual Reports**: promptly upon the written request of the Agent, deliver to the Agent copies of each Schedule B (*Actuarial Information*) to the annual report (Form 5500 Series) most recently filed by it with the Employee Benefits Security Administration of the United States with respect to each Plan; and

- (d) **Multiemployer Plan Notices**: promptly and in any event within five Business Days after receipt thereof by it or any ERISA Affiliate from the sponsor of a Multiemployer Plan, deliver to the Agent copies of each written notice concerning:
 - (i) the imposition of Withdrawal Liability by any such Multiemployer Plan;
 - (ii) the reorganisation or termination, within the meaning of Title IV of ERISA, of any such Multiemployer Plan; or
 - (iii) the amount of liability incurred, or that may be incurred, by such Obligor or any ERISA Affiliate in connection with any event described in paragraph (i) or (ii) above.

23.35 Interest Reserve Account

The Borrower shall promptly convert the US\$2,500,000 credited to the initial Interest Reserve Account into Euro once it is no longer in the initial Interest Reserve Account, upon the request of the Agent. No amounts shall be withdrawn from the Interest Reserve Account and the amounts credited to such account shall not be dealt with in any way except to pay the arrangement fee described in Clause 13.1 (*Arrangement Fee*) and to pay interest under Clause 10.2 (*Payment of Interest*).

23.36 Conditions Subsequent in relation to Warrants

The Company and the Agent shall use their reasonable endeavours in good faith to agree forms of the Warrant Agreement and each of the other Warrant Documents and the constitutional documents of the Acquiror. As soon as possible and in any event within five Business Days after the forms of such documentation are agreed, the Company shall procure that:

- (a) the Warrant Agreement and each of the other Warrant Documents, each in the agreed form, are duly executed and delivered by all the parties thereto and all the Warrants provided for thereunder are duly and validly issued to the Arranger or an affiliate of the Arranger, nominated by the Arranger;
- (b) the constitutional documents of the Acquiror are in the agreed form; and
- (c) the Agent receives in such form and with such substance as it shall consider satisfactory such evidence of corporate, shareholder and other authorisations and approvals and filings and such legal opinions as it shall consider appropriate in the context of the execution and delivery of the Warrant Documents, any amendments to the constitutional documents of the Acquiror and the issue of the Warrants.

23.37 Conditions Subsequent in relation to German and Dutch Transaction Security

(a) The Borrower and the Acquiror shall execute and deliver to the Agent, and the Company shall procure that the Borrower and the Acquiror shall execute and deliver to the Agent, all of the documents and evidence listed in Part III of Schedule 2 (*Conditions Precedent*), as soon as possible and in any event no

102

Table of Contents

later than 10 Business Days after the date of this Agreement (except in the case of the execution and delivery of the Assignment Agreement referred to in paragraph 8 of Part III of Schedule 2 (*Conditions Precedent*) which shall be as soon as possible and in any event no later than 10 Business Days after the obtaining of the Vendor s consent to such Assignment Agreement), each in a form and substance satisfactory to the Agent.

(b) The Borrower undertakes that it will make and give, and the Company undertakes that it will procure that the Borrower and the Acquiror make and give (i) all filings, registrations and notices that are necessary to create and/or perfect any Transaction Security to be created in accordance with paragraph (a) above and (ii) all reasonable endeavours to obtain the consent of the Vendor to the Assignment Agreement referred to above.

23.38 Conditions Subsequent in relation to Reliance Letters

The Company shall use its best endeavours to procure that there is delivered to the Agent as soon as practicable after the date of this Agreement:

- (a) reliance letters, in form and substance satisfactory to the Agent, in relation to the PWC valuation report referred to in paragraph 5(i) of Part I of Schedule 2, the Legal Due Diligence Report and the Structure Memorandum:
- (b) a copy of the final PWC valuation report referred to in paragraph 5(i) of Part I of Schedule 2; and
- (c) a revised opinion of Dechert LLP covering enforceability of judgements and choice of law on a basis satisfactory to the Agent.

23.39 Dividend Account

The Company shall procure that the Acquiror shall promptly pay all:

- (a) dividends, charges, fees or other distributions; and
- (b) interest on any unpaid dividend, charge, fee or other distribution, received by the Acquiror in cash from the Target or of the Target s Subsidiaries on or in respect of the Target s or the Target s Subsidiaries share capital into the Dividend Account.

24. EVENTS OF DEFAULT

Each of the events or circumstances set out in this Clause 24 is an Event of Default.

24.1 Non-payment

An Obligor does not pay on the due date any amount payable pursuant to a Finance Document at the place at and in the currency in which it is expressed to be payable unless:

(a) its failure to pay is caused by:

103

Table of Contents

- (i) administrative or technical error; or
- (ii) a Disruption Event; and
- (b) payment is made within five Business Days of its due date.

24.2 Financial covenants and other obligations

- (a) Any requirement of Clause 22 (*Financial covenants*) is not satisfied or an Obligor does not comply with the provisions of Clauses 21 (*Information Undertakings*) and/or Clause 23.14 (*Negative pledge*) or 23.21 (*Financial Indebtedness*) or 23.36 (*Conditions Subsequent in relation to Warrants*) or 23.37 (*Conditions Subsequent in relation to German and Dutch Transaction Security*).
- (b) An Obligor or the Acquiror does not comply with any provision of any Transaction Security Document.
- (c) The Company does not comply with its obligations under the Engagement Letter.

24.3 Other obligations

- (a) An Obligor or the Acquiror does not comply with any provision of the Finance Documents (other than those referred to in Clause 24.1 (*Non-payment*) and Clause 24.2 (*Financial covenants and other obligations*)).
- (b) No Event of Default under paragraph (a) above will occur if the failure to comply is capable of remedy and is remedied within 10 Business Days of the Agent giving notice to the Company, the Acquiror or relevant Obligor or the Company, the Acquiror or an Obligor becoming aware of the failure to comply.

24.4 Misrepresentation

- (a) Any representation or statement made or deemed to be made by an Obligor or the Acquiror in the Finance Documents or any other document delivered by or on behalf of any Obligor or the Acquiror under or in connection with any Finance Document is or proves to have been incorrect or misleading when made or deemed to be made.
- (b) No Event of Default under paragraph (a) above will occur if the failure to comply is capable of remedy and is remedied within 10 Business Days of the Agent giving notice to the Company, the Acquiror or the relevant Obligor or the Company, the Acquiror or an Obligor becoming aware of the failure to comply.

24.5 Cross default

(a) Any Financial Indebtedness of any member of the Group is not paid when due nor within any originally applicable grace period.

104

Table of Contents

- (b) Any Financial Indebtedness of any member of the Group is declared to be or otherwise becomes due and payable prior to its specified maturity as a result of an event of default (however described).
- (c) Any commitment for any Financial Indebtedness of any member of the Group is cancelled or suspended by a creditor of any member of the Group as a result of an event of default (however described).
- (d) Any creditor of any member of the Group becomes entitled to declare any Financial Indebtedness of any member of the Group due and payable prior to its specified maturity as a result of an event of default (however described).
- (e) No Event of Default will occur under this Clause 24.5 if the aggregate amount of Financial Indebtedness or commitment for Financial Indebtedness falling within paragraphs (a) to (d) above is less than 500,000 (or its equivalent in any other currency or currencies).

24.6 Insolvency

- (a) A member of the Group is unable or admits inability to pay its debts as they fall due or is deemed to or declared to be unable to pay its debts under applicable law, suspends or threatens to suspend making payments on any of its debts or, by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors with a view to rescheduling any of its indebtedness.
- (b) The value of the assets of any member of the Group is less than its liabilities (taking into account contingent and prospective liabilities).
- (c) A moratorium is declared in respect of any indebtedness of any member of the Group. If a moratorium occurs, the ending of the moratorium will not remedy any Event of Default caused by that moratorium.
- (d) The Acquiror gives notice under section 36(2) of the Dutch 1990 Tax Collection Act (*Invorderingswet 1990*) separately or in conjunction with section 60 of the Dutch Act on the Financing of Social Insurance (*Wet Financiering sociale verzekeringen*).

24.7 Insolvency proceedings

- (a) Any corporate action, legal proceedings or other procedure or step is taken in relation to:
 - (i) the suspension of payments, a moratorium of any indebtedness, bankruptcy, winding-up, dissolution, administration or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise) of any member of the Group;
 - (ii) a composition, compromise, assignment or arrangement with any creditor of any member of the Group;

105

Table of Contents

- (iii) the appointment of a liquidator, receiver, administrator, administrative receiver, compulsory manager or other similar officer in respect of any member of the Group or any of its assets; or
- (iv) enforcement of any Security over any assets of any member of the Group, or any analogous procedure or step is taken in any jurisdiction.
 - (b) Paragraph (a) shall not apply to:
 - (i) any winding-up petition which is frivolous or vexatious and is discharged, stayed or dismissed within 14 days of commencement or, if earlier, the date on which it is advertised; or
 - (ii) any step or procedure contemplated by paragraph (c) of the definition of Permitted Transaction.

24.8 Creditors process

Any expropriation, attachment, sequestration, distress or execution or any analogous process in any jurisdiction affects any asset or assets of a member of the Group having an aggregate value of 500,000 and is not discharged within 21 days.

24.9 Unlawfulness and invalidity

- (a) It is or becomes unlawful for an Obligor or the Acquiror to perform any of its obligations under the Finance Documents or any Transaction Security created or expressed to be created or evidenced by the Transaction Security Documents ceases to be effective or any subordination created under the Intercreditor Agreement is or becomes unlawful.
- (b) Any obligation or obligations of any Obligor or the Acquiror under any Finance Documents are not (subject to the Legal Reservations) or cease to be legal, valid, binding or enforceable and the cessation individually or cumulatively materially and adversely affects the interests of the Lenders under the Finance Documents.
- (c) Any Finance Document ceases to be in full force and effect or any Transaction Security or any subordination created under the Intercreditor Agreement ceases to be legal, valid, binding, enforceable or effective or is alleged by a party to it (other than a Finance Party) to be ineffective.

24.10 Cessation of business

Any member of the Group suspends or ceases to carry on (or threatens to suspend or cease to carry on) all or a material part of its business except as a result of a Permitted Disposal or a Permitted Transaction.

24.11 Change of ownership

106

Table of Contents

- (a) An Obligor (other than the Company) or the Acquiror ceases to be a wholly-owned Subsidiary of the Company; or
- (b) After the Closing Date, the Target and its Subsidiaries cease to be wholly-owned Subsidiaries of the Company,

except, in either case, as a result of a disposal which is a Permitted Disposal or a Permitted Transaction.

24.12 Audit qualification

The Auditors of the Group qualify the audited annual consolidated financial statements of the Company.

24.13 Expropriation

The authority or ability of any member of the Group to conduct its business is limited or wholly or substantially curtailed by any seizure, expropriation, nationalisation, intervention, restriction or other action by or on behalf of any governmental, regulatory or other authority or other person in relation to any member of the Group or any of its assets.

24.14 Repudiation and rescission of agreements

- (a) An Obligor or the Acquiror (or any other relevant party) rescinds or purports to rescind or repudiates or purports to repudiate a Finance Document or any of the Transaction Security to which it is a party or evidences an intention to rescind or repudiate a Finance Document or any Transaction Security to which it is a party.
- (b) Any party to the Acquisition Documents or the Warrant Documents rescinds or purports to rescind or repudiates or purports to repudiate any of those agreements or instruments in whole or in part where to do so has or is, in the reasonable opinion of the Majority Lenders, likely to have a material adverse effect on the interests of the Lenders under the Finance Documents.

24.15 Litigation

Any litigation, arbitration, administrative, governmental, regulatory or other investigations, proceedings or disputes are commenced or threatened in relation to the Transaction Documents or the transactions contemplated in the Transaction Documents or against any member of the Group or its assets which has or is reasonably likely to have a Material Adverse Effect.

24.16 Material adverse change

Any event or circumstance occurs which the Majority Lenders reasonably believe has or is reasonably likely to have a Material Adverse Effect.

107

Table of Contents

24.17 ERISA Event of Default

- (a) Any ERISA Event shall have occurred and the sum (determined as of the date of occurrence of such ERISA Event) of the Insufficiency of such Plan and the Insufficiency of any and all other Plans with respect to which an ERISA Event shall have occurred and then exist (or the liability of the Obligors and the ERISA Affiliates related to such ERISA Event) exceeds 500,000 (or its equivalent in other currencies).
- (b) Any Obligor or any ERISA Affiliate shall have been notified in writing by the sponsor of a Multiemployer Plan that it has incurred Withdrawal Liability to such Multiemployer Plan in an amount that, when aggregated with all other amounts required to be paid to Multiemployer Plans by the Obligors and the ERISA Affiliates as Withdrawal Liability (determined as of the date of such notification), exceeds 500,000 (or its equivalent in other currencies) or requires payments exceeding 500,000 (or its equivalent in other currencies) per annum.
- (c) Any Obligor or any ERISA Affiliate shall have been notified in writing by the sponsor of a Multiemployer Plan that such Multiemployer Plan is in reorganisation or is being terminated, within the meaning of Title IV of ERISA, and as a result of such reorganisation or termination the aggregate annual contributions of the Obligors and the ERISA Affiliates to all Multiemployer Plans that are then in reorganisation or being terminated have been or will be increased over the amounts contributed to such Multiemployer Plans for the plan years of such Multiemployer Plans immediately preceding the plan year in which such reorganisation or termination occurs by an amount exceeding 500,000 (or its equivalent in other currencies).

24.18 Acceleration

- (a) On and at any time after the occurrence of an Event of Default which is continuing the Agent may, and shall if so directed by the Majority Lenders, by notice to the Company:
 - (i) cancel the Total Commitments at which time they shall immediately be cancelled;
 - (ii) declare that all or part of the Utilisations, together with accrued interest, and all other amounts accrued or outstanding under the Finance Documents be immediately due and payable, at which time they shall become immediately due and payable;
 - (iii) declare that all or part of the Utilisations be payable on demand, at which time they shall immediately become payable on demand by the Agent on the instructions of the Majority Lenders;
 - (iv) declare that cash cover in respect of the Letter of Credit is immediately due and payable at which time it shall become immediately due and payable;

108

- (v) declare that cash cover in respect of the Letter of Credit is payable on demand at which time it shall immediately become due and payable on demand by the Agent on the instructions of the Majority Lenders;
- (vi) exercise or direct the Security Agent to exercise any or all of its rights, remedies, powers or discretions under the Finance Documents.
- (b) In the event of a declaration pursuant to paragraph (a)(ii) or (a)(iv) or demand as provided for under (a)(iii) or (a)(v) above, the Borrower shall, within three Business Days of such declaration or demand, pay to the Agent a sum calculated using the formula:

$$VP \times (M \times \frac{N}{360})$$

$$\frac{1 + (Discount N)}{Rate \times 360}$$

where

VP = the amount notified to the Borrower as due and payable pursuant to paragraph (a)(ii), (a)(iii), (a)(iv) or (a)(v) above.

M = the Margin on the date of such declaration or demand.

N = the number of days between the date of notification of the Borrower by the Agent of the amount due and payable and the Make Whole Date.

Discount Rate = EURIBOR in relation to the declaration or demand date.

24.19 US Bankruptcy

Notwithstanding Clause 24.18 (*Acceleration*), upon the actual or deemed entry of an order for relief under the US Bankruptcy Code with respect to any US Obligor, the Facility shall cease to be available to such Obligor, all Utilisations outstanding to such Obligor, and all other amounts owed by such Obligor under the Finance Documents, shall become immediately due and payable, in each case automatically and without any further action by any party hereto.

109

SECTION 9 CHANGES TO PARTIES

25. CHANGES TO THE LENDERS AND OBLIGORS

25.1 Assignments and transfers by the Lenders

Subject to this Clause 25, a Lender (the **Existing Lender**) may:

- (a) assign any of its rights; or
- (b) transfer by novation any of its rights and obligations, under any Finance Document to another bank or financial institution or to a trust, fund or other entity which is regularly engaged in or established for the purpose of making, purchasing or investing in loans, securities or other financial assets (the **New Lender**).

25.2 Conditions of assignment or transfer

- (a) An Existing Lender must consult with the Company for not more than two Business Days before it makes an assignment or transfer under Clause 25.1 (*Assignment and transfer by the Lenders*), but the consent of the Company and the Borrower is not required for an assignment or transfer by a Lender unless it is to a person whose principal business is in competition with the principal business of the Company, in which case the prior written consent of the Borrower (such consent not to be unreasonably withheld or delayed and to be deemed given if no response is received within three Business Days of consent being requested) shall be required for such assignment or transfer. However no consent or consultation is required if the assignment or transfer is:
 - (i) to another Lender or an Affiliate of a Lender;
 - (ii) if the Existing Lender is a fund (the **first fund**), to a fund which is managed or advised by the same investment manager or adviser as the first fund or, if it is managed by a different investment manager or adviser, a fund whose investment adviser or manager is an Affiliate of the investment manager or adviser of the first fund;
 - (iii) made at a time when an Event of Default is continuing.
- (b) The consent of the Issuing Bank is required for an assignment or transfer by a Lender if the Letter of Credit is outstanding.
- (c) An assignment will only be effective on:
 - (i) receipt by the Agent (whether in the Assignment Agreement or otherwise) of written confirmation from the New Lender (in form and substance satisfactory to the Agent) that the New Lender will assume the same obligations to the other Finance Parties and the other Secured Parties as it would have been under if it was an Original Lender;

110

Table of Contents

- (ii) the recordation of such assignment on the Register pursuant to Clause 25.11 (*The Register*);
- (iii) performance by the Agent of all necessary know your customer or other similar checks under all applicable laws and regulations in relation to such assignment to a New Lender, the completion of which the Agent shall promptly notify to the Existing Lender and the New Lender; and
- (iv) compliance with the procedure set out in Clause 25.6 (*Procedure for assignment*).
- (d) A transfer will only be effective if the procedure set out in Clause 25.5 (*Procedure for transfer*) is complied with.
- (e) If:
 - (i) a Lender assigns or transfers any of its rights or obligations under the Finance Documents or changes its Facility Office; and
 - (ii) as a result of circumstances existing at the date the assignment, transfer or change occurs, an Obligor or, at the procurement of the Company, the Acquiror would be obliged to make a payment to the New Lender or Lender acting through its new Facility Office under Clause 14 (*Tax gross-up and indemnities*) or Clause 15.1 (*Increased Costs*),

then the New Lender or Lender acting through its new Facility Office is only entitled to receive payment under those Clauses to the same extent as the Existing Lender or Lender acting through its previous Facility Office would have been if the assignment, transfer or change had not occurred.

25.3 Assignment or transfer fee

The New Lender shall, on the date upon which an assignment or transfer takes effect, pay to the Agent (for its own account) a fee of 1,500.

25.4 Limitation of responsibility of Existing Lenders

- (a) Unless expressly agreed to the contrary, an Existing Lender makes no representation or warranty and assumes no responsibility to a New Lender for:
 - (i) the legality, validity, effectiveness, adequacy or enforceability of the Transaction Documents, the Transaction Security or any other documents;
 - (ii) the financial condition of any Obligor or the Acquiror;
 - (iii) the performance and observance by any Obligor or any other member of the Group of its obligations under the Transaction Documents or any other documents; or

111

Table of Contents

- (iv) the accuracy of any statements (whether written or oral) made in or in connection with any Transaction Document or any other document, and any representations or warranties implied by law are excluded.
 - (b) Each New Lender confirms to the Existing Lender, the other Finance Parties and the Secured Parties that it:
 - (i) has made (and shall continue to make) its own independent investigation and assessment of the financial condition and affairs of each Obligor, the Acquiror and their related entities in connection with its participation in this Agreement and has not relied exclusively on any information provided to it by the Existing Lender or any other Finance Party in connection with any Transaction Document or the Transaction Security; and
 - (ii) will continue to make its own independent appraisal of the creditworthiness of each Obligor, the Acquiror and their related entities whilst any amount is or may be outstanding under the Finance Documents or any Commitment is in force.
 - (c) Nothing in any Finance Document obliges an Existing Lender to:
 - (i) accept a re-transfer from a New Lender of any of the rights and obligations assigned or transferred under this Clause 25; or
 - (ii) support any losses directly or indirectly incurred by the New Lender by reason of the non-performance by any Obligor or, at the procurement of the Company, the Acquiror of its obligations under the Transaction Documents or otherwise.

25.5 **Procedure for transfer**

- (a) Subject to the conditions set out in Clause 25.2 (*Conditions of assignment or transfer*) a transfer is effected in accordance with paragraph (c) below when (i) the Agent executes an otherwise duly completed Transfer Certificate delivered to it by the Existing Lender and the New Lender; and (ii) the transfer is recorded on the Register. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Transfer Certificate appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Transfer Certificate and record the transfer on the Register.
- (b) The Agent shall only be obliged to execute a Transfer Certificate delivered to it by the Existing Lender and the New Lender once it is satisfied it has complied with all necessary know your customer or other similar checks under all applicable laws and regulations in relation to the transfer to such New Lender.
- (c) On the Transfer Date:

112

Table of Contents

- (i) to the extent that in the Transfer Certificate the Existing Lender seeks to transfer by novation its rights and obligations under the Finance Documents each of the Obligors, the Acquiror and the Existing Lender shall be released from further obligations towards one another under the Finance Documents and in respect of the Transaction Security and their respective rights against one another under the Finance Documents and in respect of the Transaction Security shall be cancelled (being the **Discharged Rights and Obligations**);
- (ii) each of the Obligors, at the procurement of the Company, the Acquiror and the New Lender shall assume obligations towards one another and/or acquire rights against one another which differ from the Discharged Rights and Obligations only insofar as that Obligor, the Acquiror or other member of the Group and the New Lender have assumed and/or acquired the same in place of that Obligor, the Acquiror and the Existing Lender;
- (iii) the Agent, the Arranger, the New Lender, the Security Agent, the Issuing Bank and other Lenders shall acquire the same rights and assume the same obligations between themselves and in respect of the Transaction Security as they would have acquired and assumed had the New Lender been an Original Lender with the rights and/or obligations acquired or assumed by it as a result of the transfer and to that extent the Agent, the Arranger, the Security Agent, the Issuing Bank and the Existing Lender shall each be released from further obligations to each other under the Finance Documents; and
- (iv) the New Lender shall become a Party as a Lender .

25.6 **Procedure for assignment**

- (a) Subject to the conditions set out in Clause 25.2 (*Conditions of assignment or transfer*) an assignment may be effected in accordance with paragraph (c) below when (i) the Agent executes an otherwise duly completed Assignment Agreement delivered to it by the Existing Lender and the New Lender and (ii) the transfer is recorded on the Register. The Agent shall, subject to paragraph (b) below, as soon as reasonably practicable after receipt by it of a duly completed Assignment Agreement appearing on its face to comply with the terms of this Agreement and delivered in accordance with the terms of this Agreement, execute that Assignment Agreement and record the transfer on the Register.
- (b) The Agent shall only be obliged to execute an Assignment Agreement delivered to it by the Existing Lender and the New Lender upon its completion of all know your customer or other checks relating to any person that it is required to carry out in relation to the assignment to such New Lender.
- (c) On the Transfer Date:
 - (i) the Existing Lender will assign absolutely to the New Lender its rights under the Finance Documents and in respect of the Transaction

113

Table of Contents

Security expressed to be the subject of the assignment in the Assignment Agreement;

- (ii) the Existing Lender will be released from the obligations (the **Relevant Obligations**) expressed to be the subject of the release in the Assignment Agreement (and any corresponding obligations by which it is bound in respect of the Transaction Security); and
- (iii) the New Lender shall become a Party as a Lender and will be bound by obligations equivalent to the Relevant Obligations.
- (d) Lenders may utilise procedures other than those set out in this Clause 29.6 to assign their rights under the Finance Documents **provided that** they comply with the conditions set out in Clause 25.2 (*Conditions of assignment or transfer*).

25.7 Copy of Transfer Certificate or Assignment Agreement to Borrower

The Agent shall, as soon as reasonably practicable after it has executed a Transfer Certificate or an Assignment Agreement, send to the Borrower a copy of that Transfer Certificate or Assignment Agreement.

25.8 **Disclosure of information**

Any Lender may disclose to any of its Affiliates and any other person:

- (a) to (or through) whom that Lender assigns or transfers (or may potentially assign or transfer) all or any of its rights and obligations under the Finance Documents;
- (b) with (or through) whom that Lender enters into (or may potentially enter into) any sub-participation in relation to, or any other transaction under which payments are to be made by reference to, the Finance Documents or any Obligor or the Acquiror; or
- (c) to whom, and to the extent that, information is required to be disclosed by any applicable law or regulation.

any information about any Obligor, the Acquiror, the Group and the Finance Documents as that Lender shall consider appropriate if in relation to paragraph (a) or (b) above, the person to whom the information is to be given has entered into a Confidentiality Undertaking.

25.9 Assignment and transfers by Obligors

No Obligor, nor, at the procurement of the Company, the Acquiror may assign any of its rights or transfer any of its rights or obligations under the Finance Documents to which it is a party.

114

Table of Contents

25.10 Additional Guarantors

- (a) The Company shall ensure that each member of the Group identified in Part IV of Schedule 2 (*Conditions Precedent*) as an Additional Guarantor shall become an Additional Guarantor and shall grant the Transaction Security identified opposite the name of that member of the Group in Part IV of Schedule 2 (*Conditions Precedent*) on or prior to the date specified in Part IV of that Schedule.
- (b) Subject to the limitations described in Clause 23.31(b) (*Guarantors*), the Company shall procure that any other member of the Group shall, as soon as possible after becoming a Subsidiary of the Company, shall become an Additional Guarantor and grant Security as the Agent may require.
- (c) A member of the Group shall become an Additional Guarantor if:
 - (i) the Company and the proposed Additional Guarantor deliver to the Agent a duly completed and executed Accession Letter; and
 - (ii) the Agent has received all of the documents and other evidence listed in Part II and, if applicable, Part III of Schedule 2 (*Conditions Precedent*) in relation to that Additional Guarantor, each in form and substance satisfactory to the Agent.
- (d) The Agent shall notify the Company and the Lenders promptly upon being satisfied that it has received (in form and substance satisfactory to it) all the documents and other evidence listed in Part II and, if applicable, Part III of Schedule 2 (*Conditions Precedent*)

25.11 The Register

- (a) The Agent, acting solely for this purpose as the agent of the Obligors and at the procurement of the Company, the Acquiror, shall maintain at its address referred to in Clause 31 (*Notices*):
 - (i) a copy of each notice and confirmation referred to in Clause 25.2 (*Conditions of assignment or transfer*) and each Transfer Certificate and Assignment Agreement referred to in Clause 25.5 (*Procedure for transfer*) or 25.6 (*Procedure for assignment*) delivered to and accepted by it; and
 - (ii) a register for the recording of the names and addresses of the Lenders and the Commitment of, and principal amount owing to, each Lender from time to time (the **Register**).

The entries in the Register shall be conclusive and binding for all purposes, absent manifest error, and the Obligors, at the procurement of the Company, the Acquiror, the Agent, the Security Agent, the Issuing Bank and the Lenders shall treat each person whose name is recorded in the Register as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by any Obligor or, at the procurement of the Company, the

Table of Contents

Acquiror at any reasonable time and from time to time upon reasonable prior notice.

(b) Each Party to this Agreement irrevocably authorises the Agent to make the relevant entry in the Register on its behalf for the purposes of this Clause 25.11 (*The Register*) without any further consent of, or consultation with, such Party.

25.12 Repetition of Representations

Delivery of an Accession Letter constitutes confirmation by the relevant Subsidiary that the representations and warranties referred to in paragraph (c) of Clause 20.33 (*Times when representations made*) are true and correct in relation to it as at the date of delivery as if made by reference to the facts and circumstances then existing.

116

SECTION 10 THE FINANCE PARTIES

26. ROLE OF THE AGENT, SECURITY AGENT, THE ARRANGER, THE ISSUING BANK AND OTHERS

26.1 Appointment of the Agent and Security Agent

- (a) Each of the Arranger, the Lenders and the Issuing Bank appoints the Agent to act as its agent under and in connection with the Finance Documents.
- (b) Each of the Arranger, the Lenders and the Issuing Bank authorises the Agent to exercise the rights, powers, authorities and discretions specifically given to the Agent under or in connection with the Finance Documents together with any other incidental rights, powers, authorities and discretions.
- (c) Each of the Finance Parties hereby appoints the Security Agent to act as security agent and trustee in connection with the Finance Documents and authorises the Security Agent to exercise such rights, powers, authorities and discretions as are specifically delegated to the Security Agent by the terms of the Finance Documents, together with all such rights, powers, authorities and discretions as are reasonably incidental thereto. Without limitation to the foregoing, each Finance Party, by executing this Agreement, irrevocably appoints the Security Agent to act as its agent under and in connection with the Security Documents and irrevocably authorises the Security Agent on its behalf to execute such Security Documents.
- (d) Each Finance Party (other than, in each case, the appointee) authorises each of the Agent and the Security Agent to:
 - (i) execute each Finance Document to which it is expressed to be a party in its name and on its behalf and to enter into any agreements or other documents or certificates incidental or ancillary thereto; and
 - (ii) appear before any public notaries, registrars, public officers, courts and governmental authorities to file any necessary documents and to take such action as may be required to give full effect to this Clause 26.1(d).

26.2 Duties of the Agent and Security Agent

- (a) Each of the Security Agent and the Agent shall promptly forward to a Party the original or a copy of any document which is delivered to the Agent or Security Agent, as the case may be, for that Party by any other Party.
- (b) Except where a Finance Document specifically provides otherwise, neither the Agent nor the Security Agent is obliged to review or check the adequacy, accuracy or completeness of any document it forwards to another Party.
- (c) If the Agent or the Security Agent receives notice from a Party referring to this Agreement, describing a Default and stating that the circumstance described is a Default, it shall promptly notify the Finance Parties.

117

Table of Contents

- (d) If the Agent or the Security Agent is aware of the non-payment of any principal, interest, commitment fee or other fee payable to a Finance Party (other than the Agent, the Arranger or the Security Agent) under this Agreement it shall promptly notify the other Finance Parties.
- (e) The Agent, acting solely for this purpose as the agent of the Obligors and, at the procurement of the Company, the Acquiror, shall maintain the Register as provided in Clause 25.11 (*The Register*).
- (f) The Agent's duties under the Finance Documents are solely mechanical and administrative in nature.

26.3 Role of the Arranger

Except as specifically provided in the Finance Documents, the Arranger has no obligations of any kind to any other Party under or in connection with any Finance Document.

26.4 No fiduciary duties

- (a) Nothing in this Agreement constitutes the Agent, the Arranger, the Security Agent or the Issuing Bank as a fiduciary of any other person.
- (b) None of the Agent, the Security Agent, the Arranger or the Issuing Bank shall be bound to account to any Lender for any sum or the profit element of any sum received by it for its own account.

26.5 **Business with the Group**

The Agent, the Security Agent, the Issuing Bank and the Arranger may accept deposits from, lend money to and generally engage in any kind of banking or other business with any member of the Group.

26.6 Rights and discretions of the Agent, Security Agent, Issuing Bank and Arranger

- (a) Each of the Security Agent and the Agent may rely on:
 - (i) any representation, notice or document believed by it to be genuine, correct and appropriately authorised; and
 - (ii) any statement made by a director, authorised signatory or employee of any person regarding any matters which may reasonably be assumed to be within his knowledge or within his power to verify.
- (b) Each of the Security Agent and the Agent may assume (unless it has received notice to the contrary in its capacity as agent or security agent for the Lenders) that:
 - (i) no Default has occurred (unless it has actual knowledge of a Default arising under Clause 24.1 (*Non-payment*));

118

Table of Contents

- (ii) any right, power, authority or discretion vested in any Party or the Majority Lenders has not been exercised:
- (iii) any notice or request made by the Borrower (other than a Utilisation Request) is made on behalf of and with the consent and knowledge of all the Obligors and, at the procurement of the Company, the Acquiror; and
- (iv) any Transaction Security created under any Transaction Security Document is legal, valid and enforceable and creates a valid security interest in favour of the Finance Parties without any obligations for the Security Agent to carry out any specific or independent investigation unless otherwise provided under this Agreement.
- (c) Each of the Security Agent and the Agent may engage, pay for and rely on the advice or services of any lawyers, accountants, surveyors or other experts.
- (d) Each of the Security Agent and the Agent may act in relation to the Finance Documents through its personnel and agents.
- (e) Each of the Security Agent and the Agent may disclose to any other Party any information it reasonably believes it has received as agent or security agent under this Agreement.
- (f) Notwithstanding any other provision of any Finance Document to the contrary, none of the Agent, the Security Agent, the Issuing Bank or the Arranger is obliged to do or omit to do anything if it would or might in its reasonable opinion constitute a breach of any law or regulation or a breach of a fiduciary duty or duty of confidentiality.
- (g) Each of the Security Agent and the Agent shall be at liberty to place any Finance Document and any other instruments, documents or deeds delivered to it pursuant thereto or in connection therewith for the time being in its possession in any safe deposit, safe or receptacle selected by the Security Agent or the Agent, as the case may be, or with any bank, any company whose business includes undertaking the safe custody of documents or any firm of lawyers of good repute and, in the absence of gross negligence or wilful default on the part of the Security Agent or the Agent it shall not be responsible for any loss thereby incurred.
- (h) Each of the Security Agent and the Agent may, whenever it thinks fit, delegate by power of attorney, appointment of agent or otherwise to any person or persons selected by it with reasonable care all or any of the rights, trusts, powers, authorities and discretions vested in it by any Finance Document and such delegation may be made upon such terms and subject to such conditions and subject to such regulations as the Security Agent or the Agent, as the case may be, may think fit and neither the Security Agent nor the Agent shall be bound to supervise the proceedings or (in the absence of gross negligence or wilful default on the part of the Security Agent or the Agent, as the case may be,) be in any way responsible for any loss incurred by reason of any misconduct or default on the part of any such delegate.

119

- (i) Each of the Security Agent and the Agent and every attorney, agent or other person appointed by it under any Finance Document may indemnify itself or himself out of the Trust Property against all of the Agent s and Security Agent s liabilities and/or in respect of any other matter or thing done or omitted to be done in any way relating to any Finance Document or by law and/or acting as Security Agent or Agent unless the liability is caused by the gross negligence or wilful misconduct of the Security Agent or Agent.
- (j) The Security Agent and the Agent shall each have power to institute, prosecute and defend any suits or actions or other proceedings affecting the Agent or the Security Agent respectively or the Trust Property and to compromise any matter or difference or submit any such matter to arbitration and to compromise or compound any debts owing to the Agent or the Security Agent respectively as such agent or security agent or any other claims against it upon such evidence as the Agent or the Security Agent, as the case may be, as shall deem sufficient and to make partition upon such terms as the Agent or the Security Agent, as the case may be, shall deem desirable.
- (k) Save as otherwise expressly provided herein, each of the Security Agent and the Agent shall have absolute discretion as to the exercise or non exercise (and as to the manner and time of any such exercise) of all rights, trust, powers, authorities and discretions vested in it by any of the Finance Documents but shall be entitled to refrain from exercising any right, power or discretion vested in it as agent or trustee under any Finance Document unless and until instructed by the Majority Lenders as to whether or not such right, power or discretion is to be exercised and, if it is to be exercised, as to the manner in which it should be exercised and (in the absence of gross negligence or wilful default on the part of the Agent or Security Agent, as the case may be) shall not be responsible for any loss occasioned by such exercise or non exercise.
- (1) At any time after the Security Documents have become enforceable in accordance with their respective terms, each of the Security Agent and the Agent shall have power to give or enter into any indemnity, warranty, guarantee, undertaking or covenant or to enter into any type of agreement as it shall, with the approval of the Majority Lenders and subject to all other provisions of the Finance Documents, think fit in relation to the Trust Property.

26.7 Majority Lenders instructions

- (a) Unless a contrary indication appears in a Finance Document, each of the Security Agent and the Agent shall (i) exercise any right, power, authority or discretion vested in it as Agent in accordance with any instructions given to it by the Majority Lenders (or, if so instructed by the Majority Lenders, refrain from exercising any right, power, authority or discretion vested in it as Agent or Security Agent) and (ii) not be liable for any act (or omission) if it acts (or refrains from taking any action) in accordance with an instruction of the Majority Lenders.
- (b) Unless a contrary indication appears in a Finance Document, any instructions given by the Majority Lenders will be binding on all the Finance Parties.

120

Table of Contents

- (c) Each of the Security Agent and the Agent may refrain from acting in accordance with the instructions of the Majority Lenders (or, if appropriate, the Lenders) until it has received such security as it may require for any cost, loss or liability (together with any associated VAT) which it may incur in complying with the instructions.
- (d) In the absence of instructions from the Majority Lenders, (or, if appropriate, the Lenders) the Agent or the Security Agent may act (or refrain from taking action) as it considers to be in the best interest of the Lenders.
- (e) The Agent and the Security Agent may refrain from exercising any right, power or discretion vested in it as agent hereunder unless and until instructed by the Majority Lenders as to whether or not such right, power or discretion is to be exercised and, if it is to be exercised, as to the manner in which it should be exercised.
- (f) The Agent is not authorised to act on behalf of a Lender (without first obtaining that Lender s consent) in any legal or arbitration proceedings relating to any Finance Document. This paragraph (e) shall not apply to any legal or arbitration proceeding relating to the perfection, preservation or protection of rights under the Transaction Security Documents or enforcement of the Transaction Security or Transaction Security Documents.

26.8 Responsibility for documentation

None of the Agent, the Security Agent, the Arranger or the Issuing Bank:

- (a) is responsible for the adequacy, accuracy and/or completeness of any information (whether oral or written) supplied by the Agent, the Arranger, the Security Agent, the Issuing Bank, an Obligor, at the procurement of the Company, the Acquiror or any other person given in or in connection with any Finance Document or the Reports or the transactions contemplated by the Finance Documents; or
- (b) is responsible for the legality, validity, effectiveness, adequacy or enforceability of any Finance Document or the Transaction Security or any other agreement, arrangement or document entered into, made or executed in anticipation of or in connection with any Finance Document or the Transaction Security.

26.9 Exclusion of liability

(a) Without limiting paragraph (b) below (and without prejudice to the provisions of paragraph (e) of Clause 29.10 (*Disruption to Payment Systems etc.*), neither the Issuing Bank, the Security Agent nor the Agent will be liable (including, without limitation, for negligence or any other category of liability whatsoever) for any action taken by it under or in connection with any Finance Document or the Transaction Security, unless directly caused by its gross negligence or wilful misconduct.

121

Table of Contents

- (b) No Party (other than the Agent, the Security Agent or the Issuing Bank) may take any proceedings against any officer, employee or agent of the Agent, the Security Agent or the Issuing Bank in respect of any claim it might have against the Agent, the Security Agent or the Issuing Bank or in respect of any act or omission of any kind by that officer, employee or agent in relation to any Finance Document or any Transaction Security and any officer, employee or agent of the Agent, the Security Agent or the Issuing Bank may rely on this Clause subject to Clause 1.3 (*Third Party Rights*) and the provisions of the Third Parties Act.
- (c) Neither the Agent nor the Security Agent will not be liable for any delay (or any related consequences) in crediting an account with an amount required under the Finance Documents to be paid by the Agent or the Security Agent if the Agent or the Security Agent has taken all necessary steps as soon as reasonably practicable to comply with the regulations or operating procedures of any recognised clearing or settlement system used by the Agent or the Security Agent for that purpose.
- (d) Nothing in this Agreement shall oblige the Agent, the Security Agent or the Arranger to carry out any know your customer or other checks in relation to any person on behalf of any Lender and each Lender confirms to the Agent, the Security Agent and the Arranger that it is solely responsible for any such checks it is required to carry out and that it may not rely on any statement in relation to such checks made by the Agent, the Security Agent or the Arranger.
- (e) Neither the Agent nor the Security Agent shall be bound to enquire as to (i) whether or not any representation made or deemed to be made by an Obligor in connection with the Finance Documents is true, (ii) the occurrence or otherwise of any Default, (iii) the performance by an Obligor or the Acquiror of its obligations under the Finance Documents or (iv) any breach of or default by an Obligor or the Acquiror of or under its obligations under the Finance Documents;

26.10 Lenders indemnity to the Agent or the Security Agent

Each Lender shall (in proportion to its share of the Total Commitments or, if the Total Commitments are then zero, to its share of the Total Commitments immediately prior to their reduction to zero) indemnify the Agent or the Security Agent, within three Business Days of demand, against any cost, loss or liability (including, without limitation, for negligence or any other category of liability whatsoever) incurred by the Agent or the Security Agent (otherwise than by reason of the Agent s or the Security Agent s gross negligence or wilful misconduct) (or, in the case of any cost, loss or liability pursuant to Clause 29.10 (*Disruption to Payment Systems etc.*) notwithstanding the Agent s or the Security Agent s negligence, gross negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent or the Security Agent) in acting as Agent or the Security Agent under the Finance Documents (unless the Agent or the Security Agent has been reimbursed by an Obligor or the Acquiror pursuant to a Finance Document).

26.11 Resignation of the Agent and Security Agent

122

Table of Contents

- (a) Each of the Agent and Security Agent may resign and appoint one of its Affiliates acting through an office in the United Kingdom as successor by giving notice to the other Finance Parties and the Company.
- (b) Alternatively the Agent or the Security Agent may resign by giving notice to the other Finance Parties and the Company, in which case the Majority Lenders (after consultation with the Company) may appoint a successor Agent or Security Agent as the case may be.
- (c) If the Majority Lenders have not appointed a successor Agent or Security Agent as the case may be in accordance with paragraph (b) above within 30 days after notice of resignation was given, the Agent or Security Agent as the case may be (after consultation with the Borrower) may appoint a successor Agent (acting through an office in the United Kingdom).
- (d) The retiring Agent or Security Agent as the case may be shall, at its own cost, make available to the successor Agent or Security Agent as the case may be such documents and records and provide such assistance as the successor Agent or Security Agent as the case may be may reasonably request for the purposes of performing its functions as Agent or Security Agent as the case may be under the Finance Documents.
- (e) The Agent s or Security Agent s, as the case may be, resignation notice shall only take effect upon the appointment of a successor.
- (f) Upon the appointment of a successor, the retiring Agent or Security Agent as the case may be shall be discharged from any further obligation in respect of the Finance Documents but shall remain entitled to the benefit of this Clause 26. Its successor and each of the other Parties shall have the same rights and obligations amongst themselves as they would have had if such successor had been an original Party.
- (g) After consultation with the Company, the Majority Lenders may, by notice to the Agent or Security Agent as the case may be, require it to resign in accordance with paragraph (b) above. In this event, the Agent or Security Agent as the case may be shall resign in accordance with paragraph (b) above.

26.12 Confidentiality

- (a) In acting as agent or security agent for the Finance Parties, each of the Agent and the Security Agent shall be regarded as acting through its agency division which shall be treated as a separate entity from any other of its divisions or departments.
- (b) If information is received by another division or department of the Agent or the Security Agent, it may be treated as confidential to that division or department and the Agent and the Security Agent shall not be deemed to have notice of it.
- (c) Notwithstanding any other provision of any Finance Document to the contrary, none of the Agent, the Security Agent or the Arranger are obliged to

123

Table of Contents

disclose to any other person (i) any confidential information or (ii) any other information if the disclosure would or might in its reasonable opinion constitute a breach of any law or a breach of a fiduciary duty.

26.13 Relationship with the Lenders

- (a) The Agent and the Security Agent may treat each Lender as a Lender, entitled to payments under this Agreement and acting through its Facility Office unless it has received not less than five Business Days prior notice from that Lender to the contrary in accordance with the terms of this Agreement.
- (b) Each Lender shall supply the Agent with any information required by the Agent in order to calculate the Mandatory Cost in accordance with Schedule 4 (*Mandatory Cost formula*).
- (c) Each Lender shall supply the Agent with any information that the Security Agent may reasonably specify (through the Agent) as being necessary or desirable to enable the Security Agent to perform its functions as Security Agent. Each Lender shall deal with the Security Agent exclusively through the Agent and shall not deal directly with the Security Agent.
- (d) The Security Agent shall be entitled to, and shall, carry out all dealings with the other Finance Parties through the Facility Agent and shall be entitled to rely on the Agent s certificate as to the entitlement of all or any of the Finance Parties.
- (e) Each of the Obligors hereby waives, to the extent permitted under applicable law, all rights it may otherwise have to require that the Transaction Security created pursuant to the Finance Documents be enforced in any particular order or manner.

26.14 Credit appraisal by the Lenders

Without affecting the responsibility of any Obligor or, at the procurement of the Company, the Acquiror for information supplied by it or on its behalf in connection with any Finance Document, each Lender and the Issuing Bank confirms to the Agent, the Security Agent, the Issuing Bank and the Arranger that it has been, and will continue to be, solely responsible for making its own independent appraisal and investigation of all risks arising under or in connection with any Finance Document including but not limited to:

- (a) the financial condition, status and nature of each member of the Group;
- (b) the legality, validity, effectiveness, adequacy or enforceability of any Finance Document and the Transaction Security and any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document or the Transaction Security;
- (c) whether that Lender has recourse, and the nature and extent of that recourse, against any Party or any of its respective assets under or in connection with any Finance Document, the Transaction Security, the transactions

124

Table of Contents

contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document;

- (d) the adequacy, accuracy and/or completeness of the Reports and any information provided by the Agent, the Security Agent, any Party or by any other person under or in connection with any Finance Document, the transactions contemplated by the Finance Documents or any other agreement, arrangement or document entered into, made or executed in anticipation of, under or in connection with any Finance Document; and
- (e) the right or title of any person, in or to, or the value or sufficiency of the Charged Property, the priority of any of the Transaction Security or existence of any Security affecting the Charged Property.

26.15 Agent s and the Security Agent s Management Time

Any amount payable to the Agent or the Security Agent under Clause 16.3 (*Indemnity to the Agent*), Clause 18 (*Costs and expenses*) and Clause 26.10 (*Lenders indemnity to the Agent or the Security Agent*) shall include the cost of utilising the Agent s or the Security Agent s management time or other resources and will be calculated on the basis of such reasonable daily or hourly rates as the Agent or Security Agent may notify to the Company and the Lenders, and is in addition to any fee paid or payable to the Agent or Security Agent under Clause 13 (*Fees*).

26.16 Deduction from amounts payable by the Agent or the Security Agent

If any Party owes an amount to the Agent or the Security Agent under the Finance Documents the Agent or the Security Agent may, after giving notice to that Party, deduct an amount not exceeding that amount from any payment to that Party which the Agent or the Security Agent would otherwise be obliged to make under the Finance Documents and apply the amount deducted in or towards satisfaction of the amount owed. For the purposes of the Finance Documents that Party shall be regarded as having received any amount so deducted.

26.17 Reliance and engagement letters

Each Finance Party and Secured Party confirms that each of the Arranger, the Agent and the Security Agent has authority to accept on its behalf (and ratifies the acceptance on its behalf of any letters or reports already accepted by the Arranger or Agent or the Security Agent) the terms of any reliance letter or engagement letters relating to the Reports or any reports or letters provided by accountants in connection with the Finance Documents or the transactions contemplated in the Finance Documents and to bind it in respect of those Reports, reports or letters and to sign such letters on its behalf and further confirms that it accepts the terms and qualifications set out in such letters.

26.18 Dutch Parallel Debts

Notwithstanding any provision to the contrary in any Finance Document, in relation to the Dutch Parallel Debts and any security governed by Dutch law:

125

Table of Contents

- (a) the Security Agent shall act in its own name and not as agent of any Finance Party (but always for the benefit of the Finance Parties in accordance with the provisions of the Finance Documents); and
- (b) the rights, powers and authorities vested in the Security Agent pursuant to the Finance Documents are subject to any restrictions imposed by mandatory Dutch law.

26.19 **The Trust Property**

The Security Agent hereby declares that, during the Trust Period, it shall hold the Trust Property as agent and trustee upon trust for the Finance Parties from time to time and the obligations, rights and benefits vested or to be vested in the Security Agent by the Finance Documents or any document entered into pursuant thereto shall (as well after as before enforcement) be performed and (as the case may be) exercised in accordance with the provisions of this Clause 26 (*Role of the Agent, the Security Agent, the Arranger, the Issuing Bank and Others*).

26.20 The Security Agent

The Finance Parties hereby acknowledge that to the extent security is pledged, assigned or mortgaged directly to the Lenders party hereto under the terms of any of the Security Documents and not to the benefit of the Security Agent only, the Security Agent may act as agent (mandataire) for and on behalf of the pledgees, assignees or mortgagees and that all protections and indemnities granted to the Security Agent in this Clause 26 (Role of the Agent, the Security Agent, the Arranger, the Issuing Bank and Others) shall also apply to the Security Agent in such role as pledgees, assignees or mortgagors agent.

For the purposes of Clause 26 (*Role of the Agent, the Security Agent, the Arranger, the Issuing Bank and Others*):

Agent s and Security Agent s Liabilities means all liabilities, costs, fees, charges, claims, damages, losses and expenses (including legal fees and expenses and, in each case, VAT or any similar tax charged or chargeable in respect thereof):

- (a) to which the Agent and the Security Agent becomes subject by reason of it acting as agent or trustee under the Finance Documents; and
- (b) incurred by the Agent and the Security Agent or any attorney, agent, delegate or other person appointed by the Agent or the Security Agent under any Finance Document in relation to or arising out of the taking or holding of any of the security given or created by or pursuant to any of the Finance Documents or in the execution or purported execution of the rights, trust, powers, authorities, discretions and obligations vested in it.

in each case under the Finance Documents or by law, including but not limited to those relating to all actions, proceedings, claims and demands in respect of any matter or thing done or omitted in any way relating to the Finance Documents and all amounts due to the Agent and the Security Agent by way of remuneration for acting as agent or trustee (as the case may be) under any of the Finance Documents other

126

than any such liabilities, costs, fees, charges, damages, losses or expenses arising as a result of the gross negligence or wilful default of such person;

Trust Period means the period ending on the earlier of:

- (a) the last day of the period of 80 years from the date of this Agreement which period (and no other) shall be the applicable perpetuity period; and
- (b) such date as the Security Agent may specify under Clause 26.24 (*Winding-up of trusts*); and **Trust Property** means all or any part of the rights, titles, interest, assets and income that may now or hereafter be mortgaged, charged, assigned or granted or the subject of an Encumbrance in favour of the Security Agent by or pursuant to the Finance Documents and the proceeds of any such security other than all or any rights, titles, interest, assets and income that may now or hereafter be mortgaged, charged, assigned or granted or be the subject of Security in favour of the Lenders represented by the Security Agent in its capacity as agent.

26.21 Security Agent s powers

The Security Agent shall have all the powers and discretions conferred upon trustees by the Trustee Act 1925 and the Trustee Act 2000 (to the extent not inconsistent herewith) and upon the Security Agent by this Agreement and the other Finance Documents.

26.22 No Independent Power

Each of the Finance Parties (other than the Security Agent) hereby agrees that they shall not have any independent power to enforce, or have recourse to, any of the Security Documents or to exercise any rights or powers arising under the Security Documents except through the Security Agent including when the Security Agent is acting as agent in accordance with Clause 26.20 (*The Security Agent*).

26.23 Disposals of Charged Property

The Security Agent may:

- (a) upon a disposal of any property the subject of any of the Transaction Security by any receiver, or by any of the Obligors or the Acquiror where the Security Agent has consented to the disposal, to any third party, release such property from such Security; and
- (b) perform all of its obligations under the Finance Documents, notwithstanding anything contained in this Agreement.

26.24 Winding-up of trusts

If the Security Agent, with the approval of the Lenders, shall determine that all of the obligations under the Finance Documents and all other obligations the discharge of which is secured by any of the Transaction Security Documents have been fully and finally discharged and none of the Finance Parties is under any commitment.

127

Table of Contents

obligation or liability (whether actual or contingent) to make advances or provide other financial accommodation to the Borrower under or pursuant to this Agreement or any other Finance Document and no Letter of Credit is outstanding, the trusts herein set out shall be wound up and the Security Agent shall release, without recourse or warranty, all of the Transaction Security then held by it, whereupon each of the Finance Parties and the Obligors and the Acquiror shall be released from its obligations hereunder (save for those which arose prior to such winding-up).

26.25 German Security

- (a) The Security Agent shall hold (to the extent possible under the applicable law) on trust (*treuhänderisch*) or administer the benefit of the Transaction Security Documents for itself and the other Finance Parties and will apply all payments and other benefits received by it under the Transaction Security Documents in accordance with the provisions of the Finance Documents.
- (b) Each Finance Party (other than the Security Agent) hereby authorises the Security Agent to act as its representative (*Stellvertreter*) with regard to any pledge or other creation of any accessory security right granted to such Finance Party pursuant to a Transaction Security Document and to act and execute as its representative with regard to any amendments of, accessions to, release of and any similar dealings with regard to any Transaction Security Document which creates a pledge or other accessory security right.
- (c) Each Finance Party (other than the Security Agent) hereby authorises the Security Agent to act as its representative (*Stellvertreter*) with regard to the release or confirmation of release of any pledge or other accessory security right (*akzessorische Sicherheit*) of such Finance Party pursuant to a Transaction Security Document.
- (d) The Security Agent shall be released from the restrictions of Section 181 of the German Civil Code (*BGB Bürgerliches Gesetzbuch*) and shall be authorised to delegate its power of attorney granted hereunder.
- (e) The Security Agent shall:
 - (i) hold each Transaction Security granted under any Transaction Security Document governed by German law (a **German Encumbrance**), if any, which is transferred or assigned by way of security (Sicherungsübereignung / Sicherungsabtretung) or otherwise granted under a non-accessory security right (nicht akzessorische Sicherheit) as trustee (Treuhänder) for the benefit of the Finance Parties; and
 - (ii) administer in the name and on behalf of the Finance Parties each German Encumbrance which is pledged (*Verpfändung*) or otherwise transferred under an accessory security right (*akzessorische Sicherheit*) to the Security Agent and other Finance Parties.
- (f) Each Finance Party hereby irrevocably authorises the Security Agent to act on its behalf and, if required under applicable law or if otherwise appropriate, in its name in connection with the preparation, execution and delivery of each

128

Transaction Security Document governed by German law (a **German Security Document**) and the perfection and monitoring of each German Encumbrance, including but not limited to, any share pledge agreement with respect to shares in a German company in notarial form, as well as any other pledge, mortgage, assignment or transfer of title for security purposes. The Security Agent shall be authorised to make all statements necessary or appropriate in this connection.

(g) Each Finance Party hereby irrevocably authorises the Security Agent to act on its behalf and, if required under applicable law, or if otherwise appropriate, in its name and on its behalf in connection with the preparation, execution and delivery of each Transaction Security Document governed by German law and the security interests thereunder and the perfection and monitoring thereof.

26.26 Security Agent and Agent as Finance Parties

Each of the Security Agent and the Agent is, at the date of this Agreement, a Finance Party and shall be entitled, notwithstanding that it is acting as agent or trustee, to take, or refrain from taking, any action which it would be entitled so to take in its capacity as a Finance Party if it were not acting as agent or trustee and shall not be precluded by virtue of its position as Finance Party from exercising any of its discretions, power and duties as agent or trustee.

26.27 Application of receipts and recoveries

To the extent that the Security Agent receives or recovers monies to be applied in discharging any obligation (whether actual or contingent, present or future) of any Obligor or, at the procurement of the Company, the Acquiror under any Finance Document, such monies shall (without prejudice to the respective rights of the Agent or the Security Agent pursuant to any Finance Document to credit any monies received by it to any suspense account) be applied as between the Finance Parties:

- (a) first, in or towards payment of all Agent s and Security Agent s Liabilities;
- (b) secondly, in or towards payment of any amounts which are due and payable at such time by any Obligor or, at the procurement of the Company, the Acquiror to any Finance Party under any of the Finance Documents pro rata to the respective entitlements of such Finance Parties;
- (c) thirdly, if the Obligors and the Acquiror are under no further actual or contingent liability hereunder or under any other Finance Document, in payment to any person to whom the Security Agent is obliged to pay in priority to the Obligor or, as the case may be, the Acquiror otherwise entitled thereto, to the extent it is so obliged; and
- (d) fourthly, thereafter, in payment to the Obligor or, as the case may be the Acquiror entitled thereto.

26.28 Additional Applications by Security Agent

129

Table of Contents

The fact that the Security Agent may make payment pursuant to Clause 26.27(b) (*Application of Receipts and Recoveries*) or may determine that the Obligors and the Acquiror are under no further actual or contingent liability under the Finance Documents and make a payment under sub-clauses 26.27(c) or 26.27(d) thereof, will not thereafter prevent the Security Agent from applying any further monies, or any credit balance on any account into which monies are paid in contemplation of or in connection with the arrangements set out herein, in the order set out in Clause 26.27 (*Application of Receipts and Recoveries*).

26.29 Deductions by Security Agent

The Security Agent shall be entitled to make the deductions and withholdings (on account of taxes or otherwise) from payments hereunder which it is required by any applicable law to make, and to pay all taxes which may be assessed against it in respect of any of the Trust Property, in respect of anything done by it in its capacity as Security Agent or otherwise by virtue of its capacity as Security Agent. Each of the Obligors agrees that the obligations under the Finance Documents shall only be discharged by virtue of receipt or recovery by the Security Agent of monies, or of payments made by the Security Agent hereunder, to the extent that the ultimate recipient actually receives monies from the Security Agent hereunder.

26.30 Receipts by Obligors

If any of the Obligors or the Acquiror receives any sum from any person which, pursuant to the Finance Documents, should have been paid to the Security Agent, such Obligor shall and the Company shall procure that the Acquiror shall hold such sum on trust for the Finance Parties, and shall forthwith be paid over to the Security Agent for application in accordance with this Clause 26.

27. CONDUCT OF BUSINESS BY THE FINANCE PARTIES

No provision of this Agreement will:

- (a) interfere with the right of any Finance Party to arrange its affairs (tax or otherwise) in whatever manner it thinks fit;
- (b) oblige any Finance Party to investigate or claim any credit, relief, remission or repayment available to it or the extent, order and manner of any claim; or
- (c) oblige any Finance Party to disclose any information relating to its affairs (tax or otherwise) or any computations in respect of Tax.

28. SHARING AMONG THE FINANCE PARTIES

28.1 **Payments to Finance Parties**

If a Finance Party (a **Recovering Finance Party**) receives or recovers any amount from an Obligor or the Acquiror other than in accordance with Clause 29 (*Payment Mechanics*) and applies that amount to a payment due under the Finance Documents then:

130

Table of Contents

- (a) the Recovering Finance Party shall, within three Business Days, notify details of the receipt or recovery, to the Agent;
- (b) the Agent shall determine whether the receipt or recovery is in excess of the amount the Recovering Finance Party would have been paid had the receipt or recovery been received or made by the Agent and distributed in accordance with Clause 29 (*Payment Mechanics*), without taking account of any Tax which would be imposed on the Agent in relation to the receipt, recovery or distribution; and
- (c) the Recovering Finance Party shall, within three Business Days of demand by the Agent, pay to the Agent an amount (the **Sharing Payment**) equal to such receipt or recovery less any amount which the Agent determines may be retained by the Recovering Finance Party as its share of any payment to be made, in accordance with Clause 29.5 (*Partial payments*).

28.2 Redistribution of payments

The Agent shall treat the Sharing Payment as if it had been paid by the relevant Obligor or, as the case may be, the Acquiror and distribute it between the Finance Parties (other than the Recovering Finance Party) in accordance with Clause 29.5 (*Partial payments*).

28.3 Recovering Finance Party s rights

- (a) On a distribution by the Agent under Clause 28.2 (*Redistribution of payments*), the Recovering Finance Party will be subrogated to the rights of the Finance Parties which have shared in the redistribution.
- (b) If and to the extent that the Recovering Finance Party is not able to rely on its rights under paragraph (a) above, the relevant Obligor or, as the case may be, at the procurement of the Company, the Acquiror shall be liable to the Recovering Finance Party for a debt equal to the Sharing Payment which is immediately due and payable.

28.4 Reversal of redistribution

If any part of the Sharing Payment received or recovered by a Recovering Finance Party becomes repayable and is repaid by that Recovering Finance Party, then:

- (a) each Finance Party which has received a share of the relevant Sharing Payment pursuant to Clause 28.2 (*Redistribution of payments*) shall, upon request of the Agent, pay to the Agent for account of that Recovering Finance Party an amount equal to the appropriate part of its share of the Sharing Payment (together with an amount as is necessary to reimburse that Recovering Finance Party for its proportion of any interest on the Sharing Payment which that Recovering Finance Party is required to pay); and
- (b) that Recovering Finance Party s rights of subrogation in respect of any reimbursement shall be cancelled and the relevant Obligor or, as the case may

131

Table of Contents

be, at the procurement of the Company, the Acquiror will be liable to the reimbursing Finance Party for the amount so reimbursed.

28.5 Exceptions

- (a) This Clause 28 shall not apply to the extent that the Recovering Finance Party would not, after making any payment pursuant to this Clause, have a valid and enforceable claim against the relevant Obligor or, as the case may be, the Acquiror.
- (b) A Recovering Finance Party is not obliged to share with any other Finance Party any amount which the Recovering Finance Party has received or recovered as a result of taking legal or arbitration proceedings, if:
 - (i) it notified that other Finance Party of the legal or arbitration proceedings; and
 - (ii) that other Finance Party had an opportunity to participate in those legal or arbitration proceedings but did not do so as soon as reasonably practicable having received notice and did not take separate legal or arbitration proceedings.

132

SECTION 11 ADMINISTRATION

29. PAYMENT MECHANICS

29.1 Payments to the Agent

- (a) On each date on which an Obligor, at the procurement of the Company, the Acquiror or a Lender is required to make a payment under a Finance Document, that Obligor, at the procurement of the Company, the Acquiror or Lender shall make the same available to the Agent (unless a contrary indication appears in a Finance Document) for value on the due date at the time and in such funds specified by the Agent as being customary at the time for settlement of transactions in the relevant currency in the place of payment.
- (b) Payment shall be made to such account in the principal financial centre of the country of that currency (or, in relation to Euro, in a principal financial centre in a Participating Member State or London) with such bank as the Agent specifies.

29.2 Distributions by the Agent

Each payment received by the Agent under the Finance Documents for another Party shall, subject to Clause 29.3 (*Distributions to an Obligor*) and Clause 29.4 (*Clawback*) be made available by the Agent as soon as practicable after receipt to the Party entitled to receive payment in accordance with this Agreement (in the case of a Lender, for the account of its Facility Office), to such account as that Party may notify to the Agent by not less than five Business Days notice with a bank in the principal financial centre of the country of that currency (or, in relation to Euro, in the principal financial centre of a Participating Member State or London).

29.3 **Distributions to an Obligor**

The Agent may (with the consent of the Obligor or, as the case may be, at the procurement of the Company, the Acquiror or in accordance with Clause 30 (*Set-off*)) apply any amount received by it for that Obligor or, as the case may be, the Acquiror, in or towards payment (on the date and in the currency and funds of receipt) of any amount due from that Obligor or, as the case may be, at the procurement of the Company, the Acquiror under the Finance Documents or in or towards purchase of any amount of any currency to be so applied.

29.4 Clawback

- (a) Where a sum is to be paid to the Agent under the Finance Documents for another Party, the Agent is not obliged to pay that sum to that other Party (or to enter into or perform any related exchange contract) until it has been able to establish to its satisfaction that it has actually received that sum.
- (b) If the Agent pays an amount to another Party and it proves to be the case that the Agent had not actually received that amount, then the Party to whom that amount (or the proceeds of any related exchange contract) was paid by the

133

Table of Contents

Agent shall on demand refund the same to the Agent together with interest on that amount from the date of payment to the date of receipt by the Agent, calculated by the Agent to reflect its cost of funds.

29.5 Partial payments

- (a) If the Agent receives a payment that is insufficient to discharge all the amounts then due and payable by an Obligor or, at the procurement of the Company, the Acquiror under the Finance Documents, the Agent shall apply that payment towards the obligations of that Obligor or, as the case may be, the Acquiror under the Finance Documents in the following order:
 - (i) **first**, in or towards payment pro rata of any unpaid fees, costs and expenses of the Agent, the Issuing Bank and the Security Agent under the Finance Documents;
 - (ii) **secondly**, in or towards payment pro rata of any accrued interest, fee or commission due but unpaid under the Finance Documents;
 - (iii) **thirdly**, in or towards payment pro rata of any principal due but unpaid under the Finance Documents and any amount due but unpaid under Clause 7.2 (*Claims under the Letter of Credit*) or Clause 7.3 (*Indemnities*); and
 - (iv) **fourthly**, in or towards payment pro rata of any other sum due but unpaid under the Finance Documents.
- (b) The Agent shall, if so directed by the Majority Lenders, vary the order set out in paragraphs (a)(ii) to (iv) above.
- (c) Paragraphs (a) and (b) above will override any appropriation made by an Obligor or, at the procurement of the Company, the Acquiror.

29.6 No set-off by Obligors

All payments to be made by an Obligor or, at the procurement of the Company, the Acquiror under the Finance Documents shall be calculated and be made without (and free and clear of any deduction for) set-off or counterclaim.

29.7 **Business Days**

- (a) Any payment which is due to be made on a day that is not a Business Day shall be made on the next Business Day in the same calendar month (if there is one) or the preceding Business Day (if there is not).
- (b) During any extension of the due date for payment of any principal or Unpaid Sum under this Agreement interest is payable on the principal or Unpaid Sum at the rate payable on the original due date.

29.8 Currency of account

134

Table of Contents

- (a) Subject to paragraph (b) below, Euro is the currency of account and payment for any sum due from an Obligor or, at the procurement of the Company, the Acquiror under any Finance Document, except that each payment in respect of costs, expenses or Taxes shall be made in the currency in which the costs, expenses or Taxes are incurred.
- (b) Any amount expressed to be payable in a currency other than Euro shall be paid in that other currency.

29.9 Change of currency

- (a) Unless otherwise prohibited by law, if more than one currency or currency unit are at the same time recognised by the central bank of any country as the lawful currency of that country, then:
 - (i) any reference in the Finance Documents to, and any obligations arising under the Finance Documents in, the currency of that country shall be translated into, or paid in, the currency or currency unit of that country designated by the Agent (after consultation with the Company); and
 - (ii) any translation from one currency or currency unit to another shall be at the official rate of exchange recognised by the central bank for the conversion of that currency or currency unit into the other, rounded up or down by the Agent (acting reasonably).
- (b) If a change in any currency of a country occurs, this Agreement will, to the extent the Agent (acting reasonably and after consultation with the Company) specifies to be necessary, be amended to comply with any generally accepted conventions and market practice in the Relevant Interbank Market and otherwise to reflect the change in currency.

29.10 Disruption to Payment Systems etc.

If either the Agent determines (in its discretion) that a Disruption Event has occurred or the Agent is notified by the Company that a Disruption Event has occurred:

- (a) the Agent may, and shall if requested to do so by the Company, consult with the Company with a view to agreeing with the Company such changes to the operation or administration of the Facility as the Agent may deem necessary in the circumstances;
- (b) the Agent shall not be obliged to consult with the Company in relation to any changes mentioned in paragraph (a) if, in its opinion, it is not practicable to do so in the circumstances and, in any event, shall have no obligation to agree to such changes;
- (c) the Agent may consult with the Finance Parties in relation to any changes mentioned in paragraph (a) but shall not be obliged to do so if, in its opinion, it is not practicable to do so in the circumstances;
- (d) any such changes agreed upon by the Agent and the Company shall (whether or not it is finally determined that a Disruption Event has occurred) be binding

135

Table of Contents

upon the Parties as an amendment to (or, as the case may be, waiver of) the terms of the Finance Documents notwithstanding the provisions of Clause 35 (*Amendments and Waivers*);

- (e) the Agent shall not be liable for any damages, costs or losses whatsoever (including, without limitation for negligence, gross negligence or any other category of liability whatsoever but not including any claim based on the fraud of the Agent) arising as a result of its taking, or failing to take, any actions pursuant to or in connection with this Clause 29.10; and
- (f) the Agent shall notify the Finance Parties of all changes agreed pursuant to paragraph (d) above.

30. **SET-OFF**

A Finance Party may set off any matured obligation due from an Obligor or, at the procurement of the Company, the Acquiror under the Finance Documents (to the extent beneficially owned by that Finance Party) against any matured obligation owed by that Finance Party to that Obligor or, as the case may be, at the procurement of the Company, the Acquiror, regardless of the place of payment, booking branch or currency of either obligation. If the obligations are in different currencies, the Finance Party may convert either obligation at a market rate of exchange in its usual course of business for the purpose of the set-off.

31. NOTICES

31.1 Communications in writing

Any communication to be made under or in connection with the Finance Documents shall be made in writing and, unless otherwise stated, may be made by fax or letter or e-mail.

31.2 Addresses

The address and fax number (and the department or officer, if any, for whose attention the communication is to be made) of each Party for any communication or document to be made or delivered under or in connection with the Finance Documents is:

- (a) in the case of the Company or the Borrower, that identified with its name below;
- (b) in the case of each Lender, the Issuing Bank or any other Obligor, that notified in writing to the Agent on or prior to the date on which it becomes a Party; and
- (c) in the case of the Agent or the Security Agent, that identified with its name below, or any substitute address or fax number or department or officer as the Party may notify to the Agent (or the Agent may notify to the other Parties, if a change is made by the Agent) by not less than five Business Days notice.

31.3 **Delivery**

136

Table of Contents

- (a) Any communication or document made or delivered by one person to another under or in connection with the Finance Documents will only be effective:
 - (i) if by way of fax or e-mail, when received in legible form; or
 - (ii) if by way of letter, when it has been left at the relevant address or five Business Days after being deposited in the post postage prepaid in an envelope addressed to it at that address; and, if a particular department or officer is specified as part of its address details provided under Clause

31.2 (Addresses), if addressed to that department or officer.

- (b) Any communication or document to be made or delivered to the Agent or the Security Agent will be effective only when actually received by the Agent or the Security Agent and then only if it is expressly marked for the attention of the department or officer identified with the Agent s or the Security Agent s signature below (or any substitute department or officer as the Agent or the Security Agent shall specify for this purpose).
- (c) All notices from or to an Obligor or, at the procurement of the Company, the Acquiror shall be sent through the Agent.
- (d) Any communication or document made or delivered to the Company in accordance with this Clause will be deemed to have been made or delivered to each of the Obligors and, at the procurement of the Company, the Acquiror.

31.4 Notification of address and fax number

Promptly upon receipt of notification of an address or fax number or change of address or fax number pursuant to Clause 31.2 (*Addresses*) or changing its own address or fax number, the Agent shall notify the other Parties.

31.5 Electronic communication

- (a) Any communication to be made between the Agent or the Security Agent and a Lender under or in connection with the Finance Documents may be made by electronic mail or other electronic means, if the Agent, the Security Agent and the relevant Lender:
 - (i) agree that, unless and until notified to the contrary, this is to be an accepted form of communication;
 - (ii) notify each other in writing of their electronic mail address and/or any other information required to enable the sending and receipt of information by that means; and
 - (iii) notify each other of any change to their address or any other such information supplied by them.

137

Table of Contents

(b) Any electronic communication made between the Agent and a Lender or the Security Agent will be effective only when actually received in readable form and in the case of any electronic communication made by a Lender to the Agent or the Security Agent only if it is addressed in such a manner as the Agent or the Security Agent shall specify for this purpose.

31.6 English language

- (a) Any notice given under or in connection with any Finance Document must be in English.
- (b) All other documents provided under or in connection with any Finance Document must be:
 - (i) in English; or
 - (ii) if not in English, and if so required by the Agent, accompanied by a certified English translation and, in this case, the English translation will prevail unless the document is a constitutional, statutory or other official document.

32. CALCULATIONS AND CERTIFICATES

32.1 Accounts

In any litigation or arbitration proceedings arising out of or in connection with a Finance Document, the entries made in the accounts maintained by a Finance Party are *prima facie* evidence of the matters to which they relate.

32.2 Certificates and Determinations

Any certification or determination by a Finance Party of a rate or amount under any Finance Document is, in the absence of manifest error, prima facie evidence of the matters to which it relates.

32.3 **Day count convention**

Any interest, commission or fee accruing under a Finance Document will accrue from day to day and is calculated on the basis of the actual number of days elapsed and a year of 360 days or, in any case where the practice in the Relevant Interbank Market differs, in accordance with that market practice.

33. PARTIAL INVALIDITY

If, at any time, any provision of the Finance Documents is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will in any way be affected or impaired.

138

34. REMEDIES AND WAIVERS

No failure to exercise, nor any delay in exercising, on the part of any Finance Party or Secured Party, any right or remedy under the Finance Documents shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in this Agreement are cumulative and not exclusive of any rights or remedies provided by law.

35. AMENDMENTS AND WAIVERS

35.1 Required consents

- (a) Subject to Clause 35.2 (*Exceptions*) any term of the Finance Documents may be amended or waived only with the consent of the Majority Lenders and the Company and any such amendment or waiver will be binding on all Parties.
- (b) The Agent may effect, on behalf of any Finance Party, any amendment or waiver permitted by this Clause.
- (c) Each Obligor agrees to any such amendment or waiver permitted under this Clause 35 which is agreed to by the Company. This includes any amendment or waiver which would, but for this paragraph (c) require the consent of all the Guarantors.

35.2 Exceptions

- (a) An amendment or waiver that has the effect of changing or which relates to:
 - (i) the definition of **Majority Lenders** in Clause 1.1 (*Definitions*);
 - (ii) an extension to the date of payment of any amount under the Finance Documents;
 - (iii) a reduction in the Margin or a reduction in the amount of any payment of principal, interest, fees or commission payable;
 - (iv) an increase in or an extension of any Commitment or the Total Commitments;
 - (v) a change to the Borrower or the Guarantors except in accordance with Clause 25 (*Changes to the Lenders and Obligors*);
 - (vi) any provision which expressly requires the consent of all the Lenders;
 - (vii) Clause 2.2 (Finance Parties rights and obligations), Clause 25 (Changes to the Lenders and Obligors) or this Clause 35;
 - (viii) the nature or scope of the Charged Property or the manner in which the proceeds of enforcement of the Transaction Security are distributed (except insofar as it relates to a sale or disposal of an asset which is the subject of the Transaction Security where such sale or disposal is

139

Table of Contents

expressly permitted under this Agreement or any other Finance Document); or

(ix) the release of any Transaction Security unless permitted under this Agreement or any other Finance Document or relating to a sale or disposal of an asset which is the subject of the Transaction Security where such sale or disposal is expressly permitted under this Agreement or any other Finance Document,

shall not be made without the prior consent of all the Lenders.

(b) An amendment or waiver which relates to the rights or obligations of the Agent, the Issuing Bank, the Security Agent or the Arranger may not be effected without the consent of the Agent, the Issuing Bank, the Security Agent or the Arranger.

36. **CONFIDENTIALITY**

The Parties will keep confidential the Finance Documents and all information which they acquire under or in connection with the Finance Documents save that such information may be disclosed:

- (a) if so required by law or regulation, or if requested by any governmental, banking, taxation or other regulatory authority with jurisdiction over any Finance Party or any Affiliate of any Finance Party;
- (b) if it comes into the public domain (other than as a result of a breach of this Clause 36);
- (c) to auditors, professional advisers or rating agencies who are required in the course of their duties or role to receive such information;
- (d) in connection with any legal or arbitration proceedings;
- (e) to the extent permitted and subject to the provisions of Clause 25.8 (Disclosure of information); or
- (f) with the consent of the Company.

The provisions of this Clause 36 shall supersede any undertakings with respect to confidentiality previously given by any Finance Party in favour of any Obligor or the Acquiror.

37. **COUNTERPARTS**

Each Finance Document may be executed in any number of counterparts, and this has the same effect as if the signatures on the counterparts were on a single copy of the Finance Document.

140

38. PATRIOT ACT

Each Lender hereby notifies each Obligor that pursuant to the requirements of the Patriot Act, such Lender is required to obtain, verify and record information that identifies each Obligor and such information includes the name and address of such Obligor and other information that will allow such Lender to identify such Obligor in accordance with the Patriot Act. Each Obligor shall, and shall cause each of its Subsidiaries to provide such information and take such actions as are reasonably requested by the Agent or any other Finance Party in order to assist the Agent and the other Finance Parties in maintaining compliance with the Patriot Act.

141

SECTION 12 GOVERNING LAW AND ENFORCEMENT

39. **GOVERNING LAW**

This Agreement is governed by English law.

40. **ENFORCEMENT**

40.1 Jurisdiction

- (a) The courts of England have exclusive jurisdiction to settle any dispute arising out of or in connection with this Agreement (including a dispute regarding the existence, validity or termination of this Agreement) (a **Dispute**).
- (b) The Parties agree that the courts of England are the most appropriate and convenient courts to settle Disputes and accordingly no Party will argue to the contrary.
- (c) This Clause 40.1 is for the benefit of the Finance Parties and the Secured Parties only. As a result, no Finance Party or Secured Party shall be prevented from taking proceedings relating to a Dispute in any other courts with jurisdiction. To the extent allowed by law, the Finance Parties and Secured Parties may take concurrent proceedings in any number of jurisdictions.

40.2 Service of process

Without prejudice to any other mode of service allowed under any relevant law, each Obligor (other than an Obligor incorporated in England and Wales):

- (a) irrevocably appoints Crown NorthCorp Limited as its agent for service of process in relation to any proceedings before the English courts in connection with any Finance Document and Crown NorthCorp Limited by its execution of this Agreement, accepts that appointment; and
- (b) agrees that failure by a process agent to notify the relevant Obligor of the process will not invalidate the proceedings concerned.
- (c) If any person appointed as an agent for service of process is unable for any reason to act as agent for service of process, the Company (on behalf of all the Obligors) must immediately (and in any event within 14 days of such event taking place) appoint another agent on terms acceptable to the Agent. Failing this, the Agent may appoint another agent for this purpose.
- (d) Crown Westfalen LLC and Crown NorthCorp Inc. each expressly agrees and consents to the provisions of this Clause 40 and Clause 39 (*Governing law*).

This Agreement has been entered into on the date stated at the beginning of this Agreement.

142

SCHEDULE 1 THE ORIGINAL PARTIES Part I The Obligors

Name of the Borrower Registration number (or equivalent, if any)

Crown Westfalen LLC 4172696

Name of the Guarantors Registration number (or equivalent, if any)

Crown NorthCorp Inc. 2298568
Crown Northcorp Limited 03431892
Crown Asset Management Limited 03802849
Crown Mortgage Management Limited 00912411
Regal Mortgages LLP 0C300246

Table of Contents

Part II The Original Lender

Name of Original Lender

Credit Suisse, London Branch

25,000,000

144

SCHEDULE 2 CONDITIONS PRECEDENT Part I

Conditions Precedent to First Utilisation

1. **Obligors**

- (a) A copy of the constitutional documents (including, in relation to the Acquiror, a recent extract from the Dutch trade register (*handelsregister*) relating to it) of each of the Company, the Borrower, each of the other Obligors and of each member of the Group that is not an Obligor, whose shares are charged or pledged under the Transaction Security Documents.
- (b) A copy of a resolution of the board of directors of each of the Company, the Borrower and each of the other Obligors:
 - (i) approving the terms of, and the transactions contemplated by, the Finance Documents to which it is a party and resolving that it execute the Finance Documents to which it is a party;
 - (ii) authorising a specified person or persons to execute the Finance Documents to which it is a party on its behalf;
 - (iii) authorising a specified person or persons, on its behalf, to sign and/or despatch all documents and notices (including, if relevant, any Utilisation Request) to be signed and/or despatched by it under or in connection with the Finance Documents to which it is a party;
 - (iv) in the case of an Obligor other than the Company, authorising the Company to act as its agent in connection with the Finance Documents; and
 - (v) in respect of the Acquiror:
 - (A) a copy of a resolution of its board of supervisory directors (if any) approving its execution and the terms of, and the transactions contemplated by, the Finance Documents to which it is a party; and
 - (B) if it is an Guarantor or if required by law or its constitutional documents, a copy of a resolution of its general meeting of shareholders approving its execution and the terms of, and the transactions contemplated by, the Finance Documents to which it is a party.
- (c) A specimen of the signature of each person authorised by the resolution referred to in paragraph (b) above.
- (d) A copy of each resolution and other formalities (if any) necessary to ensure that the constitutional documents of all relevant members of the Group do not restrict the transfer of any pledged/charged shares.

145

- (e) A certificate of the Borrower and each other Obligor (each signed by a director) confirming that borrowing or guaranteeing, as appropriate, the Total Commitments would not cause any borrowing, guaranteeing or similar limit binding on the Borrower or such Obligor respectively to be exceeded.
- (f) A certificate of an authorised signatory of each of the Company and each relevant Obligor certifying that each copy document relating to it specified in this Part I of Schedule 2 is correct, complete and in full force and effect as at a date no earlier than the date of the Facility Agreement.
- (g) A solvency certificate issued by each of the Company and the Borrower and addressed to the Agent confirming the solvency of the Company and the Borrower, respectively, immediately before and after entry by each into any Finance Document to which it is a party.
- (h) A copy of a good standing certificate (including verification of tax status) with respect to the Company and the Borrower, issued as of a recent date of the date of the Facility Agreement and any Transaction Security Document to which it is a party by the Secretary of State or other appropriate official of the jurisdiction of incorporation or organisation of the Company and the Borrower and the equivalent, if applicable, in other relevant jurisdictions.

2. Acquisition Documents

- (a) A copy of the executed Acquisition Agreement and all other Acquisition Documents.
- (b) Evidence that all conditions in the Acquisition Agreement have or will be been satisfied without waiver or amendment.

3. Finance Documents

- (a) The Facility Agreement executed by each of the Company, the Borrower and other Obligors.
- (b) The Fee Letters executed by each of the Company and the Borrower.
- (c) The Engagement Letter executed by each of the Company and the Borrower.
- (d) At least two originals of the Transaction Security Documents executed by the relevant Obligor party thereto as specified below:

Name of Obligor Transaction Security Document

Crown NorthCorp Inc. US Security Agreement

English share charge

Crown NorthCorp Limited English Law Debenture

Swedish share charge

146

Name of Obligor Transaction Security Document

Crown Asset Management Limited English Law Debenture

Crown Mortgage Management English Law Debenture

Limited

Regal Mortgages LLP English Law Debenture

Crown Westfalen LLC US Security Agreement

- (e) A copy of all notices required to be sent under the Transaction Security Documents executed by the relevant Obligor duly acknowledged by the addressees (to the extent such acknowledgement is required on or before the Utilisation Date under the terms of the relevant Transaction Security Documents) and of all third party consents required in connection with the creation, perfection or registration of the Transaction Security Documents.
- (f) A copy of all share certificates, transfers and stock transfer forms or equivalent duly executed by the relevant Obligor in blank in relation to the assets subject to or expressed to be subject to the Transaction Security and other documents of title which are required to be provided under the Transaction Security Documents.

4. Legal opinions

The following legal opinions, each addressed to the Agent, the Security Agent and any person who becomes a Lender:

- (a) legal opinions of Shearman & Sterling (London) LLP and Shearman & Sterling LLP, legal adviser to the Agent and the Arranger as to matters of English law and German law;
- (b) a legal opinion of De Brauw Blackstone Westbroek, legal adviser to the Agent and the Arranger as to matters of Dutch law:
- (c) a legal opinion of Gernandt & Danielsson Advokatbyrå KB, legal adviser to the Agent and the Arranger as to matters of Swedish law; and
- (d) a legal opinion of Dechert LLP, legal advisor to the Company and the Borrower as to matters of US law.

5. Other documents and evidence

- (a) Evidence that any process agent referred to in Clause 40.2 (*Service of process*) of the Facility Agreement, if not an Obligor, has accepted its appointment.
- (b) Evidence that the German Federal Agency for Supervision of Financial Services does not and will not raise objections to the Acquisition nor the entry into and performance of the Acquisition Documents and/or the Finance

147

Table of Contents

Documents nor the validity and enforceability of the Acquisition Documents and/or the Finance Documents.

- (c) Evidence that any other governmental, shareholder and third party Authorisation required in connection with the Acquisition Documents and the validity and enforceability of the Acquisition Documents have been obtained and are in full force and effect and all applicable waiting periods have expired or been terminated.
- (d) A copy of any other governmental, shareholder and third party Authorisation or other document, opinion or assurance which the Agent considers to be necessary or desirable in connection with the entry into and performance of the transactions contemplated by any Finance Document or for the validity and enforceability of any Finance Document (including, without limitation, in relation to a joint-venture arrangement).
- (e) The Structure Memorandum which shows the Group assuming the Closing Date has occurred and Transaction 1, Transaction 2 and Transaction 3 (as each are defined in the BT Treasury Letter) have been completed and shows such other matters as are referred to in Clause 20.23 (*Structure Memorandum*) of the Facility Agreement.
- (f) Evidence of completion of due diligence to the satisfaction of the Arranger including financial due diligence, review of the Target s balance sheet and liquidity, finalisation of the Structure Memorandum (including the terms of the proposed sale of the Target s non-performing loan portfolio) and credit committee and other internal approvals of the Arranger.
- (g) Originals of the Reports (including the Structure Memorandum) and reliance letters (other than the reliance letters described in Clause 23.38 (*Conditions Precedent in relation to Reliance Letters*)) relating thereto.
- (h) A copy, certified by a director of the Company to be a true copy, of the Original Financial Statements of the Company.
- (i) All relevant documentation for the Permitted Reorganisations, including informal approval(s) (and application(s)), and liquidity analysis.
- (j) The results of know your customer and other checks in relation to the Company and the Borrower under the FSA Money Laundering Regulations, the Patriot Act and any other applicable laws and regulations being satisfactory in all respects to the Arranger and the Agent.
- (k) Evidence that US\$2,500,000 has been deposited into the Interest Reserve Account in respect of the arrangement fee and in respect of interest and that the fees, costs and expenses then due from the Company pursuant to Clause 13 (*Fees*) and Clause 18 (*Costs and expenses*) of the Facility Agreement have been paid or will be paid by the earlier of the Closing Date and the last day of the Availability Period.

148

Table of Contents

- (l) All relevant documentation relating to Rooftop Holdings Limited, including current constitutional documents, shareholder agreements and other relevant arrangements.
- (m) Copies of financing statements (Form UCC-1) in appropriate form for filing under the Uniform Commercial Code of each applicable jurisdiction as may be necessary to perfect the Transaction Security purported to be created by each Transaction Security Document entered into by the Company and the Borrower or in respect of shares or indebtedness of any such company.
- (n) Copies of lien searches, each of a recent date listing all effective financing statements that name the Company or the Borrower, or a division or operating unit of such person, as debtor and that are filed in the jurisdiction referred to in paragraph (m) above, together with appropriate termination statements executed by the secured lender thereunder shall be delivered to the Agent (except in respect of financing statements related to Security permitted to subsist under the Facility Agreement).
- (o) Satisfactory searches and filings in the United Kingdom, Sweden, Germany and the Netherlands.
- (p) The Base Case Model.

Part II

Conditions precedent required to be delivered by an Additional Guarantor

- 1. An Accession Letter executed by the Additional Guarantor and the Company.
- 2. A copy of the constitutional documents of the Additional Guarantor (including, in relation to an Additional Guarantor incoporated under Dutch law, a recent extract from the Dutch trade register (*handelsregister*)) relating to it.
- 3. A copy of a resolution of the board or, if applicable, a committee of the board of directors of the Additional Guarantor:
 - (a) approving the terms of, and the transactions contemplated by, the Accession Letter and the Finance Documents and resolving that it execute, deliver and perform the Accession Letter and any other Finance Document to which it is party;
 - (b) authorising a specified person or persons to execute the Accession Letter and other Finance Documents on its behalf;
 - (c) authorising a specified person or persons, on its behalf, to sign and/or despatch all other documents and notices to be signed and/or despatched by it under or in connection with the Finance Documents to which it is a party;
 - (d) authorising the Company to act as its agent in connection with the Finance Documents; and 149

Table of Contents

- (e) in relation to an Additional Guarantor incorporated under Dutch law:
 - (i) a copy of a resolution of its board of supervisory directors (if any) approving its execution and the terms of, and the transactions contemplated by, the Finance Documents to which it is a party;
 - (ii) a copy of a resolution of its general meeting of shareholders approving its execution and the terms of, and the transactions contemplated by, the Finance Documents to which it is a party; and
 - (iii) if it is required by law or any arrangement binding on it to obtain works council advice in respect of its or any other person s entry into the Finance Documents, a copy of a positive advice from its (central) works council (and, if such advice is not unconditional, confirmation from the Company that (A) the conditions set by the works council are and will be complied with and (B) such compliance does and will not have a Material Adverse Effect.
- 4. If applicable, a copy of a resolution of the board of directors of the Additional Guarantor, establishing the committee referred to in paragraph 3 above.
- 5. A specimen of the signature of each person authorised by the resolution referred to in paragraph 3 above.
- 6. A copy of a resolution signed by all the holders of the issued shares of the Additional Guarantor, approving the terms of, and the transactions contemplated by, the Finance Documents to which the Additional Guarantor is a party.
- 7. A copy of a resolution of the board of directors of each corporate shareholder of each Additional Guarantor approving the terms of the resolution referred to in paragraph 6 above.
- 8. A certificate of the Additional Guarantor (signed by a director) confirming that guaranteeing or securing, as appropriate, the Total Commitments would not cause any borrowing, guarantee, security or similar limit binding on it to be exceeded.
- 9. A certificate of an authorised signatory of the Additional Guarantor certifying that each copy document listed in this Part II of Schedule 2 is correct, complete and in full force and effect and has not been amended or superseded as at a date no earlier than the date of the Accession Letter.
- 10. If available, the latest audited financial statements of the Additional Guarantor.
- 11. The following legal opinions, each addressed to the Agent, the Security Agent and the Lenders:
 - (a) A legal opinion of the legal advisers to the Agent in England, as to English law in the form distributed to the Lenders prior to signing the Accession Letter.
 - (b) For the purposes of The Council of the European Union Regulation No. 1346/2000 on Insolvency Proceedings (the **Regulation**) if the Additional

150

Table of Contents

Guarantor is incorporated in or has its centre of main interest (as that term is used in Article 3(1) of the Regulation) or establishment (as that term is used in Article 2(h) of the Regulation) in a jurisdiction other than England and Wales or is executing a Finance Document which is governed by a law other than English law, a legal opinion of the legal advisers to the Agent in the jurisdiction of its incorporation, centre of main interest or establishment (as applicable) or, as the case may be, the jurisdiction of the governing law of that Finance Document (the **Applicable Jurisdiction**) as to the law of the Applicable Jurisdiction and in the form distributed to the Lenders prior to signing the Accession Letter.

- 12. If the proposed Additional Guarantor is incorporated in a jurisdiction other than England and Wales, evidence that the process agent specified in Clause 40.2 (*Service of process*), if not an Obligor, has accepted its appointment in relation to the proposed Additional Guarantor.
- 13. Any security documents which are required by the Agent to be executed by the proposed Additional Guarantor.
- 14. Any notices or documents required to be given or executed under the terms of those security documents.
- 15. Such documentary evidence as legal counsel to the Agent may require, that such Additional Guarantor has complied with any law in its jurisdiction relating to financial assistance or analogous process.

Part III

Conditions Subsequent for Clause 23.37 (Conditions Subsequent in relation to German and Dutch Transaction Security)

- 1. A copy of the constitutional documents (including, in relation to the Acquiror, a recent extract from the Dutch trade register (*handelsregister*) relating to it) of each of the Acquiror and the Target.
- 2. A copy of a resolution of the board of directors of each of the Borrower and the Acquiror
 - (a) approving the terms of, and the transactions contemplated by, the Transaction Security Documents referred to below to which it is a party and resolving that it execute such Transaction Security Documents;
 - (e) authorising a specified person or persons to execute such Transaction Security Documents on its behalf;
 - (f) authorising a specified person or persons, on its behalf, to sign and/or despatch all documents and notices to be signed and/or despatched by it under or in connection with such Transaction Security Documents;
 - (g) in respect of the Acquiror:

151

Table of Contents

6011

- (i) a copy of a resolution of its board of supervisory directors (if any) approving its execution and the terms of, and the transactions contemplated by, the Transaction Security Documents referred to below to which it is a party; and
- (ii) a copy of a resolution of its general meeting of shareholders approving its execution and the terms of, and the transactions contemplated by, the Transaction Security Documents referred to below to which it is a party.
- 3. A specimen of the signature of each person authorised by the resolution referred to in paragraph 2 above.
- 4. A copy of each resolution and other formalities (if any) necessary to ensure that the constitutional documents of the Acquiror and the Target do not restrict the transfer of any pledged/charged shares.
- 5. A certificate of an authorised signatory of each of the Company, the Borrower and the Acquiror certifying that each copy document relating to it specified in this Part I of Schedule 2 is correct, complete and in full force and effect as at a date no earlier than the date of the Facility Agreement.
- 6. A solvency certificate issued by the Borrower and addressed to the Agent confirming the solvency of the Borrower, immediately before and after entry by it into any Transaction Security Document referred to below to which it is a party.
- 7. A copy of a good standing certificate (including verification of tax status) with respect to the Borrower, issued as of a recent date of the date of any Transaction Security Document referred to below to which it is a party by the Secretary of State or other appropriate official of the jurisdiction of incorporation or organisation of the Borrower.
- 8. At least two originals of the Transaction Security Documents executed by the relevant Obligor party thereto as specified below:

Name of Obligor	Transaction Security Document
Crown Westfalen LLC	Dutch Share Pledge
Crown Westfalen B.V.	Dutch Pledge over Dividend Account
	German Security Assignment Agreement over Acquisition

Agreement

German Share Pledge

9. A copy of all notices required to be sent under such Transaction Security Documents executed by the relevant Obligor duly acknowledged by the addressees and of all third party consents required in connection with the creation, perfection or registration of the Transaction Security Documents.

152

Table of Contents

10. A copy of all share certificates, transfers and stock transfer forms or equivalent duly executed by the relevant Obligor in blank in relation to the assets subject to or expressed to be subject to the Transaction Security and other documents of title which are required to be provided under the Transaction Security Documents.

11. **Legal opinions**

The following legal opinions, each addressed to the Agent, the Security Agent and any person who becomes a Lender:

- (a) legal opinions of Shearman & Sterling LLP, legal adviser to the Agent and the Arranger as to matters of German law;
- (b) a legal opinion of De Brauw Blackstone Westbroek, legal adviser to the Agent and the Arranger as to matters of Dutch law; and
- (c) a legal opinion of Dechert LLP, legal advisor to the Borrower as to matters of US law.
- 12. Such documentary evidence as legal counsel to the Agent may require that such Obligor has complied with any laws in its jurisdictions relating to financial assistance or analogous process.

153

Part IV

Transaction Security Documents and security related documents to be delivered by Additional Obligors

					Description of
					Security related
					documents and other
					action to be taken
					by Additional
					Obligor to protect
					or perfect or give
			Description of	Date by which	priority to
			Transaction	Transaction	Transaction
		Date by which must	Security Document	Security Document	Security and date
Name of Additional		become Additional	and Transaction	to be executed and	by which action is
Obligor	Capacity (Guarantor)	Obligor	Security	delivered to Agent	to be completed.
[insert name]	[Guarantor]	[]	[insert description]	[Closing Date]	
		15	4		

SCHEDULE 3 REQUESTS Part I Utilisation Request

From: Crown Westfalen LLC

To: Credit Suisse, London Branch

Dated:

Dear Sirs

Crown Westfalen LLC Term Facility Agreement dated 31 July, 2006 (the Agreement)

- 1. We refer to the Agreement. This is a Utilisation Request. Terms defined in the Agreement have the same meaning in this Utilisation Request unless given a different meaning in this Utilisation Request.
- 2. [Loan: We wish to borrow the Loan on the following terms:
 - (a) Proposed Utilisation Date: [] (or, if that is not a Business Day, the next Business Day)
 - (b) Amount: [] or, if less, the Available Facility] [Letter of Credit: We wish to arrange for the Letter of Credit to be issued by the Issuing Bank on the following terms:
 - (a) Borrower: Crown Westfalen LLC
 - (b) Proposed Utilisation Date: 31 July, 2006
 - (c) Currency and amount: Euro 25 million
 - (d) Expiry Date: 31 December, 2006]
- 3. We confirm that each condition specified in Clause 4.2 (*Further conditions precedent*) is satisfied on the date of this Utilisation Request.
- 4. [Loan: The proceeds of this Loan should be credited to [account].]
- 5. This Utilisation Request is irrevocable.

Yours faithfully

authorised signatory for

CROWN WESTFALEN LLC

155

Part II **Capitalisation Notice**

From: Crown Westfalen LLC

To: [The Agent]

Dated: **Dear Sirs**

Crown Westfalen LLC Term Facility Agreement dated 31 July, 2006 (the Agreement)

- We refer to the Agreement. This is a Capitalisation Notice. Terms defined in the Agreement have the same 1. meaning in this Capitalisation Notice unless given a different meaning in this Capitalisation Notice.
- 2. We refer to [the following Loan with an Interest Period ending on []] [to the following payment date for accrued Letter of Credit fee]:
- 3. We request that the following portion of the accrued interest or accrued Letter of Credit fee payable on the last day of such Interest Period or such fee payment date be capitalised in accordance with the terms of the Agreement: [specify amount].
- 4. This Capitalisation Notice is irrevocable.

Yours faithfully

authorised signatory for

CROWN WESTFALEN LLC

156

SCHEDULE 4 MANDATORY COST FORMULA

- 1. The Mandatory Cost is an addition to the interest rate in relation to the cost of compliance with (a) the requirements of the Bank of England and/or the Financial Services Authority (or, in either case, any other authority which replaces all or any of its functions) or (b) the requirements of the European Central Bank.
- 2. On the first day of each Interest Period (or as soon as possible thereafter) the Agent shall calculate, as a percentage rate, a rate (the **Additional Cost Rate**) in accordance with the paragraphs set out below. The Mandatory Cost will be calculated by the Agent by reference to the Agent s own rates and will be expressed as a percentage rate per annum.
- 3. The Additional Cost Rate for any Lender lending from a Facility Office in a Participating Member State will be the percentage determined by the Agent as the cost of complying with the minimum reserve requirements of the European Central Bank.
- 4. The Additional Cost Rate for any Lender lending from a Facility Office in the United Kingdom will be calculated by the Agent as follows:

Ex0.01 per cent.per annum 300

Where:

E is the rate of charge payable by the Agent to the Financial Services Authority pursuant to the Fees Regulations (but, for this purpose, ignoring any minimum fee required pursuant to the Fees Regulations) and expressed in pounds per £1,000,000 of the Fee Base of the Agent.

- 5. For the purposes of this Schedule:
 - (a) **Fees Regulations** means the Banking Supervision (Fees) Regulations 2001 or such other law or regulation as may be in force from time to time in respect of the payment of fees for banking supervision; and
 - (b) **Fee Base** has the meaning given to it, and will be calculated in accordance with, the Fees Regulations.
- 6. The Agent shall have no liability to any person if such determination results in an Additional Cost Rate which over or under compensates any Lender.
- 7. Any determination by the Agent pursuant to this Schedule in relation to a formula, the Mandatory Cost, an Additional Cost Rate or any amount payable to a Lender shall, in the absence of manifest error, be conclusive and binding on all Parties.
- 8. The Agent may from time to time, after consultation with the Company and the Lenders, determine and notify to all Parties any amendments which are required to be made to this Schedule 4 in order to comply with any change in law, regulation or any requirements from time to time imposed by the Bank of England, the Financial

157

Table of Contents

Services Authority or the European Central Bank (or, in any case, any other authority which replaces all or any of its functions) and any such determination shall, in the absence of manifest error, be conclusive and binding on all Parties.

158

SCHEDULE 5 FORM OF TRANSFER CERTIFICATE

dated 3	31 July, 2006 (the	Agreem	ent)			
Dated: Crown	Westfalen LLC	Term Fa	acility Agreement			
	[The Existing Lenc	der] (the	Existing Lender) and [The New Lender] (the	New Lender)
To:	[] as Agent					

- 1. We refer to the Agreement. This is a Transfer Certificate. Terms defined in the Agreement have the same meaning in this Transfer Certificate unless given a different meaning in this Transfer Certificate.
- 2. We refer to Clause 25.5 (*Procedure for transfer*):
 - (a) The Existing Lender and the New Lender agree to the Existing Lender transferring to the New Lender by novation all or part of the Existing Lender s Commitment, rights and obligations referred to in the Schedule in accordance with Clause 25.5 (*Procedure for transfer*).
 - (b) The proposed Transfer Date is [].
 - (c) The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 31.2 (*Addresses*) are set out in the Schedule.
- 3. The New Lender expressly acknowledges the limitations on the Existing Lender s obligations set out in paragraph (c) of Clause 25.4 (*Limitation of responsibility of Existing Lenders*).
- 4. This Transfer Certificate may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Transfer Certificate.
- 5. The New Lender hereby expressly consents to and ratifies the declaration of the Security Agent made on behalf of and in the name of the New Lender as Future Pledgee (as defined in the German share pledge agreement, the German Share Pledge Agreement) in the German Share Pledge Agreement. The New Lender confirms that it is aware of the German Share Pledge Agreement.
- 6. This Transfer Certificate is governed by English law.

159

Table of Contents

THE SCHEDULE

Commitment/rights and obligations to be transferred

[insert	t relevant	
[Facility Office address, fax number	er and attention details for notices and account details for payments,]	
[Existing Lender]	[New Lender]	
By:	By:	
This Transfer Certificate is accepte	ed by the Agent and the Transfer Date is confirmed as [].	
[Agent]		
Ву:		
	160	

SCHEDULE 6 FORM OF ASSIGNMENT AGREEMENT

т.,	г		۸ ~	4
To:		as	Αg	em

From: [the Existing Lender] (the Existing Lender) and [the New Lender] (the New Lender)

Dated:

Crown Westfalen LLC Term Facility Agreement dated 31 July, 2006 (the Agreement)

1. We refer to the Agreement. This is an Assignment Agreement.

2.

- (a) We refer to Clause 25.6 (*Procedure for assignment*).
- (b) The Existing Lender assigns absolutely to the New Lender all the rights of the Existing Lender under the Agreement, the other Finance Documents and in respect of the Transaction Security which correspond to that portion of the Existing Lender s Commitment and participations in Utilisations under the Agreement as specified in the Schedule;
- (c) The Existing Lender is released from all the obligations of the Existing Lender which correspond to that portion of the Existing Lender s Commitment and participations in Utilisations under the Agreement specified in the Schedule.
- (d) The New Lender becomes a Party as a Lender and is bound by obligations equivalent to those from which the Existing Lender is released under paragraph (c) above.
- 3. The proposed Transfer Date is [].
- 4. On the Transfer Date the New Lender becomes Party to the Finance Documents as a Lender.
- 5. The New Lender expressly acknowledges the limitations on the Existing Lender s obligations set out in paragraph (c) of Clause 25.4 (*Limitation of responsibility of Existing Lenders*).
- 6. The Facility Office and address, fax number and attention details for notices of the New Lender for the purposes of Clause 31.2 (*Addresses*) are set out in the Schedule.
- 7. This Assignment Agreement may be executed in any number of counterparts and this has the same effect as if the signatures on the counterparts were on a single copy of this Assignment Agreement.

161

Table of Contents

- 8. The New Lender hereby expressly consents to and ratifies the declaration of the Security Agent made on behalf of and in the name of the New Lender as Future Pledgee (as defined in the German share pledge agreement, the German Share Pledge Agreement) in the German Share Pledge Agreement. The New Lender confirms that it is aware of the German Share Pledge Agreement.
- 9. This Assignment Agreement is governed by English law.
- 10. This Assignment Agreement has been entered into on the date stated at the beginning of this Assignment Agreement.

162

THE SCHEDULE

Commitment/rights and obligations to be transferred by assignment, release and accession

[insert relevant details]	
	ad attention details for notices and account details for payments]
[Existing Lender]	[New Lender]
By:	Ву:
[Signature of this Assignment Agreem	ed by the Agent and the Transfer Date is confirmed as []. ent by the Agent constitutes confirmation by the Agent of receipt of notice of the otice the Agent receives on behalf of each Finance Party.]
[Agent]	
By:	
	163

SCHEDULE 7 FORM OF ACCESSION LETTER

То:	[] as Agent
From:	[Subsidiary] and Crown Northcorp Inc.
Dated:	
Dear S	
	n Westfalen LLC Term Facility Agreement
dated	31 July, 2006 (the Agreement)
1.	We refer to the Agreement. This is an Accession Letter. Terms defined in the Agreement have the same meaning in this Accession Letter unless given a different meaning in this Accession Letter.
2.	[Subsidiary] agrees to become an Additional Guarantor and to be bound by the terms of the Agreement and the other Finance Documents as an Additional Guarantor pursuant to Clause 25.10 (Additional Guarantors) of the Agreement. [Subsidiary] is a company duly incorporated under the laws of [name of relevant jurisdiction] and is a limited liability company and registered number [].
3.	[Subsidiary s] administrative details are as follows:
	Address:
	Fax No.:
	Attention:
4.	This Accession Letter is governed by English law.
	This Guarantor Accession Letter is entered into by deed. Crown Northcorp Inc. [Subsidiary]

as Agent

To:

SCHEDULE 8 FORM OF COMPLIANCE CERTIFICATE

	-
From:	Crown Westfalen LLC
Dated:	
Dear S	irs
Crown	Westfalen LLC Term Facility Agreement
dated 3	31 July, 2006 (the Agreement)
1.	We refer to the Agreement. This is a Compliance Certificate. Terms defined in the Agreement have the same

2. We confirm that:

(a) in respect of the three month period ending on [], Cashflow for such three month period was [] and Total Debt Service for such three month period was []. Therefore Cashflow for such three month period was [] times Total Debt Service for such three month period and the covenant contained in Clause 22.2(a) (*Fixed Charge Cover*) of the Agreement [has/has not] been complied with;

meaning in this Compliance Certificate unless given a different meaning in this Compliance Certificate.

- (b) in respect of the Relevant Period ending on [], EBITDA for the Relevant Period was [] and Net Interest Payable for the Relevant Period was []. Therefore EBITDA for such Relevant Period was [] times Net Interest Payable for such Relevant Period and the covenant contained in Clause 22.2(b) (*Interest Cover*) of the Agreement [*has/has not*] been complied with;
- (c) on the last day of the Relevant Period ending on [], Total Debt was [] and EBITDA for such Relevant Period was []. Therefore Total Net Debt at such time [did/did not] exceed [] times EBITDA for such Relevant Period and the covenant contained in Clause 22.2(c) (Total Leverage) of the Agreement [has/has not] been complied with;
- (d) on the last day of the Relevant Period ending on [], Total Net Worth was []. Therefore Total Net Worth at such time was [lower/higher] than 28 million and the covenant contained in Clause 22.2(d) (Minimum Net Worth) of the Agreement [has/has not] been complied with;
- (e) on the last day of the Relevant Period ending on [], Total Net Debt was [] and EBITDA was [] for such Relevant Period. Therefore Total Net Debt at such time [did/did not] exceed [] times EBITDA for such Relevant Period and the covenant contained in Clause 22.2(e) (Total Net Leverage) of the Agreement [has/has not] been complied with;

165

Table of Contents

- (f) Capital Expenditure for the Financial Year of the Group ending on [] was []. Therefore Capital Expenditure during such Financial Year [was/was not] in excess of [] being the maximum expenditure permitted in that period [after taking into account unused capital expenditure for the preceding Financial Year equal to []] and the covenant contained in Clause 22.2(e) (Capital Expenditure) of the Agreement [has/has not] been complied with;
- 3. [*We confirm that no Default is continuing.*]*

Signed Signed

Director Director

Of Of

Crown Northcorp Inc.

[insert applicable certification language]**

for and on behalf of [name of auditors of the Company]***

- * If this statement cannot be made, the certificate should identify any Default that is continuing and the steps, if any, being taken to remedy it.
- ** To be agreed with the Company s auditors and the Lenders prior to signing the Agreement.
- *** Only applicable if the Compliance Certificate accompanies the audited financial statements and is to be signed by the auditors. To be agreed with the

Crown Northcorp Inc.

Company s auditors prior to signing the Agreement.

166

SCHEDULE 9 TIMETABLE Part I Loans

Delivery of a duly completed Utilisation Request (Clause 5.1 (Delivery of a Utilisation Request) U-5 9.30 a.m.

Agent notifies the Lenders of the Loan in accordance with Clause 5.3 (*Lenders* U-5 noon

Clause 5.3 (Lena participation)

EURIBOR is fixed

Quotation Day as of 11.00 a.m. London time

> Part II Letter of Credit

Delivery of a duly completed Utilisation Request (Clause

6.2 (Delivery of a Utilisation Request for the Letter of Credit) 10.00 a.m.

Credit)

Agent notifies the Lenders of the Letter of Credit in accordance with Clause 6.4(d) (Issue of the Letter of Credit) U noon

U

"U = date of Utilisation

"U-X = X Business Days prior to date of Utilisation.

167

SCHEDULE 10 FORM OF BANK GUARANTEE

To: Bayerische Hypo-und Vereinsbank AG (the **Beneficiary**) 31st July, 2006

Guarantee no. LS 2006/8

At the request of Crown Westfalen LLC, Credit Suisse, London Branch (the **Issuing Bank**) issues this guarantee (**Guarantee**) in your favour on the following terms and conditions:

1. **Definitions**

In this Guarantee:

Business Day means a day (other than a Saturday or a Sunday) on which banks are open for general business in London.

Demand means a demand for a payment under this Guarantee in the form of the schedule to this Guarantee.

Expiry Date means 31 December, 2006.

Total Guaranteed Amount means 25,000,000.

2. Issuing Bank s agreement

- (a) The Beneficiary may request a single drawing under this Guarantee by giving to the Issuing Bank a duly completed Demand in respect of the Total Guaranteed Amount. A Demand must be received by the Issuing Bank before the Expiry Date or by no later than 5.00 p.m. (London time) on the Expiry Date.
- (b) Subject to the terms of this Guarantee, the Issuing Bank unconditionally and irrevocably undertakes to the Beneficiary that, within five Business Days of receipt by it of a Demand, it must pay to the Beneficiary the amount demanded in that Demand.
- (c) The Issuing Bank will not be obliged to make any further payment under this Guarantee if as a result the aggregate of all payments made by it under this Guarantee would exceed the Total Guaranteed Amount.

3. Expiry

- (a) The Issuing Bank will be released from its obligations under this Guarantee on the date (if any) notified by the Beneficiary to the Issuing Bank as the date upon which the obligations of the Issuing Bank under this Guarantee are released.
- (b) Unless previously released under paragraph (a) above, on 5.00 p.m.(London time) on the Expiry Date the obligations of the Issuing Bank under this

168

Table of Contents

Guarantee will cease with no further liability on the part of the Issuing Bank except for any Demand validly presented under the Guarantee that remains unpaid.

(c) When the Issuing Bank is no longer under any further obligations under this Guarantee, the Beneficiary must return the original of this Guarantee to the Issuing Bank.

4. **Payments**

All payments under this Guarantee shall be made in Euro and for value on the due date to the account of the Beneficiary specified in the Demand.

5. **Delivery of Demand**

Each Demand shall be in writing, and, unless otherwise stated, may be made by letter, fax or telex and must be received in legible form by the Issuing Bank at its address and by the particular department or office (if any) as follows:

Credit Suisse, London Branch Loan Agency Department 1 Cabot Square Canary Wharf LONDON E14 4QJ

Attention: Debbie Bardwell

Fax No: +44 (0)207 888 8398

6. **Assignment**

The Beneficiary s rights under this Guarantee may not be assigned or transferred.

7. **Governing Law**

This Guarantee is governed by German law.

8. **Jurisdiction**

The courts in Munich, Germany shall have jurisdiction to settle any dispute arising out of or in connection with this Guarantee.

Yours faithfully

Credit Suisse, London Branch

By:	
By:	
	169

SCHEDULE FORM OF DEMAND

To: Credit Suisse, London Branch
[], 2006
Dears Sirs

Guarantee no. LS 2006/8 issued in favour of Bayerische Hypo-und Vereinsbank AG (the Guarantee)

We refer to the Guarantee. Terms defined in the Guarantee have the same meaning when used in this Demand.

- 1. We certify that the conditions to closing (other than payment by Crown Westfalen B.V.) under the Share Purchase and Transfer Agreement dated 31 July, 2006 between Crown Northcorp Inc., Crown Westfalen B.V. and ourselves in relation to the purchase of the share capital of Westfalen Bank AG (the **Agreement**) have been satisfied and that the sum of Euro 25,000,000 is due from Crown Westfalen B.V. and is unpaid under the Agreement on the Closing Date (as defined in the Agreement). We therefore demand payment of the sum of Euro 25,000,000.
- Payment should be made to the following account:

 Name:
 Account Number:
 Bank:

 The date of this Demand is not later than the Expiry Date.

 Yours faithfully

 (Authorised Signatory)
 For
 Bayerische Hypo-und Vereinsbank
 AG

Table of Contents 218

170

SCHEDULE 11 MODEL

171

Fax:

SIGNATURES

THE BORROWER **CROWN WESTFALEN LLC** By: Address: Fax: THE COMPANY CROWN NORTHCORP INC. By: Address: Fax: **CROWN NORTHCORP LIMITED** By: Address: Fax: **CROWN ASSET MANAGEMENT LIMITED** By: Address:

172

CROWN MORTGAGE MANAGEMENT LIMITED

By: Address: Fax:

REGAL MORTGAGES LLP

By: Address: Fax:

173

THE ARRANGER	
CREDIT SUISSE, LONDON BRANCH	
By: Address: Fax: Attention:	By:
THE ORIGINAL LENDER	
CREDIT SUISSE, LONDON BRANCH	
By: Address: Fax: Attention:	By:
THE AGENT	
CREDIT SUISSE, LONDON BRANCH	
By: Address: Fax: Attention:	By:
	174

Table of Contents

THE ISSUING BANK	
CREDIT SUISSE, LONDON BRANCH	
By: Address: Fax: Attention:	By:
THE SECURITY AGENT	
CREDIT SUISSE, LONDON BRANCH	
By: Address: Fax: Attention:	By:
	175

SIGNATORIES

EXECUTED AS A DEED on)	
behalf	,	
of CROWN WESTFALEN LLC)	
by)	
and)	Authorised Signatory
being persons who, in accordance with)	Authorised Signatory
the laws of the territory in which Crown)	
Westfalen LLC is incorporated, are)	
acting under the authority of)	
Crown		
Westfalen LLC)	
		Authorised Signatory
EXECUTED AS A DEED on)	
behalf		
of CROWN NORTHCORP INC.)	
by)	
and)	A 41 ' 10' 4
being persons who, in accordance with)	Authorised Signatory
the laws of the territory in which)	
Crown	`	
Northcorp Inc. is incorporated, are)	
acting under the authority of)	
Crown	`	
Northcorp Inc.)	Authorised Signatory
EXECUTED AS A DEED on)	
behalf		
of CROWN WESTFALEN B.V.)	
by)	
and)	
		Authorised Signatory
being persons who, in accordance with)	
the laws of the territory in which)	
Crown Westfalen B.V. is incorporated,)	
are acting under the authority)	

Edgar	Filing: O	wens Realty Mortgage, Inc Form DEF 14A
of Crown Westfalen B.V.)	Authorised Signatory
EXECUTED AS A DEED by CROWN NORTHCORP LIMITED acting by two directors or a director and its secretary))	
		Signature: Name of Director: Signature: Name of Director/Secretary: 176

EXECUTED AS A DEED by CROWN ASSET)	
MANAGEMENT	,	
LIMITED acting by two directors or a director)	
and its secretary)	
		Signature: Name of Director:
		Signature:
		Name of Director/Secretary:
EXECUTED AS A DEED by CROWN MORTGAGE)	
MANAGEMENT LIMITED acting by two directors or a director)	
)	
and its secretary)	
		Signature:
		Name of Director:
		Signature:
		Name of Director/Secretary:
EXECUTED AS A DEED by REGAL MORTGAGES LLP acting by two members))	
		Signature:
		Name of Member:
		Signature:
		Name of Member: 177

EXECUTED AS A DEED on)	
behalf		
of CREDIT SUISSE, LONDON)	
BRANCH acting in its capacity as)	
Arranger, Original Lender,)	
Agent,		
Issuing Bank and Security)	
Trustee		Authorised Signatory
by)	
and)	
being persons who, in accordance with)	
the laws of the territory in which)	
Credit Suisse, London Branch	í	
210000 200000, 2000000 210000	,	Authorised Signatory
is incorporated, are acting under the)	Travilorison Signatory
authority of Credit Suisse, London)	
Branch)	
		178